

inda Bha

Pesticide Use in Nepal

The Assessment of Residues and Risks



Propositions

 Data on the average use of pesticides in Nepal underestimates the current use of pesticides on vegetables. (this thesis)

2. The use of organophosphate insecticides, particularly chlorpyrifos, poses risks to human health and the environment in Nepal.

(this thesis)

3. Soil health is important to ensure food safety and security.

4. Scientific data and information should be made easily available without charge for researchers from non-profit organisations.

5. With hard work you will get there but a good guide can help you to reach your destination faster.

6. Completing a PhD is a learning process, keep calm and carry on even when the process gets tough.

Propositions belonging to the thesis entitled:

Pesticide Use in Nepal: The Assessment of Residues and Risks

Govinda Bhandari

February 2, 2021

Wageningen, The Netherlands

Pesticide Use in Nepal: The Assessment of Residues and Risks

Govinda Bhandari

Thesis committee

Promotor

Prof. Dr V. Geissen Personal chair, Soil Physics and Land Management Group Wageningen University & Research

Co-promotors

Dr K. Atreya Research Lead of PHASE Nepal Suryabinayak 4, Dadhikot, Bhaktapur

Dr X. Yang Lecturer at Soil Physics and Land Management Group Wageningen University & Research

Other members

Prof. Dr Jakub Hofman, Masaryk University, Czech Republic Dr habil. Abdallah Alaoui, University of Bern, Switzerland Dr Anke Huss, Utrecht University Prof. Dr Frank Langevelde, Wageningen University & Research

This research was conducted under the auspices of the Research School for Socio-Economic and Natural Sciences of the Environment (SENSE)

Pesticide Use in Nepal: The Assessment of Residues and Risks

Govinda Bhandari

Thesis submitted in fulfilment of the requirements for the degree of doctor at Wageningen University by the authority of the Rector Magnificus Prof. Dr A.P.J. Mol in the presence of the Thesis Committee appointed by the Academic Board to be defended in public on Tuesday 2 February 2021 at 01:30 p.m. in the Aula.

Govinda Bhandari Pesticide Use in Nepal: The Assessment of Residues and Risks, 192 pages.

PhD thesis, Wageningen University, Wageningen, the Netherlands (2021) With references, with summary in English

ISBN: 978-94-6395-640-6 DOI: https://doi.org/10.18174/536691

Table of contents

Chapter 1. General introduction7
Chapter 2. Factors affecting pesticide safety behaviour: The perceptions of Nepalese farmers and retailers
Chapter 3. Pesticide residues in Nepalese vegetables and potential health risks
Chapter 4. Concentration and distribution of pesticide residues in soil: Non-dietary human health risk assessment73
Chapter 5. Ecological risk assessment of pesticide residues in soils from vegetable production areas: A case study in S-Nepal
Chapter 6. Synthesis
Literature cited 151
English summary
Acknowledgements185
About the author

1. General introduction

1.1 Pesticide use in conventional and integrated farming systems

A pesticide is a chemical or a complex mixture of different chemicals used in agriculture to keep plants healthy by protecting them from pests and infestations. In modern agriculture, the use of pesticides has become an integral part of minimizing the loss of crop yields and improving the storage of grains. Worldwide use of pesticides has been growing rapidly and expected to reach 3.5 million tonnes by the end of 2020 (Sharma et al., 2019). There are >1000 active ingredients in use globally (WHO, 2018). One of the reasons for increasing the global application of pesticides is due to the growing demand for food to feed the rising global population. The world population is expected to surpass >8 billion by 2030, and to surpass 9 billion by 2050 (UN, 2019). The risks related to pesticide use have become a worldwide concern exacerbated by poor agricultural practices (Sharma and Peshin, 2016). Risks are higher due to the toxicity of the pesticides being used. Pesticide exposure contributes to the deaths of 200 million people per year globally (Tariq et al., 2007). The challenge is how to feed the world in a sustainable way.

Although pesticides are considered easy and effective to use and offer immediate results (Aktar et al., 2009) for reducing crop losses, their misuse has become a serious issue in agricultural sustainability in Asia (Enserink et al., 2013). In this region, farmers often overuse pesticides on cash crops such as vegetables (Schreinemachers et al., 2020). In the absence of effective pesticide policies in developing countries, the use of toxic and persistent pesticides is increasing (Ecobichon, 2001; Phung et al., 2012). Integrated pest management (IPM) has emerged as a method to support sustainable agriculture and the wise use of chemical pesticides while increasing crop yields and household incomes (FAO, 2013). Global increases in crop yields can help nations meet the second of the United Nations 17 Sustainable Development Goals: "End hunger, achieve food security and improved nutrition and promote sustainable agriculture" by 2030 (UN, 2015). Sustainable agriculture reduces hunger, secures ecosystem functions and conserves biological diversity (Rockström et al., 2016).

Integrated farming system is based on cultural, chemical and biological methods for sustainable management of pests and diseases. This farming system emphasises growing healthy plants without hampering ecosystems and motivates biological pest control by empowering farmers. The benefits of this farming system are fewer pesticide applications and replacement of high-dose, more-toxic pesticides with low-dose, less-toxic and reduced-risk pesticides, including biopesticides. Integrated farming that comprises integrated pest management (IPM) methods is a viable option for sustainable agricultural production and meeting the future demand for food without any risk (Archer et al., 2019; Romeh, 2018; Rose et al., 2019). IPM methods are environmentally-friendly methods that are promoted

by the EU as a framework for community action through its 2009/128/EC Directive related to pesticides (EU, 2009). IPM has shown to be affective for food safety and pest control (Abrol and Shankar, 2014; Mladenova and Shtereva, 2009). One of the goals of integrated farming system is to minimise pesticide use and reduce risk. Pesticide use in integrated farming system decreased by >50% compared to conventional farming system (Ahuja et al., 2015). In the same study, farmers using IPM methods increased net income and reduced risk, compared to conventional farmers. The IPM activities are limited in Nepal (Aryal et al., 2014), thus further investigation on the strength of IPM in Nepal would be an innovative task.

Conventional farming system has been shown to be based entirely on chemical pesticides disrupting a balance between organisms of the field ecosystems, pests and beneficial faunas and floras. Although productive, conventional farming has proven to be unsustainable, as it demands more investments, pesticides and technologies (Cristache et al., 2018). This farming system accelerates non-judicious application of pesticides resulting in the presence of pesticide residues in foodstuffs and soil, affecting human health and environment (Aktar et al., 2009). Despite the proven negative effects, conventional farming has become the top choice globally. Concentrations of pesticides in grapes from conventional farms exceeded their acceptable limits and posed higher risks than IPM farms (Turgut et al., 2011).

1.2 Pesticide use and human health

Although pesticides are beneficial, they can be fatal if not handled carefully. Farmers experience several health consequences after exposure (Esechie and Ibitayo, 2011; Gesesew et al., 2016; Karunamoorthi et al., 2012; Kim et al., 2017; Maumbe and Swinton, 2003; Ngowi et al., 2007; Zyoud et al., 2010). Of all stakeholders, exposure to pesticides typically occurs among farmers who work regularly in their fields (Damalas and Koutroubas, 2016) increasing their health risk. Developing countries are more vulnerable to the risks posed by pesticides due to the lack of training resources that would allow farmers and retailers to effectively deal with the hazards associated with handling pesticides (WHO, 2012). The other reasons for facing higher risks from pesticide exposure are linked to the use of banned pesticides, incorrect application methods, poor spraying equipment, inadequate storage facilities and the reuse of old pesticide containers for domestic purposes (Ecobichon, 2001; Ibitayo, 2006). Pesticide risk is also affected by the behaviour of a farmer during pesticide application (Lewis et al., 2016a). Most farmers and retailers in developing countries lack knowledge about the proper safety procedures as well as the technical skills needed when handling pesticides. Farmers are mainly exposed to pesticides during preparation and application, sprayer maintenance, re-entering the fields and spillage. Likewise, retailers are exposed while handling pesticides in their shops. Individuals are at risk not only during the handling of pesticides but also after its application through dietary and non-dietary exposure. The risk of pesticides depends upon many elements, including toxicity dose, frequency of application, duration of exposure and safety measures taken (Damalas et al., 2019). Previous studies stated knowledge, attitude and behaviour (KAB) of the farmers and retailers as the most influential factors for the safe use of pesticides (Akter et al., 2018; Gesesew et al., 2016; Jørs et al., 2014; Khan and Damalas, 2015b; Lekei et al., 2014b; Mohanty et al., 2013; Oesterlund et al., 2014). Moreover, fear of economic losses, absence of pesticide alternatives, use of faulty spraying equipment, misuse of pesticides, lack of proper maintenance of spraying equipment were the main barriers for farmers' safety behaviour (Khan, 2010; Raksanam et al., 2014b). Literature on safety behaviour of Nepalese farmers and pesticide retailers ("agrovets") are limited however, indicating a need for research on the KAB of pesticide handlers. Use of personal protective equipment (PPE) can reduce exposure to pesticides and associated risks, however there are a number of factors that prevented PPE use among farmers and retailers (Damalas and Hashemi, 2010; Walton et al., 2017; Yang et al., 2014). The risk of pesticide use is often underestimated in developing countries including Nepal. Convincing farmers and retailers to adopt adequate personal safety measures is a major challenge.

Safety precautions during pesticide application, storage and transport are minimal. A pesticide's label should appear clearly on packets and containers and must follow FAO and WHO guidelines (FAO, 1995). The USEPA and FAO pesticide safety suggestions include wearing impermeable hand gloves, long pants, long sleeve shirts, boots and a mask as major personal protective equipment. The correct use of PPE reduces pesticide exposure and associated risk (Garrigou et al., 2020). However, even the use of PPE at the time of pesticide application is found to be insufficient in many countries (Damalas et al., 2019; Kumari and Reddy, 2013; Levesque et al., 2012; Naidoo et al., 2010; Palis et al., 2006) including Nepal (Atreya et al., 2012; Khanal and Singh, 2016; Rijal et al., 2018).

1.3 Pesticide residues in the environment

After application, pesticides can be transported to unintended places via different routes and mechanisms such as uptake, drifting, volatilization, leaching, runoff, and adsorption all of which can cause risks to the environment and human health (Figure 1.1 A, B and C). Pesticides degrade slowly in the environment by different processes, including degradation by soil organism, chemical degradation (hydrolysis) and photo-degradation (photolysis). They form a long chain of residues in soil, ultimately reaching plants through pesticide uptake (Fantke et al., 2013; Ghanbari et al., 2017; Lehmann et al., 2017; Liu et al., 2016a; Mtashobya, 2017; Valcke et al., 2017), and soil organisms (Aamir et al., 2018; Akoto et al., 2013; Gao et al., 2019; Ghanbari et al., 2017; Vasickova et al., 2019). The uptake of pesticides depends upon physiochemical properties of soil and pesticide as well as environmental factors such as precipitation, temperature and humidity (Biswas et al., 2018). The absorbed residues of pesticide are either metabolized by plant and animal system or accumulate in plants and animals, causing biomagnification in the environment (Gupta and Gupta, 2020). Pesticide drift, a source of pesticide exposure, has negative impacts on farm workers as well as people living near pesticide-treated farms (Coronado et al., 2011).



Figure 1.1 Pesticide pathways and related risks to humans and the environment.

1.4 Risk assessment related to pesticide used in agriculture

1.4.1 Human health risk

Pesticide risk for humans can be estimated using different methods such as hazard quotient (HQ) and hazard index (HI). The HQ measures risk of a single pesticide for humans, while HI measures cumulative risk of multiple pesticides with the same mode of action. Values from methods such as HQ or HI >1 indicate unacceptable risks to human health. Previous studies have also used these methods for human health risk assessment (Akoto et al., 2015; Chen et al., 2015; Gad Alla et al., 2015; Lozowicka, 2015; Seo et al., 2013). Pesticide risk assessment refers to an assessment process that is based on the characterization of effect and exposure (Hardy et al., 2012). The exposure assessment includes the environmental fate and behaviour of pesticides in different matrices such as soil, water and air (Figure 1.1 B). Likewise, the effect assessment refers to the characterisation of risk. This PhD study has adopted EFSA's 4 steps of risk assessment: i) identification of hazard, ii) characterisation of exposure and iv) characterisation of risk.

Nowadays, food quality has become a serious issue due to the presence of significant pesticide residues in foodstuffs. Pesticides used in agriculture are more toxic and persistent and their residues deposited in foods and soils can pose risks for humans (Figure 1.1 C). For developing countries including Nepal, where hazardous pesticides are overused and misused (Atreya et al., 2011; Jallow et al., 2017a; Panuwet et al., 2012; Schreinemachers et al., 2012; Shammi et al., 2020; Tariq et al., 2007), it is a great challenge to make sure that pesticide concentrations in foods are safe to eat. Pesticides exposure due to food consumption has been reported globally (Aamir et al., 2018; Akomea-Frempong et al., 2017; Fang et al., 2015; Lehmann et al., 2017; Nougadere et al., 2020; Sieke et al., 2018; Valcke et al., 2017). Monitoring pesticide residues and their concentrations in food is crucial for the risk assessment. Databases compiled by FAO/WHO and the EU are the major which provide information on maximum residual limits (MRLs) of pesticides and their degradation products found globally in different foodstuffs. An MRL is the maximum concentration of a pesticide residue (expressed as mg/kg) in food that is legally accepted after the proper application of the pesticide. Concentrations of pesticides in food may exceed their MRLs, indicating poor agricultural practices (FAO, 2016). Furthermore, the EFSA and JMPR have their own human health-based toxic reference values for pesticides such as acute reference dose (ARfD) and acceptable daily intake (ADI) (both expressed in mg kg⁻¹ body weight day⁻ ¹), which are used in risk assessment for humans.

A dietary risk in humans indicates exposure to pesticides via consumption of food. The acute and chronic dietary risk of pesticides is concerning i) when estimated short-term intake (ESTI) of a pesticide exceeds the acute reference dose (ARfD) of the pesticide and ii) when estimated daily intake (EDI) of a pesticide exceeds acceptable daily intake (ADI) of the pesticide. The short-term exposure to a pesticide is based on the presence of its residual concentration in a "large portion" consumption of a specific food in a day, while the longterm exposure is based on the presence of its residue on the average consumption of the food in question (EFSA et al., 2018). In the dietary risk assessment, there are many approaches for handling the food samples that have concentrations below the limit of detection (left-censored data). Of all the approaches, the use of substitution methods is frequently used (EFSA, 2010). The dietary risk of individual pesticides in humans can be evaluated by an index such as HQ, which is equal to ESTI/ARfD and EDI/ADI for the acute and chronic risks, respectively.

Non-dietary routes of human exposure to pesticides include inhalation, ingestion and skin contact (Damalas and Koutroubas, 2016). Looking at these exposure pathways and how they connect with pesticide concentrations found in soils may allow researchers to estimate cancer and non-cancer risks for humans. Human diseases related to genetics, reproduction, and endocrine disruption can be a result of chronic pesticide exposure (Gupta, 2004;

Sabarwal et al., 2018). Organochlorines have the possibility to induce lifetime carcinogenic health risks in humans (IARC, 2015). A recent study conducted in Nepal demonstrated a moderate cancer risk due to a non-dietary exposure to pesticides (Yadav et al., 2016). Based on the average daily non-dietary intake of organochlorines (CDI, mg/kg body weight), the lifetime cancer risk from the pesticide in humans can be estimated by using the guidance provided by USEPA and their associated threshold values. Cancer risks from pesticides can be calculated by multiplying CDI with the carcinogenicity slope factor (CSF) (USEPA, 1989). The cancer risk concerns scientists when the value from this calculation is >1x10⁻⁴. Noncancer risks from pesticides become concerns when the CDI value of a pesticide exceeds its human health-based toxic reference value. As mentioned in the dietary risk assessment above, the non-cancer risks are also evaluated based on indices such as HQ and HI.

1.4.2 Environmental risk

The environmental risk of pesticides consists of harmful effects on non-target soil organisms (FAO, 2017b; Stanley and Preetha, 2016; Uwizeyimana et al., 2017). Pesticide concentrations in soil exceeding their threshold levels impact ecosystem services, which is inevitable (Rodríguez-Eugenio et al., 2018). Furthermore, pesticide residues bioaccumulate in different trophic levels in ecosystems, gradually increasing ecotoxicity (Kapsi et al., 2019; Palma et al., 2004; Vasickova et al., 2019). The ecological risk assessment (EcoRA) of pesticides has been prioritized in many places, but not in developing countries like Nepal. Ecotoxicity differs with differences in soil physical, chemical and biological factors (Ellis et al., 2007; Lavtizar et al., 2016; Liu et al., 2013). Thus, to monitor environmental risk of pesticides, a country should focus on their own legislative guidelines and thresholds (Wee and Aris, 2017). As there are limited ecotoxicological studies available in Nepal, the risk of pesticides used in agriculture for soil organisms is unknown. With this PhD thesis, we carried out EcoRA of chemical pesticides on earthworms, enchytraeids, springtails, mites and nitrogen as well as carbon mineralization organisms.

Pesticide risk for the environment can be estimated using different methods such as risk quotient (RQ) and toxicity exposure ratio (TER). The RQ and TER are used for the assessment of ecological risk of pesticides. The thesis adopted SANCO/10329/2002 recommended chronic toxicity exposure ratios (cTER) and indexes (RQ) for calculating toxicity of pesticides for earthworms and other soil organisms. RQ of pesticides can be categorised into 4 groups: no risk (RQ<0.01), lower risk (0.01 \leq RQ<0.1), moderate risk (0.1 \leq RQ<1) and higher risk (RQ \geq 1) (Sánchez-Bayo et al., 2002). Furthermore, pesticide risk is unacceptable when its TER doesn't exceed trigger point values. For acute and chronic pesticide exposure, TER \geq 10 and

≥5 assigned by the EU, respectively indicated acceptable trigger point values to soil organisms (EC, 2002; Jaabiri Kamoun et al., 2017). Studies done elsewhere have used these methods for the assessment of the environmental risk (Vasickova et al., 2019; Wee and Aris, 2017) of pesticide use, however in Nepal, there are nearly no similar kinds of risk assessment studies. The cTER is the ratio between the no observed effect concentration (NOEC) and the measured environmental concentration (MEC) or predicted environmental concentrations (PEC). For a precise risk assessment, pesticide concentrations in soils, such as MECs, based on two scenarios: i) general (considering the mean concentration of pesticide) and ii) worst-case (considering the maximum concentration of pesticide), were used. PECs are often estimated using mathematical models with default values based on the EUs' FOCUS-scenarios (EFSA et al., 2017). The index RQ of an individual pesticide is the ratio of the MEC or PEC to PNEC. The PNEC rests on the most susceptible organism and is obtained by dividing its concentration with the European Commission guidance assessment factor (AF) (EC, 2003). The AF ranged from 10 to 1000. The selection of the AF depended on the amount of accessible ecotoxicity data from literatures and databases.

1.5 Pesticide use in Nepal and associated risks

Nepal is an agricultural country with 3 ecological regions: Terai, Hilly and Mountainous. Over 65% of the population is engaged in agriculture. Agriculture's share of the GDP was large: 28% (AICC, 2018; MoF, 2017). Depending on the region, different cereals such as rice, wheat, maize, millet, legumes, oilseeds and vegetables are cultivated. Vegetables are planted as a cash crop across flat lands including river and road corridors. The average pesticide use is higher in the Terai region (flat land) than in other regions (PPD, 2015). Farmers in Nepal, especially in these specific areas, have gradually been shifting to commercial farming. Due to available human resources, employment opportunities, easy access to roads and markets, and the support of agricultural technology, production per unit area has been increased (Dahal et al., 2008; Raut et al., 2011). However, 35% of the agricultural production is lost due to pests, diseases and storage problems (PPD, 2013). The government of Nepal (GoN) has committed to intensifying agriculture with the aim of expanding agribusiness via higher inputs, including pesticides. As a consequence, pesticide consumption has been increasing by about 10-20% per year (Diwakar et al., 2008). In recent years, the import and application of pesticides in Nepal have grown notably, indicating farmers' dependency on chemical pesticides to grow crops. Overall, the average use of pesticides is approximately 396 g of active ingredient per ha (PPD, 2015), although pesticide use in vegetable farming is comparatively high. Jha and Regmi (2009) estimated that farmers used 2.63 kg of active ingredients per ha for Cole crops, for example. Therefore, commercial farmers who practiced conventional farming overused chemical pesticides

mainly aiming at a maximum profit. Along with overuse, the other serious problems of pesticide use are: i) the lack of adequate knowledge of farmers and retailers on pesticide risks for humans and the environment, ii) the inadequate safety measures followed during pesticide handling, iii) the use of highly toxic pesticides that are banned in the EU, iv) the absence of effective monitoring systems for pesticide residues in food and the environment, and v) the inadequate pesticide rules and regulations. Up until the end of 2018, 170 different chemical compounds had been officially registered and used in Nepal. Most of the compounds used were insecticides (>35%) (PQPMC, 2019). From the total imported pesticides, >80% was used in vegetable production and storage (Adhikari, 2017; PPD, 2015), indicating a high risk for farmers cultivating vegetables. Soil quality on Nepalese farms has been decreasing as a result of the high chemical inputs (MoPE, 2000). Furthermore, pest resurgence has accelerated the application of higher doses of pesticides than recommended and farmers typically use a variety of pesticide cocktails (Chhetri et al., 2014). To avoid such misuse and associated risks to human and environmental health, the GoN has been upscaling good IPM-based agricultural practices (MoAD, 2016).

IPM focussed on reduced reliance on chemical pesticides. A previous study in Nepal has claimed that IPM-practicing farmers had reduced pesticide applications by 36% over non-practitioners (Kafle et al., 2014). IPM trained farmers have practised IPM techniques, for example, use of improved seeds, biopesticides, non-toxic chemical pesticides, proper irrigation and fertilizer (Bhandari, 2012). The farmers were motivated to adopt good agricultural practises such as appropriate use of pesticides, crop rotation and intercropping. Likewise, studies carried out elsewhere stated lower concentrations of residues on the IPM-farmed foodstuffs than on non-IPM (Baker et al., 2002; Mladenova and Shtereva, 2009; Singh et al., 2009). This clearly indicated that IPM farming could contribute to minimizing the effects of pesticides on health and the environment without decreasing crop yields (Bürger et al., 2008; Mariyono, 2008).

The pictorial representation of pesticides used in Nepal based on their target organism and toxic behaviour shown in Figure 1.2 and Figure 1.3, respectively.





Figure 1.3 Hazard categories of pesticides in Nepal based on WHO class (PQPMC, 2019).

Vegetable farmers excessively applied pesticides and did not follow safety measures which resulted in higher risks of exposure (Aryal et al., 2014). While dietary exposure from pesticides can carry risks from both acute and chronic illnesses, non-dietary exposure from pesticides can carry both cancerous and non-cancerous risks. Due to acute exposure to pesticides, the most common health diseases reported were eye and skin irritation, headaches, respiratory discomfort, and asthma (Aryal et al., 2014; Atreya, 2008b; Vaidya et al., 2017). The health problems increased with the application of more hazardous pesticides (Lamichhane et al., 2019). Studies conducted elsewhere linked pesticide exposure with

chronic diseases such as diabetes and Parkinson's disease (Evangelou et al., 2016; Schneider Medeiros et al., 2020). Nepal has already banned 16 compounds that are highly toxic, while 8 other compounds are in the process of being banned (PQPMC, 2019).

1.6 Overall aim of the thesis

This PhD study expands our understanding of chemical pesticide residues and their risks to human health and the environment in conventional and integrated farming systems. In this thesis, attention has been given to the estimation of pesticide residues in vegetables and soil, and to their risk assessments: dietary, non-dietary and ecological (Figure 1.1). Specifically, the research objectives are as follows:

- 1. Estimate pesticide use in vegetables and identify the factors affecting the safety behaviour of farmers and pesticide retailers on pesticide use and handling.
- 2. Evaluate the presence of pesticides and their degradation products in vegetables and agricultural soils from both IPM and conventional fields.
- 3. Assess the dietary risk to humans from pesticides in vegetable crops that were exposed to the highest doses of pesticides in the field.
- 4. Identify the human health risks from non-dietary intake of pesticides present in agricultural soils.
- 5. Investigate the ecological risk of pesticides detected in soils from both IPM and conventional fields.

The detail outline of the thesis is shown in Figure 1.4.



Figure 1.4 Thesis outline.

1.7 Outline of the thesis

This thesis includes 6 chapters. Chapter 1 comprises a general introduction to pesticide use, safety behaviour during pesticide application/handling, pesticide residues and the appropriate risk assessment methods to follow. The Chapter also discusses the significance of integrated farming, and increased use of pesticides in conventional farming and the associated risks.

Chapter 2 describes the perceptions of farmers and retailers for changing their pesticide behaviour based on a widely used model: Health Belief Model. Pesticide perceived threats, benefits and barriers to safety behaviour of farmers and retailers were analysed. Several ways of improving pesticide safety behaviour were recommended.

Chapter 3 identifies the pesticide residues in chilli, eggplant and tomato samples collected from two farming systems: i) conventional and ii) IPM. Dietary risk assessment of the pesticides was performed and the major contaminants were identified.

Chapter 4 identifies the pesticide residues in soil at three different depths: 0-5, 15-20 and 35-40 cm. The soils from conventional and IPM farms were studied separately. Non-dietary risk assessments of the pesticides including cancer and non-cancer risks were performed.

Chapter 5 assesses the ecological risks posed by pesticides in different depths of soil. Furthermore, it characterized the pesticide risk in different fields of farms: IPM and conventional, and the risk was correlated with farmers' knowledge and behaviour.

Chapter 6 summarizes the major outcomes of this study and discusses their strengths. Findings of this PhD thesis have significance for the i) safety and health of farmers and retailers; ii) food safety; iii) conservation of soil organisms, and iv) promotion of IPM techniques.

Chapters 2, 3, 4 have been published in peer-reviewed scientific journals. Methods, results, and the discussions of each chapter are presented separately and the publications can be assessed as shown at the beginning of each chapter.

1.8 Study area

The study area is located in Province 5, Rupandehi district in Nepal (Figure 1.5). The two administrative units, ward no. 6 and 7 of the Gaidahawa Rural Municipality (27° 35.429' N and 83° 19.215' E) were selected. The population is 47, 565, and the area is approximately 96.79 sq. km. Of the total land area (9679 ha), agriculture covers about 82%. The land is flat, fertile and the dominant soil textural classes are clays and silty clays, with some sandy clays, sandy clay loams and loamy sands. The sandier textured soils are found mainly along streams and rivers, near the banks where deposition occurs. At a few locations some silty soils may be found (mainly silt loams and silty clay loams) where regular flooding occurs and sediment carried by rivers or streams are deposited over a broad area. The average precipitation per annum is about 1391 mm. The maximum temperature in summer reaches 42.4°C and the minimum temperature in winter reaches 8.7°C. The duration of sunshine ranges from 4.76 hours (July) to 9.09 hours (April). The maximum evaporation is observed in the months of April and May and ranges from 4.02 mm to 11.69 mm while the minimum evaporation is observed in December, January and February and ranges from 0.09 mm to 4.18 mm. The major vegetables that are grown in agricultural fields are tomato, cauliflower, cabbage, radish, eggplant, bottle gourd, bean, bitter gourd, okra and chilli (GRM, 2018).

From the wards, a total of 12 villages were selected for questionnaire surveys (Figure 1.5). The villages were selected since most of the farmers were engaged in conventional farming

and the agricultural activities were intensive. Few of the farmers in the areas of villages such as Kadamhawa, Loadpurahawa, Mujrahawa, Bisoria, Suryapura and Kanijahawa also practised integrated farming such as IPM farming. Pesticide retailers were randomly selected on the way to the villages. Soil samples from the villages and the standing vegetable crops such as tomatoes, chillies and eggplants were selected for pesticide analysis and risk assessment for adolescents and adults.



Figure 1.5 Location of the study area.

2. Factors affecting pesticide safety behaviour: The perceptions of Nepalese farmers and retailers

Indiscriminate use of pesticides in vegetable farming is an emerging problem resulting in increasing health and environmental risks in developing countries including Nepal. As there are limited studies focusing on farmers' and retailers' knowledge related to pesticide use and associated risks as well as safety behaviour, this study assesses their perceptions of pesticide use, associated impacts on human and environmental health and safety behaviours. This study is also intended to quantify pesticide use in vegetable farming. We used the Health Belief Model (HBM) to evaluate farmers' and retailers' safety behaviour associated with pesticides. We interviewed 183 farmers and 45 retailers. The study revealed that farmers applied pesticides at an average of 2.9 kg a.i./ha per crop per season; and insecticides, especially pyrethrins and pyrethroids as well as organophosphate, were the most frequently used. Retailers were more aware of the threats surrounding pesticide use and were thus more aware of the risks to their own health as well as to the health of animals, birds, fishes, and honey bees. Headache (73.8%) was the most commonly reported acute health symptom of pesticide use. Farmers often did not adopt the appropriate safety measures when handling pesticides sighting the constrained perceived barriers (direct path coefficient, DPC = -0.837) such as feeling uncomfortable and the unavailability of safety measures. Likewise, retailers lacked the incentive (direct path coefficient, DPC = 0.397) to adopt the necessary safety measures while handling pesticides. Training and awareness programs addressing safe handling practices and safety measures as well as education concerning the long-term risks of pesticide exposure on health and the environment, through radio, television and posters, may improve the safety behaviour of farmers and retailers.

Based on:

Bhandari, G., Atreya, K., Yang, X., Fan, L., Geissen, V., 2018. Factors affecting pesticide safety behaviour: The perceptions of Nepalese farmers and retailers. Science of The Total Environment 631/632, 1560-1571.

2.1 Introduction

The Health Belief Model (HBM) is a cognitive model that attempts to explain and predict health behaviours and has been used to understand the safety behaviour of farmers while handling pesticides (Khan, 2010; Raksanam et al., 2014b). The model (Figure 2.1) says that in order to adopt safety behaviour, individuals need to perceive themselves susceptible to the possible illnesses and perceive the illnesses as serious (i.e. perceived threat), believe that the healthy behaviours are beneficial (i.e. perceived benefit), and believe that the benefits of healthy behaviours exceed the costs (i.e. perceived barriers) (Buglar et al., 2010; Coppens, 2016). If individuals believe themselves to be susceptible to a risky condition, think that the condition would have severe consequences, understand that adoption of available resources would beneficially reduce the condition of susceptibility and severity, and admit that the benefits of taking action outweigh the barriers to action, they are likely to follow safety behaviours that they believe will reduce their risk (Champion and Skinner, 2008).



Figure 2.1 The Health Belief Model (Champion and Skinner, 2008).

Unsafe use of pesticides can be considered as a threat to human health and the environment and good safety behaviours can strongly reduce the threat (Damalas and Eleftherohorinos, 2011; Houbraken et al., 2016; Jin et al., 2017). Safety behaviour depends on the perceived susceptibility, the severity of the risks and benefits as well as the current inhibiting factors to adopting good safety behaviours (Abdollahzadeh et al., 2015; Rezaei et al., 2018; Sharifzadeh et al., 2017). Raksanam et al. (2012) found a strong relationship between farmers' perceived susceptibility to pesticide exposure, the perceived severity of the consequence of exposure and the perceived benefit of the farmers' safety behaviour. Farmers and retailers may perceive the threats from pesticide differently and thus their personal actions to reduce their risk vary accordingly. Some farmers perceive higher threats from pesticides and show more safety behaviours such as not drinking, smoking or eating

during pesticide application as well as taking a bath and washing their cloths after spraying (Coppens, 2016). Similarly, farmers who had experienced health problems from applying pesticides may tend to adopt environmentally sound alternative pest management practices in order to reduce their pesticide risk (Lichtenberg and Zimmerman, 1999). The number of farmers who perceive higher threats from pesticide use corresponds to the increased use of safety measures such as gloves and shoes (Furlong et al., 2015; Hernandez-Valero et al., 2001). Considering this, our first hypothesis is that increased perceived threats from pesticide use is considered to have higher adherence to the safety behaviours.

Perceived barriers can affect the safety behaviours of farmers; the higher the perceived barrier, the lower the chances that farmers will report a higher adherence to safety behaviours (Khan et al., 2013; Raksanam et al., 2014a; Toan et al., 2013). Individual factors, such as the lack of time and comfort have been reported as barriers (Cabrera and Leckie, 2009; Levesque et al., 2012). Farmers may not use safety measures if they are an economical burden or a time restraint to performing the work (Snipes et al., 2009) or they are uncomfortable due to the heat stress and dampness experienced in the field (Walton et al., 2017). Factors such as the lack of training on safe pesticides use and the insufficient information provided on labels and package leaflets (normally in a foreign language) are considered the main barriers to the practice of good safety behaviour (Cabrera and Leckie, 2009; Damalas and Khan, 2017; Damalas and Koutroubas, 2017; Khan and Damalas, 2015a). Likewise, farmers who perceive the benefits of safety measures wear a combination of recommended safety gear such as long pants, long-sleeved shirts, aprons, hand gloves, protective masks, and hats during pesticide application (Salvatore et al., 2008; Walton et al., 2017). Thus, our second hypothesis is that increased perceived barriers decreases safety measure adherence, and increased perceived benefits of safety gear use increases the adherence to safety measures.

The Health Belief Model (HBM) comprises two additional components: cues to action and self-efficacy (Hanson and Benedict, 2002). Cues to action works as a 'trigger' and thus motivates individuals to change behaviours, while self-efficacy builds confidence in individuals to improve safety behaviours when handling pesticides (Bay and Heshmati, 2016). Farmers who are familiar with the short-term risk of poisoning during pesticide application adopt safety measures (Elmore and Arcury, 2001; Strong et al., 2008). Reoccurrence of symptoms such as headache and itching may act as internal stimuli to encourage the farmers to practice safe behaviours. External stimuli such as the provision of information via social media and trainings to facilitate the adoption of healthy behaviour (Kien, 2015) also act as triggers to encourage good pesticide practices. Safety hazards, safety culture, and production pressure can influence self-efficacy of individuals which in turn causes them to practice safe or unsafe behaviour (Brown et al., 2000; Rezaei et al., 2018). Providing proper safety equipment and work clothing would build a more positive work

experience and increase job satisfaction thereby increasing the self-confidence of individuals (Wagner et al., 2013). Safety education positively determines farmers' self-efficacy and enhances their skills to perform work more safely (Pettinger, 2000). Accordingly, our final hypothesis is that increased cues to action and self-efficacy have a positive effect on safety behaviour.

The government of Nepal has launched a number of agricultural development plans and policies with the focus on prioritizing increased production and productivity of agricultural crops through intensive use of pesticides. Especially after instituting these new policies, the misuse of pesticides by farmers has become a common practice. Researchers discovered that farmers apply pesticides to vegetable crops at rates nearly four times higher than recommended (Jha and Regmi, 2009) and this indiscriminate use is increasing (Atreya et al., 2011; CBS, 2015; Sharma et al., 2012). Intensification of agriculture has led to the overuse of pesticides in vegetable farming resulting in human health problems and ecosystem degradation (Atreya et al., 2011; Sharma, 2015). The misuse of pesticides has also resulted in pesticide poisoning (Atreya, 2008a; Atreya et al., 2011), acetylcholinesterase depression (Atreya et al., 2012; Neupane et al., 2014) and increased health burden. Pesticides have adverse effects on animals and fishes (Klemick and Lichtenberg, 2008), birds (Iwaniuk et al., 2006), and honey bees (Prisco et al., 2013). The risks posed to humans and the environment from pesticide use are evident. The adverse effects are more acute in developing countries where farmers lack training and access to awareness programs on the safe use of pesticides (Damalas and Khan, 2017; Khanal and Singh, 2016).

In many rural areas of Nepal, farmers seek solutions from retailers to help manage pests and diseases in their crops (Aryal et al., 2014). Thus, pesticide retailers play a very important role in the pesticide supply chain. Unfortunately, pesticide retailers also practice unsafe behaviours such as selling unregistered, prohibited, or date expired pesticides; mixing, reweighing and repacking pesticides; trading on the open market; and using false labels on the pesticides (Sharma et al., 2012).

Farmers' and retailers' beliefs concerning pesticide use, their understanding of the adverse consequences of such use and their own personal safety behaviours have not yet been examined using the Health Belief Model in Nepal. Thus, we initiated this study in order to assess farmers' and retailers' knowledge, attitude and safety practices regarding pesticide use and their perceptions of the associated health risks to humans and the environment. This study also quantifies pesticide use in vegetable farming. Knowledge concerning the existing use of pesticide and farmers' and retailer' perceptions regarding risks may serve as a guide when formulating policies aimed at achieving sustainable vegetable farming. Identifying cognitive factors that affect safety behaviour may prove beneficial in the design of efforts, campaigns and activities that encourage better adherence to good safety

behaviours. Identifying commonly used pesticides and educating people about the appropriate quantities of pesticides that should be used in vegetable farming could help encourage policy makers to take the necessary actions needed to prevent potential health risks of exposure.

2.2 Materials and methods

2.2.1 The study area

The study area is located in the Rupandehi district of Nepal, close to the India-Nepal border. The Gaidahawa Rural Municipality (27° 35.429' N and 83° 19.215' E) was selected (Figure 2.2) for this study. The population of the municipality is 47,565 individuals. The maximum temperature in summer reaches 42.4°C and the minimum temperature in winter reaches 8.7°C. The average annual rainfall is about 1391 mm. The main crop is rice followed by wheat, oilseeds and vegetables. Three crops are cultivated per year on irrigated land; while two crops are cultivated on non-irrigated land.



Figure 2.2 Location of the study area.

2.2.2 Sampling and data collection

The Gaidahawa Rural Municipality is comprised of nine wards – the smallest administrative unit. This study selected Ward 6 and 7 (Figure 2.2) where the majority of the farmers are engaged in commercial vegetable farming. Each ward is comprised of a number of villages. At the time of our study, the total number of households in both Wards was 1751 (CBS, 2012). We considered a random >10% of the households (183) which reflected a survey margin of error 6.8%. We used proportionate stratified random sampling of the Ward and village according to population density.

There were 13 pesticide retailers situated in the two Wards (DADO, 2015). However, a few farmers stated that they also buy pesticides from the retailers that are located on the way to a local market. Therefore, pesticide retailers situated on the way to the market were also considered. In the end, 45 pesticides retailers were interviewed.

We also conducted key informant interviews. Three officers from the Pesticide Registration and Management Division (PRMD) and four officers from District Agriculture Development Office (DADO) were interviewed. In addition, we also conducted focus group discussions. We made a questionnaire for the focus group discussions with the aim of acquiring information concerning: (i) pesticides use history, (ii) pesticides names, (iii) major crops grown and the pesticides used, (iv) pesticide poisoning, (v) acute health symptoms and possible diseases associated with exposure, (vi) pesticides and vegetable markets, (vii) safety practices and (viii) institutional support. Five focus group discussions, comprising 15-20 vegetable farmers, were conducted. Direct observations of the farmers' fields and retailers' stores were also made.

A semi-structured questionnaire containing modifying factors, individual beliefs and actions of the HBM (Figure 2.1) was designed based on previous questionnaires used in Atreya et al. (2012) and Bhandari (2014). A pilot study was conducted where 12 farmers and 2 retailers in a neighbouring village were interviewed to prepare the final survey questionnaire. Six students working on their Bachelor's degrees from the Institute of Agriculture and Animal Science (IAAS) at Tribhuvan University conducted the final household surveys. These students followed an all-day training program regarding the questionnaire where they learned how to effectively conduct the surveys. The survey questionnaire was mainly focused on (i) gender, age and education, (ii) pesticide use (iii) perceived susceptibility and severity, (iv) perceived benefits and barriers, and (v) self-efficacy and cues to action (Table 2.1). Prior informed consent from the respondent was obtained before conducting the face-to-face interview.

-		analysist		
Factors	Variable description (binary, 0 = no, 1 = yes)		Farmer	Retailer
			Mean (SD)	Mean (SD)
Perceived	Pesticides negatively affect	health	0.95 (0.22)	1 (0)
threat		Children's health	0.85 (0.36)	1 (0)
(PT)		livestock	0.92 (0.27)	1 (0)
()		hirds	0.66 (0.48)	1 (0)
		waterhodies	0 75 (0 43)	1 (0)
		fishes	0.80 (0.40)	1 (0)
		honeyhees	0.78 (0.42)	1 (0)
	Pesticides' long-term health effects	noncybees	0.70(0.42) 0.44(0.50)	0.67 (0.48)
	Exposure to pesticides while bandling		0.44 (0.50)	0.87 (0.40)
	Pesticides entering human body		0 01 (0 20)	0.84 (0.37)
Solf-	Know the colour codes of pesticides		0.01 (0.20)	0.51 (0.51)
officacy	Asking for information on the uses of posticides	during nurchasa	0.10 (0.31)	0.51 (0.51)
(SE)	Paceived advice on posticide use	utiling putchase	0.01(0.49)	
(32)	Attended training on safe use and handling of nor	ticidos	0.74 (0.44)	0 97 (0 24)
Porceived	Have participated in workshops and somipars role	ated to posticidos	0.17 (0.58)	0.87 (0.34)
harriar	Paliaved colour codes of posticidos not important			0.33 (0.30)
	Monitoring and support for posticide retailors suf	ficient		0.24 (0.45)
(PDd)	Attended training on protective equipment	ncient		0.27 (0.45)
	Attended training on protective equipment	Non government	vacuizations	0.15 (0.54)
	include experts from	Covernment organ	vizations	0.96 (0.21)
	include experts from	Government organ		0.15 (0.54)
	Training on posticido usos did pot includo	Pesticide suppliers	and companies	0.96 (0.21)
	ava ante frame	Non-government o	ingaliizations	0.98 (0.15)
	experts from	Government organ		0.13 (0.34)
	Training on protoctive devices and clothing did	Non government	and companies	0.98 (0.15)
	nating on protective devices and clothing did	Non-government o		0.98 (0.15)
	not include experts from	Bosticido cumpliore	and companies	0.15 (0.54)
	While chrowing do you horsoly a horrior on using	Pesticide suppliers		0.98 (0.15)
	write spraying up you perceive barrier on using	lide long clooved shirt	0.52 (0.50)	
		long pants	0.23 (0.43)	
		alouos	0.25 (0.45)	
		gioves	0.66 (0.55)	
Porcoived	Suggest light trap for controlling insects and post	IIIdSK	0.45 (0.50)	0 16 (0 27)
henefit	Suggest nerromone tran for controlling insects and pests	nd nests		0.10(0.37)
(PRo)	Are farmers benefited by wearing while spray	hat		0.22 (0.42)
(гве)	Are faithers benefited by wearingwhile spray	alacc		0.00 (0.30)
		long-slooved shirt		0.30 (0.30)
		dovos		0.95 (0.23)
		gioves		0.90 (0.21)
		long pants		0.96 (0.21)
	Go abroad (India) to huy pesticides	iong pants		0.30 (0.21)
	Go far away from your business, such as Kathman	du to huv pesticides		0.10 (0.37)
	Think his pesticides and their use beins environm	ent protection		0.24 (0.43)
	Do farmers need toafter spraving	wash hands	0.99 (0.10)	0.78 (0.42)
	Do farmers need toarter spraying	tako bath	0.95 (0.10)	
		wash cloths	0.85 (0.30)	
	During spraving don't	drink	0.70 (0.43)	
	During spraying, don't	smoko	0.79 (0.41)	
		SITIONE	0.92 (0.28)	
Cues to	Registered your husiness	Edl	0.95 (0.22)	0 87 (0 21)
cuesto	Registered your business			0.87 (0.34)
	Renew registration certificates			0.30 (0.30)
(CIA)	Participated in any social actions related to pastic	ides such as "No posti	cide use wook"	0.40 (0.50)
	Hoard about unfortunate incidences related to re-	nues such as inv pesti		0.47 (0.50)
	Heard about unfortunate incidences related to pe	sticiue poisoning	0.57 (0.48)	0.75 (0.45)
	Dooths in your family due to posticides	er unity		0.02 (0.15)
	De community farmare apply posticidas cofely		0 02 (0 25)	0.07 (0.25)
	Do community farmers apply pesticides safely		0.95 (0.25)	

Table 2.1 Descriptive statistics of variables used in data analysis.

Factors	Variable description (binary, 0 = no, 1 = yes)	Farmer	Retailer	
		Mean (SD)	Mean (SD)	
Behavior	Return pesticides that have expired or had lal	pels removed to dealers		0.33 (0.48)
change (BC)	Wash hands after handling pesticides			0.47 (0.50)
	While handling pesticides, do you wear	hat	0.43 (0.50)	0.04 (0.21)
		glass		0.11 (0.21)
		gloves	0.07 (0.26)	0.13 (0.34)
		mask	0.55 (0.50)	0.49 (0.51)
		long pants	0.74 (0.44)	
		long-sleeved shirt	0.75 (0.43)	

Table 2.1 Descriptive statistics of variables used in data analysis (continued).

2.2.3 Quantification of pesticide use

The active ingredients (a.i.) in the different pesticides used in each of the 15 vegetable types grown in this area were estimated. The following formula was used.

 P^{e} (kg a.i./0.03 hectare) = pesticide concentration (a.i.) per spray x total spray per frequency used for 0.03 hectare (ha) of land x total frequency of pesticide used in the particular vegetable growing season.

The average use of a pesticide for each type of vegetable was calculated by adding up the pesticide use in the major crops and dividing this number by the total number of crops. This study determined that 14 pesticides were used in the cultivation of the major 15 vegetable crops.

2.2.4 Data analysis

Survey questions consisted of a mixture of positive and negative items and were grouped according to the key factors of the HBM. Each question was given equal value: '1' for every "yes" answer and '0' for every "no" answer which is a common technique used in previous studies (Goldman et al., 2004; Strong et al., 2008). We did not check the reliability of the questionnaire due to the limited use of Cronbach's alpha (Sijtsma, 2009). The summative indices (SI) of each factor were calculated and represent the mean scores. Pearson Correlation Coefficients were calculated for all factors. The significant factors related to the safety behaviour of farmers and retailers were selected for path analysis as a supplement to separate direct and indirect effects and present the relative significance of factors (Wardell et al., 2012). Path analysis has been widely applied in past studies to investigate diverse topics, including organic food consumption (Lockie et al., 2004; Yazdanpanah et al., 2015), land transformation (Tong et al., 2016), water consumption (Fan et al., 2013) and pesticide use (Fan et al., 2015). Lockie et al. (2004), divided the effects of these factors into

three groups of differing magnitudes: minor (<0.10), medium (0.10-0.19), and major (>0.20), which we adopted for this study. The study of factors causing minor effects on the safety behaviour of farmers and retailers was not a part of the current research.

2.3 Results and discussion

2.3.1 Gender, age and education

About 90% of the farmers interviewed were males. More than 25% of the interviewed farmers were \leq 20 years old, 47% were 30 to 49 years old and the remaining 23% were above 50 years old. About 30% of the farmers were illiterate and the rest had different levels of education, such as primary (23%), lower secondary (20%), secondary (19%) and college (8.7%).

Likewise, about 80% of the retailers were males. Nearly 34% of the retailers were 40-49 years old. Forty five percent of the retailers were \leq 20-39 years old and the rest were above 50 years old. Almost all retailers were literate. Most retailers (65%) had a college level education, while >31% had a secondary education (Table 2.2).

Respondent characte	eristic	Responde	nts		
		Farmer	%	Retailer	%
		(N=183)		(N=45)	
Gender	Male	164	89.6	36	80
	Female	19	10.4	9	20
Age	≤ 20	47	25.7	10	22.2
	30-39	42	23	10	22.2
	40-49	43	23.5	15	33.3
	50-59	33	18	8	17.8
	≥ 60	18	9.8	2	4.4
Education levels	Illiterate	54	29.5	0	0
	Primary (1-5 class)	42	23	1	2.2
	Lower secondary (6-8 class)	37	20.2	1	2.2
	Secondary (9-10 class)	34	18.6	14	31.1
	College (>10)	16	8.7	29	64.4

Table 2.2 Characteristics of farmers and retailers.

2.3.2 Types of vegetable grown

Vegetable farming was the primary source of income for farmers. Farmers grew a number of different vegetables in one year. Nearly 20% of the households cultivated tomato, followed by cauliflower (17.6%), cabbage (11.3%), radish (7%), brinjal (6.3%), and bottle

gourd (5.2%) (Figure 2.3). Less than 5% of the households cultivated other vegetables such as bean, bitter gourd, okra, chilli, sponge gourd, potato, pea, broccoli, cucumber, pumpkin, chickpea, cowpea, carrot, onion, fennel, spinach, coriander, and fenugreek.



Figure 2.3 Percentage of household cultivating major vegetable in the study area over the past 12 months.

2.3.3 Use of pesticides in vegetable farming

The most commonly used pesticides in vegetable farming are listed in Table 2.3. WHO's class II insecticides were used more frequently than fungicides and herbicides which is consistent with studies done in Ethiopia (Mengistie et al., 2016), Vietnam (Dasgupta et al., 2007), Armenia (Tadevosyan et al., 2013) and Nepal (Neupane et al., 2014). A few banned organochlorine and organophosphate pesticides, such as endosulfan and phorate, were still in use. In a group discussion, farmers honestly mentioned that whenever they travel to the neighbouring country of India, they buy pesticides. The comparatively cheaper price of pesticides in India was their justification for buying the pesticides. Indian retailers are selling hazardous insecticides under the name of 'bio-pesticides' (FICCI, 2015) which might accelerate the introduction of extremely hazardous, unidentified, and banned pesticides into Nepal due to the open and porous border.

The application of pesticides differed for each type of vegetable. The average seasonal application ranged from 0.27 to 7.78 kg a.i./ha (mean 2.90, SD \pm 2.33). This estimate is higher than a previous study (Jha and Regmi, 2009) that estimated about 2.37 kg a.i./ha for fungicides and 1.96 kg a.i./ha for insecticides. Similarly, our estimate of pesticide application rates differ strongly between regions in Nepal. For example, application rates were 1.85 kg

a.i./ha in Kavre, 1.73 kg a.i./ha in Rautahat, 1.65 kg a.i./ha in Jhapa, 1.25 kg a.i./ha in Banke and 0.71 kg a.i. /ha in Chitwan (Chhetri et al., 2014).

The most commonly used pesticide, cypermethrin, was applied by 76% of the households (Table 2.3). On average, mancozeb had the highest application rate (7.78 kg a.i./ha), followed by dichlorvos, chlorpyrifos, profenfos, triazophos, dimethoate, carbendazim, metalaxyl, chlorantraniliprole, alphamethrin, imidacloprid, quinalphos, cypermethrin, and emamectin benzoate. A past study conducted in the hills of Nepal stated that mancozeb was a widely applied pesticide in vegetable farming (Atreya and Sitaula, 2011). We observed an extremely high use of dichlorvos (23.12 kg a.i./ha) on the brinjal. Farmers applied much higher doses than recommended for insecticides (Table 2.4a) and fungicides (Table 2.4b) in vegetable farming. Farmers typically made a mixture of 2 to 3 types of pesticides before application and used these mixtures on vegetables such as tomato, brinjal, cauliflower and cabbage, which is an observation consistent with previous studies (Damalas and Khan, 2017; Mengistie et al., 2015; Ngowi et al., 2007).

SN	Pesticides	Pesticides group #	Common name	WHO	% of HHs
	category		D ((class	
1	Insecticides	Organophosphates	Protenotos		55.2
			Chlorpyrifos	11	44.3
			Dimethoate	II	41.5
			Dichlorvos	Ib	10.9
			Quinalphos	Ш	7.1
			Triazophos	Ib	5.5
			Phorate ⁺	la	<1
		Organochlorines	Endosulfan ⁺	II	<1
		Pyrethrins & Pyrethroids	Cypermethrin	Ш	76
			Alphamethrin	П	11.5
			Allethrin	П	<1
			Alphacypermethrin	П	<1
			Bifenthrin	П	<1
			Deltamethrin	П	<1
			Fenpropathrin	П	<1
			Fenvalerate	П	<1
			Lambda-cyhalothrin	П	<1
			Resmethrin	Ш	<1
		N-methyl carbamates	Carbofuran	Ib	<1
			Cartap hydrochloride	П	<1
		Other Insecticides and Acaricides	Imidacloprid	П	16.9
			Chlorantraniliprole	U	6
			Acetamiprid	П	<1
			Propargite	Ш	<1
		Biologicals and Insecticides of Biological Origin	Emamectin benzoate	П	23.5
2	Herbicides	Other Herbicides	Metsulfuron Methyl	U	<1
			Pendimethalin	П	<1
3	Other pesticides	Fungicides	Mancozeb	U	53.6
			Carbendazim	U	10.9
			Metalaxyl	Ш	6.6
			Cymoxanil	Ш	<1
			Thiram	Ш	<1
4	Unidentified Trade name	Sterlik, Mastrogen, Miret, Ariun. Megathane			

Table 2.3 List of pesticides used in the study area.

Note: Ia = Extremely hazardous; Ib = Highly hazardous; II = Moderately hazardous; III = slightly hazardous; U = Unlikely to present acute hazard in normal use; +banned pesticides; # Pesticides group is based on Roberts and Reigart (2013); HHs = Households.

Table 2.4a Estimated (E) and recc	[*] papnammc	* (R) dos	ses of pestic	cides for ve	egetable farr	ning [kg activ	'e ingredier	rt (a.i.)/ha] [r	nean (±Stan	dard Devia	ition)].
Pesticides/Crop	Brinjal Ch	illi Tomato	Bottle	Bean Spon	ge Broccoli	i Cauliflower	Okra	Cucumber	Cabbage	Bitter Radish	Pea Ave	erage
			gourd	gourc						gourd	nsı	e
Dichlorvos ^E	23.12		5.47			3.92	7.84		5.16	0.73	7.7:	1
	(21.51)		(6.77)			(1.48)	(8)		(1.87)	(06.0)	(7.9	(06
Dichlorvos ^R	na		0.500			na	na		na	0.500		
Chlorpyrifos ^E		11.59	5.74	4.18 2.52	3.98	2.51	1.48	1.69	4.75	4.89 6.8	4.5(9
		(10.07)	(3.34)	(3.02) (0.95)	(4.56)	(2.06)	(0.92)	(0.45)	(4.71)	(3.44) (8.34)	(2.8	37)
Chlorpyrifos ^R		na	na	na na	0.400	0.400	na	na	0.400	na na		
Profenfos ^E	13.6 4.1	14 6.83	1.1	2.88 3.6	3.79	2.58			1.89	5.24 2.53	0.59 4.00	9
	(16.19) (3.	.35) (4.36)	(0.96)	(1.72) (1.27)	(4.72)	(3.61)			(1.88)	(7.16) (2.95)	(0.35) (3.4	16)
Triazophos ^E		2.33	5.4								3.8	7
		(0.80)	(3.05)								(2.1	17)
Triazophos ^R		0.500	0.500									
Dimethoate ^E	8.72 1.7	79 3.38	2.97	2.52 0.59	1.24	1.87	1.68	5.87	4.12	0.62 0.83	0.39 2.6	1
	(5.08) (1.	(0) (65.	(1.15)	(3.77) (0.19)	(1.43)	(2.17)	(2.57)	(6.97)	(3.81)	(0.47) (1.08)	(0.25) (2.3	34)
Dimethoate ^R	0.600 na	0.300	na	na na	0.200	0.200	0.600-0.700	na	0.200	na na	na	
Quinalphos ^E				2.4		0.81				0.64	1.28	8
				(1.7)		(0.98)				(0.58)	(0.9	(76
Quinalphos ^R				na		na				na		
Chlorantraniliprole ^E	2.1	10	-	0.92							1.5	1
	(3.	.4)		(1.1)							(0.8	33)
Chlorantraniliprole ^R	0. <i>î</i>	111	-	0.185								
Alphamethrin ^E	3.54	2.56	0.66			1.03		0.75		0.17	1.45	5
	(2.62)	(1.66)	(0.74)			(0.89)		(0.21)		(0.15)	(1.3	31)
Imidacloprid ^E	0.44	1.08	-	0.41		5.87			0.58	0.23	1.4	4
	(0.29)	(0.80)		(0.52)		(12.84)			(0.59)	(0.18)	(2.1	(61
Imidacloprid ^R	na	na		na		na			na	na		
Cypermethrin ^E	1.1	10 1.34	0.28	0.50 0.4	0.4	0.27	0.15	0.17	0.33	0.62 0.18	0.48	8
	(1.	.25) (1.91)	(0.40)	(0.34) (0.25)	(0.46)	(0.2)	(60.0)	(0.04)	(0.36)	(0.34) (0.25)	(0.3	38)
Cypermethrin ^R	na	na	na	na na	na	na	0.060-0.070	na	0.060-0.070	na na		
Emamectin Benzoate ^E		0.77				0.15			0.13	0.04	0.2	7
		(0.98)				(0.15)			(0.27)	(0.02)	(0.3	34)
Emamectin Benzoate ^R		na				na			na	na		
na = Not available; *Sour	ces: (PRMD,	2013, 2016).										

Pesticides/Crop	Brinjal	Tomato	Bottle	Bean	Potato	Cauliflower	Cabbage	Radish	Pea	Average
			gourd							use
Mancozeb ^E	12.85	16.51		10.63	5.48	4.25	2.14		2.59	7.78
	(7.06)	(22.72)		(8.28)	(7.30)	(3.59)	(1.43)		(4.22)	(5.58)
Mancozeb ^R	na	1.125-		1.125-	1.125-	1.125-1.500	na		na	
		1.500		1.500	1.500					
Carbendazim ^E		1.77	1.02	3		0.53		3.25		1.91
		(0.99)	(1.1)	(2.12)		(0.35)		(3.99)		(1.19)
Carbendazim ^R		0.050	na	0.250		0.100		na		
Metalaxyl ^E		1.73								1.73
		(1.19)								(1.19)
Metalaxyl ^R		na								

Table 2.4b Estimated (E) and recommended* (R) fungicide use pattern in vegetable cultivation [kg active inaredient (a.i.)/ha] [mean (+Standard Deviation)].

na = Not available; *Source: (PRMD, 2016).

2.3.4 Pesticide related acute symptoms

Almost all farmers perceived acute health symptoms after pesticide application. The most frequently self-reported toxicity symptoms related to pesticides were headache (73.8%), skin irritation (62.3%), eye irritation (32.8%), weakness (22.4%), and muscle pain (19.1%) (Figure 2.4). These findings are consistent with past studies done in Nepal (Atreya, 2008b) and Vietnam (Dasgupta et al., 2007).



Figure 2.4 Frequency of acute symptoms related to pesticide use.

2.3.5 Pesticide safety behaviour of farmers

Farmers' safety behaviour was affected by several factors such as age, education, perceived threat, self-efficacy, perceived benefits, perceived barriers, and cues to action (Table 2.5).
This result should be interpreted with caution since there were very few female farmers in our study. The path analysis of farmers (Figure 2.5A) indicated that a perceived barrier (direct path coefficient, DPC = -0.837) had major negative effect on the safety behaviour, while self-efficacy (direct path coefficient, DPC = 0.104) and perceived benefits (direct path coefficient, DPC = 0.107) had medium positive effects. In our study, 53% farmers did not use hats, 24% did not use long-sleeved shirts and long pants, 88% did not use gloves and 45% did not use masks while spraying pesticides.

IDIVI.			
Construct	Farmers	Retailers	
Age	-0.271**	0.017	
Education	0.278**	-0.123	
Gender	0.031	-0.164	
Perceived threat	0.350**	0.482**	
Self-efficacy	0.303**	0.186	
Perceived benefits	0.182*	0.142	
Perceived barrier	-0.886**	-0.077	
Cues to action	0.172*	0.378*	

Table 2.5 Correlations between farmers' and retailers' pesticide use safety behaviour and factors of theHBM.

* = significant at p <0.05 and ** = significant at p<0.01.



Figure 2.5A Path analysis of factors affecting the safety behaviour of farmers while handling pesticides.



Figure 2.5B Path analysis of factors affecting the safety behaviour of retailers while handling pesticides. Acronyms: PT, perceived threat; SE, self-efficacy; PBe, perceived benefit; PBa, perceived barrier; CtA, cues to action; BC, behaviour change. The solid arrows denote direct effects and the dotted arrows denote indirect effects. The numbers indicate the correlation coefficients between two variables joined by an arrow; the variable at the base of the arrow is the independent variable.

This is because they perceived barriers such as unavailability, discomfort, inappropriateness, and unawareness (Figure 2.6), which is a finding consistent with a past study (Damalas et al., 2006). Farmers perceived higher barriers such as feeling discomfort wearing long pants and long-sleeved shirts while spraying, which might be due to heat stress. Age (indirect path coefficient, IPC = -0.252) had a major negative effect on perceived barriers whereas education (indirect path coefficient, IPC = 0.213) and perceived threats (indirect path coefficient, IPC = 0.281) had major positive effects. A farmers' education positively affected their safety behaviour, whereas perceived barriers negatively affected their behaviour. Young farmers perceived higher barriers regarding safety behaviour. About half of the farmers in our study were less than 40 years old. Most vegetable farmers had low levels of education and perceived lower severe threats, especially regarding the longterm health effects that accompanied low adherence to good safety behaviours while handling pesticides, which is a finding consistent with previous studies (Damalas and Hashemi, 2010; Jallow et al., 2017a).



Figure 2.6 Frequency of perceived barriers on personal protective behaviour.

Self-efficacy (indirect path coefficient, IPC = 0.184) and cues to action (indirect path coefficient, IPC = 0.113) had medium positive effects on perceived barriers (Figure 2.5A). About 82% of the farmers had not received training concerning safe pesticide handling. "I am illiterate, not trained and unaware of pesticide safety items, I doubt on efficiency of the advertised protective measures and have no other option", a young farmer stated in a group discussion. Farmers often mixed pesticides with their bare hands due to their poor knowledge and higher illiteracy rate (Stadlinger et al., 2011). Education and training positively affected the safety behaviour of farmers (Damalas and Khan, 2017; Damalas and Koutroubas, 2017; Gaber and Abdel-Latif, 2012). Most farmers were unaware of the colour codes of pesticides and they had not yet heard about any unfortunate incidences related to pesticide poisoning. They believed that pesticide-related risks such as ill health (headache, skin and eye irritation) are short-term and things that regularly occurred during their daily agricultural life. One-quarter of farmers had not yet received any advice concerning safety precautions regarding pesticide use. "I have not meet the government extension service provider since many years, and when I need information on the use of pesticides, I go to nearby pesticide retailer", a commercial farmer in one group stated. However, about 40% of farmers did not receive information about pesticide use from retailers. Since most farmers lack training and education related to pesticide safety and are unaware of pesticide poisonings, they perceived larger barriers related to using protective items which led to their inadequate safety behaviours despite the overuse of pesticides. Vegetable farmers in the study area were at a higher risk of pesticide exposure.

2.3.6 Pesticide safety behaviour of retailers

Safety behaviour of retailers was associated with perceived threats and cues to action (Table 2.5). Path analysis of retailers indicated that cues to action (direct path coefficient, DPC = 0.397) had major positive effects on their safety behaviour (Figure 2.5B). Although 86% of retailers received training on the proper handling of pesticides, only half of them knew the colour codes of pesticides. Sixty percent of retailers did not get any support from district agricultural offices, while 53% did not participate in social actions, such as meetings, seminars and conferences against pesticide use. "Awareness on safety behaviour has not regularly been broadcasted in radio, television and newspaper, however, the new pesticides are frequently advertised in the area", a retailer stated in a group discussion. All retailers perceived higher threats to their health and the environment from pesticides. Surprisingly, some did not recognize the long term effects of pesticides and about 16% believed that they were not exposed to pesticides while handling them. A few retailers referred to pesticides as "medicines" rather than 'toxins'. When referring to pesticides as "medicines", only the functional aspect of these compounds was highlighted and retailers did not handle the pesticides safely, which is a finding similar to previous studies (Al Zadjali et al., 2014; Damalas et al., 2006; Jallow et al., 2017a). The registration of their businesses and the renewal of their certificates positively affected the safety behaviour of retailers, in line with Lekei et al. (2014b). Few retailers neither registered their business (13%) nor renewed their certificates (44%) even when they were involved in a pesticide-related business which might have affected their safety behaviour. Previous studies (Haj-Younes et al., 2015; Lekei et al., 2014b) reported that about 67% of the retailers sold expired pesticides to farmers.

This study provides useful information concerning farmers' and retailers' knowledge and behaviours related to pesticide use. The results should be interpreted with caution as there were certain limitations to this study. We conducted this study with a small sample size of farmers and retailers near to the India-Nepal border where large quantities of forbidden pesticides were available (Bhandari, 2014) and thus the results are not representative of other regions of Nepal. This study stands on the self-reports of individual perception and behaviour and that can be only partially validated since people often only want to report socially accepted behaviours on self-report studies (Damalas and Abdollahzadeh, 2016). However, this study was conducted in a friendly way and there was a good level of cooperation with farmers and retailers. Nonetheless, this study has accomplished several things: quantifies pesticide used in vegetable farming, farmers' and retailers' perceived risks from pesticide use, and their safety behaviours, all of which could help policy-makers towards sustainable management of pesticides and agricultural development in Nepal.

2.4 Conclusion

The largest amounts of pesticides were used on multiple harvest crops such as brinjal, chilli, and tomato. Farmers perceived lower pesticide threats to their health and the environment as well as lower benefits and higher barriers to their safety behaviours. The use of masks, gloves, and hats were very limited because of their availability and because farmers found them uncomfortable. In contrast to farmers, retailers perceived comparatively higher threats to their health and the environment from pesticides. However, their safety behaviour was not satisfactory due to inadequate triggers. Both farmers and retailers are exposed to higher risks from pesticide exposure. This study recommends the following interventions for improving pesticide use and safety behaviours for pesticide stakeholders: (i) increase educational programs, such as documentaries and talk shows as well as disseminating news through radio, television, and newspapers to raise awareness about good safety behaviours and the long-term consequences of pesticide use; (ii) provide training to retailers concerning safe pesticide handling practices and eco-farming alternatives; (iii) make protective safety devices more accessible and modify them to reflect local needs; and (iv) strengthen monitoring mechanisms to reduce the illegal import of banned pesticides, especially along the porous India-Nepal border.

Acknowledgements

We are grateful to the Netherlands Universities Foundation for International Cooperation (Nuffic) for funding this research (through NFP-PhD-5160957247). The authors thank local farmers and pesticide retailers for their cooperation and support during the survey. We would like to thank the students Garima Joshi, Samikshya Pandit, Sunil Ghimire, Kumar Pun Magar, Dipak Pandey, and Ravi Raj Mishra for their support during the field survey. We also thank Robin Palmer for English revision. The comments and suggestions received from the anonymous reviewers were also appreciated.

3. Pesticide residues in Nepalese vegetables and potential health risks

We conducted this study in order to assess the pesticide residues in vegetables and examine the related human health risk. Therefore, residues of 23 pesticides (organophosphates, organochlorines, acaricides, fungicides, and insecticides of biological origin) were analysed in the three main vegetable crops grown in Southern Nepal: 27 eggplant, 27 chilli and 32 tomato samples representing (i) conventional (N=67) and ii) integrated pest management (IPM) fields (N=19). Pesticide residues were found in 93% of the eggplant samples and in all of the chilli and tomato samples. Multiple residues were observed in 56% of the eggplant samples, 96% of chilli samples and all of the tomato samples. The range ($\mu g/kg$) of total detected pesticide residues in eggplants, chillies and tomatoes was 1.71-231, 4.97-507, 13.1-3465, respectively. The most frequently detected pesticides in these vegetables were carbendazim and chlorpyrifos. Chlorpyrifos residues exceeded the EU maximum residue limits (MRLs) in 11% of the eggplant and 19% of the chilli and tomato samples. In total, the residues of 4 pesticides such as triazophos, omethoate, chlorpyrifos and carbendazim exceeded the EU MRLs. Compared to chilli and eggplant crops, more carbendazim was sprayed onto tomato crops (p<0.05). Adolescent and adult dietary exposure were assessed using hazard quotient (HQ) and hazard index (HI) equations for the identified pesticides. HOs>1 were observed for chlorpyrifos and triazophos in tomatoes. The highest acute HQ (aHQ) was for triazophos (tomato) in adolescents (aHQ=3.21) and adults (aHQ=3.31), showing the highest risks of dietary exposure. The cumulative dietary exposure showed chronic HIs<1 for organophosphates in eggplants, chillies and tomatoes. The concentration of pesticide residues in the vegetable crops from the integrated pest management (IPM) fields was considerably lower, suggesting a greater ability of these systems to reduce the dietary risks from exposure to pesticides.

Based on:

Bhandari, G., Zomer, P., Atreya, K., Mol, H.G.J., Yang, X., Geissen, V., 2019. Pesticide residues in Nepalese vegetables and potential health risks. Environmental Research 172, 511-521.

3.1 Introduction

Pesticides are an important tool used in commercial agriculture to control insects, weeds and diseases and maintain crop yield by minimizing losses (Khanal and Singh, 2016; Schreinemachers et al., 2017; Wang et al., 2017b). However, pesticides have many harmful environmental impacts (Ibitayo, 2006; Yuantari et al., 2015) on non-target species, food webs and ecosystem functions (Atreya et al., 2011; Haj-Younes et al., 2015; Lekei et al., 2014a; Oesterlund et al., 2014). Many studies have documented that plants take up pesticides from soil (Fantke et al., 2013; Florence et al., 2015). The persistence and dissipation of pesticides in plants depend on plant characteristics (Lu et al., 2014). Of all the foodstuffs tested for pesticides in Nepal, vegetables were found to have the highest levels (Koirala et al., 2007). Many scholars have suggested integrated pest management (IPM) as an alternative approach to farming in order to minimize the use of chemical pesticides in the developing world (Hossain et al., 2017; Pretty and Bharucha, 2015). The government of Nepal has been promoting IPM since 2002. IPM is a technique used in agriculture that reduces a farmer's reliance on chemical pesticides and ensures the growth of healthy crops while conserving the environment and ensuring food safety and security at the same time. The integrated use of chemical pesticides, bio-pesticides (Jholmol in Nepali), and pheromone traps are a few successful and effective IPM practices used in Nepal (Joshi et al., 2017). However, the adoption of IPM at the field level has been very slow.

Pesticides residues in food pose a serious risk to consumers. In general, pesticides are sprayed directly onto crops. Infants, children, and adults can be exposed to these pesticides by consuming pesticide-contaminated food. Residues of chemical pesticides have been detected in many food commodities such as cereals (Akoto et al., 2013; Guler et al., 2010; Hou et al., 2013), seafood (Ernst et al., 2018; Kafilzadeh, 2015; Yahia and Elsharkawy, 2014), and tea (Amirahmadi et al., 2013; Chen et al., 2015; Seenivasan and Muraleedharan, 2011). Compared with other foodstuffs, even higher levels of pesticides were found in fruits and vegetables, the most important commodities foods in peoples' diets. In the worst case residual contents reach levels that render fruits and vegetables unsuitable for human consumption, as was the case in countries such as Burkina Faso (Lehmann et al., 2017), Kuwait (Jallow et al., 2017b), Tanzania (Kariathi et al., 2016), and Senegal (Diop et al., 2016). Therefore, food monitoring programs focused on testing pesticide levels have been put into place in different countries to improve food safety and agricultural practices as well as minimize economic losses. However, in Nepal, this program is in its infancy. To examine possible exposures, the government of Nepal uses the qualitative data derived from the Rapid Bioassay Pesticide Residue (RBPR) analysis program. This program is mainly focused on data collected around Kathmandu, the capital of Nepal. The country lacks scientific information concerning pesticide residues found on food in other parts of the country (Giri et al., 2012). Farmers in Southern Nepal grow different types of vegetables in one season and they often use higher doses of pesticides more frequently than recommended (Bhandari et al., 2018). This could result in chronic, sub chronic, and acute dietary exposure to pesticides (Essumang et al., 2008), especially for humans (Reiler et al., 2015; Sinha et al., 2012).

There are a number of methods used for assessing the cumulative risks of pesticides (Jensen et al., 2015; Wilkinson et al., 2000) and the indicators most often used are the hazard quotient (HQ) and hazard index (HI) (Bhanti and Taneja, 2007; Chen et al., 2011; Lehmann et al., 2017). The HQ is a tool that is used for the assessment of health risk (HR) due to human exposure to a single pesticide (Gad Alla et al., 2015). In order to assess the cumulative risk of different pesticides, the EFSA (2013b) has developed a methodology focused on cumulative assessment groups (CAGs) for assessing the cumulative risk index of pesticides that cause the same toxic effects in tissues, organs and physiological systems that can produce joint cumulative toxicity which is expressed as HI.

Since food intake is the major pathway by which most humans are exposed to pesticides, demand for food safety regulation has increased globally. To protect people from the toxic effects of exposure to pesticides, the EU, FAO/WHO and EPA have established limits on pesticide residues in foods. International Food Standards such as EU, EPA and CODEX (FAO/WHO) MRLs are applied in the benchmark assessment of exposure to pesticides (Chen et al., 2012). A maximum residue level is the highest concentration of a pesticide residue that is legally accepted in or on food when pesticides are applied correctly (refereed as Good Agricultural Practices). Safety reference values, such as maximum residue limits (MRLs), acute reference dose (ARfD), estimated daily intake (EDI) and acceptable daily intake (ADI) have been administered at national (Clever, 2017; Kurai, 2017; MPI, 2017) and international levels (EU, 2018; FAO/WHO, 2018) to prevent health risks and overexposure from pesticides. These kinds of safety regulations, particularly concerning chemical pesticides in food in Nepal, is still under development (CBS, 2013). Previous studies have found pesticide residues in vegetables but these studies did not assess the health risks associated with dietary intake of pesticides (Giri et al., 2012; Koirala et al., 2007; Lama, 2008). The dietary intake of pesticides (especially the amount consumed) differs in different age groups which may lead to different levels of exposure to pesticides. The concentration of dichlorvos in vegetables exceeded EU MRLs in Nepal (Rawal et al., 2012). In the same study, the EDI of dichlorvos exceeded the ADI in 18 types of vegetables. Higher levels of pesticide residues in vegetables may contribute to a greater health risk and hence, a detailed pesticide residue analysis and risk assessment is necessary. New knowledge about the current scenario concerning pesticides and their residues have been emerging in other parts of the world but not yet in Nepal. Research on this topic in Nepal is even more urgent considering the fact that several pesticides that have been banned for use on vegetable crops in the EU are still used in Nepal. The continuous application of these pesticides along with their persistent and toxic nature (for properties of the pesticides see Table S3.1) further support the necessity of this research. Furthermore, studies (Bhandari et al., 2018; Rijal et al., 2018) have already shown that pesticides are being applied at higher than recommended rates in vegetable farming.

Therefore, the objectives of this study were (i) to determine residues of chemical pesticides (organophosphates, OPs; organochlorines, OCs; acaricides, ACs; fungicides, FUs and insecticides of biological origin, INsB) used in vegetable farming in Nepal; and (ii) to assess the health risk of adolescents and adults due to the ingestion of pesticides in and on their vegetables.

3.2 Materials and Methods

3.2.1 Study area

The study area is located in the Rupandehi district of Nepal. The Gaidahawa Rural Municipality (27° 35.429' N and 83° 19.215' E) was selected for this study (Figure 3.1). The population of the Gaidahawa Rural Municipality is 47,565. The average annual precipitation is about 1391 mm. The maximum temperature in summer reaches 42 °C and the minimum temperature in winter reaches 9 °C.



Figure 3.1 Location map of the study area.

3.2.2 Farming systems and pesticides applied

The major crop in Nepal is rice followed by wheat, maize, oilseeds and vegetables. While IPM is an ecosystem approach to crop production, contributing to reduce use of pesticides and health risks, few farmers in our study practiced IPM techniques, for example, use of hybrid seeds, ash, animal urine and dung. Few farmers practiced crop rotation or mixed cropping and the use of insect traps was minimal. Most farmers practiced conventional/intensive farming which was heavily dependent on pesticides application (Thapa, 2017). Farmers applied higher than the recommended amounts of pesticides to vegetable crops (Bhandari et al., 2018; Jha and Regmi, 2009). Of all the pesticides used within the study area, mancozeb had the highest application rate (7.78 kg a.i /ha), followed by dichlorvos, chlorpyrifos, profenfos, triazophos, dimethoate, carbendazim, metalaxyl, chlorantraniliprole, alphamethrin, imidacloprid, guinalphos, cypermethrin, and emamectin benzoate. Tomato, eggplant and chilli are the most common vegetable crops grown in the study area. Onto the aforesaid crops, growers applied higher doses of pesticides in different combinations (Bhandari et al., 2018). The planting and harvesting period for vegetables differed by crop. A detailed description of crop harvesting is provided in the Supplementary information (see Figure S3.1). Earlier studies (Arora, 2009; Baker et al., 2002; Ranga Rao et al., 2009) showed that food grown conventionally had higher concentrations of pesticide residues than food grown using IPM methods.

3.2.3 Sampling

Vegetable samples (27 eggplant, 27 chilli, and 32 tomato) from 54 local farmers' fields were collected during the winter of 2017. At the time of normal harvest, we collected vegetable samples from two types of farming systems: i) conventional farming (N=67) and ii) IPM fields (N=19). Altogether, 2-3 units of fresh vegetables were collected from each field (> 1kg each for tomato and eggplant and 0.5kg for chilli) in accordance with the procedures described in the FAO (1999). Samples were not rinsed/washed with water. A portion of each sample, without tops such as the sepal and peduncle, was prepared according to annex I of European Commission regulation 396/2005 EU (2010) using a knife and a chopping board and then thoroughly mixed. Next, 20 g of each sample was kept in a separate plastic bag at -20°C until pesticide extraction and analysis could be carried out. Organically produced tomato samples (Greenland Organic Vegetable Grower Schalkwijk BV, Netherlands) were used as blanks.

3.2.4 Sample extraction and clean-up

A detail description of the analytical methods used in the pesticide analysis is provided in the Supplementary information (see extraction and clean-up methodology, p. S4).

3.2.5 Chemical analysis

All chemicals and reagents used were of analytical grade. Reference standards (in total 23) used for identification in this residue analysis were from Dr. Ehrenstorfer GmbH (Augsburg, Germany). Active substance standards used in this study had a purity of >95%. ¹³C-Caffeine and PCB 198 (2,2',3,3',4,5,5',6-Octachlorobiphenyl) were used as internal standards.

A standard stock solution of each pesticide was prepared in acetonitrile at a concentration of 2,000 μ g/mL. A mixed standard solution was prepared at a concentration of 10 μ g/mL from the individual stock solutions. The calibration curve for the LC measurements was prepared by diluting 10 µg/mL of the mixed standard solution to achieve final concentrations of 0.25, 0.5, 1, 2.5, 5, 10, 25, and 100 ng/mL in a mixture of acetonitrile and water (1:1, v/v). Stock and working solutions were stored at 4°C until use. Pesticides were analysed through Liquid Chromatography Triple Quadrupole Mass Spectrometry (LC-MS/MS) and Gas Chromatography Triple Quadrupole Mass Spectrometry (GC-MS/MS) using the same extraction and clean-up method, chemicals, mobile phases, column characteristics and instrumentation conditions (see the Supporting information for full details, p. S4-S6). All the validation procedures and analytical quality control criteria were in line with those described in the guidance document for pesticide residues analysis in food and feed (EU, 2017). Briefly, 23 pesticides (Table S3.1), from now on called analytes, were identified according to (i) the retention time and peak shape of the respective reference standards and (ii) the ion ratio, with ratios between the quantification and confirmation transitions within $\pm 30\%$ of the average ion ratio of the calibration standards. The concentration of the analytes was calculated based on bracketing calibration, with a matrixmatched calibration standard.

Quality assurance and quality control of the method was performed by spiking duplicate samples as well as examining recovery, linearity, ion ratios, retention time and the limit of detection (LOD, Table 3.1) in order to verify compliance with identification requirements as outlined in SANTE/11813/2017 (EU, 2017). A calibration curve was injected at the start of the sample sequences. For recovery assessment and method validation, organic tomato was used as blank samples and fortified with a standard mix solution at 1, 2.5, 5, 10, and 25

 μ g/kg. Next, 5 ng/g of a blank tomato matrix was frequently run to check any interference due to contamination from the apparatus, solvents or chemicals used.

Fortified blank tomato samples presented a recovery of all analytes between 70 and 120%. Similar recovery values (80-120%) were observed in vegetable samples fortified with the mixture of standards. The calibration curves presented satisfactory linearity of response versus concentration, with correlation coefficients above 0.99. The concentration in each of the two aliquots (replicates) was within ±35% of the mean concentration of both aliquots.

3.2.6 Risk assessment

The risk assessment process was carried out based on the EFSA's risk assessment steps: (i) hazard identification; (ii) hazard characterisation; (iii) exposure assessment; and (iv) risk characterisation.

3.2.6.1 Hazard identification

Hazard identification determines whether exposure to a pesticide can cause an increase in the incidence of specific adverse health effects. Exposure to pesticides may generate many different adverse effects in human populations such as carcinogenesis, neurotoxicity, cytogenetic damage, and endocrine disruption in addition to developmental, reproductive, and immunological disorders (Mansour, 2004; Nicolopoulou-Stamati et al., 2016; Nougadere et al., 2012). We adopted the Pesticide Properties Database (PPDB) and the Bio-Pesticides Database (BPDB) to assess the potential health issues due to consumption of vegetables contaminated with pesticides.

3.2.6.2 Hazard characterisation

Hazard characterisation seeks to identify the quantitative relationship between a dose level and the resulting incidence of disease. Intake of vegetables contaminated with pesticides can have acute and chronic risks which can be assessed adopting following equations.

Acute/short-term HQ assessment (aHQ)

The aHQ was calculated based on the estimated short-term intake (ESTI) and the acute reference dose (ARfD) as:

ESTI = (the highest level of residue x the highest large portion of food	consumption per
day)/body weight	(3.1)
aHQ = ESTI/ARfD	(3.2)

Information on ARfDs was obtained from the EU pesticides database (http://ec.europa.eu/food/plant/pesticides/eu-pesticidesdatabase/public/?event=homepage&language=EN).

Chronic/long-term HQ assessment (cHQ)

The cHQ was calculated based on the estimated daily intake (EDI) and the acceptable daily intake (ADI) as:

EDI = (mean level of residue x average food consumption per day)/body weight.....(3.3) cHQ = EDI/ADI.....(3.4)

Information on ADIs was obtained from the EU pesticides database (http://ec.europa.eu/food/plant/pesticides/eu-pesticidesdatabase/public/?event=homepage&language=EN).

3.2.6.3 Exposure assessment

We choose to estimate dietary risks of exposure to pesticides in adolescents and adults. Population groups adopted from the international standards of the WHO (2013) were 10-19 years for adolescents and >19 years for adults. As data of Nepalese large portion consumption of the vegetables was not available, for aHQ, the maximum consumption (g/day) value was assumed as: 500 (eggplant), 50 (chilli), and 300 (tomato) for adult. For cHQ, we used average vegetable consumption data derived from the national survey carried out by the Ministry of Health (MoH et al., 2017) and agricultural data from the Ministry of Agricultural Development (MoAD, 2017). Adults consumed on average 9.6 g/day of eggplant, 2.85 g/day of chilli, and 61.3 g/day of tomato. In adolescents, the vegetables consumption (g/day) was assumed to be half that of an adults' daily vegetable consumption. Individual body weight measurements adopted from the international standards of the WHO (2011) were 32 kg for adolescents and 62 kg for adults.

3.2.6.4 Risk characterisation

Risk characterisation presents an integrated picture of the adverse health effects in exposed populations. We characterised the potential health risk by using the Hazard Quotient (HQ) and, for chronic risk also the Hazard Index (HI). The HQ is used for assessing the potential risk due to a single pesticide, while the HI for mixture risk takes into account multiple pesticides (Posthuma et al., 2018). The HI is the sum of the HQs. An HQ or HI > 1 denotes potential risk to human health (Darko and Akoto, 2008) while an HQ or HI \leq 1 indicates no risk (Chabukdhara and Nema, 2013; Sun and Chen, 2018). We calculated hazard indices based on this data due to its simplicity.

49

The HI is based on the cumulative effect of pesticides with similar mechanisms of action. Pesticides causing the same physiological effects in terms of site and nature were grouped to form CAGs. It is not likely that a sample will have the highest concentration of residues of all pesticides, henceforth, only the chronic HI was calculated by adding up the cHQs of pesticides of a group (only OPs) and was expressed as a hazard index (HI) as: HI = $\sum_{i=1}^{n} cHQi$(3.5)

3.2.7 Statistical analysis

Data analysis was performed using SPSS base 23.0 software. The Shapiro-Wilk test of normality of the variables was checked prior to performing the tests (Thode, 2002). Mean concentrations of different pesticides grouped by vegetable type and farming systems were tested. The Kruskal-Wallis one-way ANOVA was used to compare distributions across groups at the 95% confidence interval.

3.3 Results

3.3.1 Pesticide residues in vegetables

Out of the 23 tested analytes, 14 (approximately 61% of the tested analytes) were detected. About 97% of the tested vegetable samples contained at least one analyte. The observed concentration of chlorpyrifos in eggplant, tomato and chilli samples ranged from 1.19 to 45.3 μg/kg (41% of the samples), 1.07 to 1772 μg/kg (94% of the samples) and 1.29 to 491 µg/kg (81% of the samples), respectively. Likewise, the concentration of carbendazim in eggplant, tomato and chilli samples ranged from 1.21 to 154 μ g/kg (78% of the samples), 1.45 to 337 μ g/kg (100% of the samples) and 1.11 to 95 μ g/kg (81% of the samples), respectively. The analysis of the vegetables showed the presence of different metabolites with the following concentrations in tomato samples: omethoate (range 6.93-27.9 μ g/kg, 6% of the samples) and N-(2,6-dimethylphenyl)-N-(methoxyacetyl) alanine (1.47 µg/kg, 3% of the samples). Likewise, the concentrations of 3,5,6-trichloro-2-pyridinol in eggplant samples and N-(2,6-dimethylphenyl)-N-(methoxyacetyl) in chilli samples ranged from 4.53 μ g/kg to 30 μ g/kg (22% of the samples) and 1.32 to 2.32 μ g/kg (7% of the samples), respectively (Table 3.1). Of all the observed concentrations of pesticides, the mean concentration of carbendazim in tomato samples was significantly higher (p<0.05) than the concentration of this compound in chilli and eggplant samples.

The observed concentrations of pesticides that belong to different groups differed with respect to the type of vegetable and the order for detection of different groups of pesticides is shown in Figure 3.2. However, concentrations of pesticides that belonged to the group of fungicides were the same in all of the vegetables. Vegetables revealed a group diversity of the presence of different pesticides with a total of 8 combinations such as AC, AC+FU, AC+FU+OP, AC+OP, FU+OP, FU, OP, and INsB (for abbreviations, see Table 3.1). The most common combination in eggplant and tomato samples, FU+OP, corresponded to 37% and 68.8% of the samples, respectively. Furthermore, the most common combination in chilli, AC+FU+OP, corresponded to 44.4% of the samples.

Figure 3.2 The order of detection of different pesticides and their metabolites in the vegetables (A) eggplant, (B) tomato, and (C) chilli. Pesticides group is based on Roberts and Reigart (2013).



	Pesticides and metabolites	Eggplant (N=2)	7)			Tomat	to (N=32)				5	hilli (N=27)		
		Number and	Min-Max	Mean(±SD) N	ARLS N	umber and	Min-Max	Mean(±SD) MRLs	Number and	Min-Max	Mean(±SD) MRL	s LOD	p-value
Pesticide	S	percentage				percentage (in	_			percentage			(pg/kg)	
group		(in brackets				brackets) of				(in brackets)				
		of positive				positive				of positive				
		samples				samples				samples				
OPs	3,5,6-trichloro-2-pyridinol ⁿ	n 6 (22)	4.53-30	13.4 (10.1) N	IA 1	5 (47)	2.71-98.9	33 (32.1)	NA	5 (19)	2.96-13.5	6.47 (4.66) NA	2.50	0.140
	Chlorpyrifos	11 (41)	1.19-45.3	11.1 (14.1) 10	.0	0 (94)	1.07-1772	117 (341)	100	22 (81)	1.29-491	28.1 (104) 10	1	0.348
	Dichlorvos	1(N)	NA	NA 1	0.	S	NA	NA	10	3 (11)	1.35-2.76	2.18 (0.74) 10	1	NA
	Dimethoate	\N	NA	NA 1	.0	(3)	NA	NA	10	1 (4)	NA	NA 10	1	0.317
	Omethoate ^m	10	NA	NA 1	.0 2	(9)	6.93-27.9	17.4 (14.8)	10	NN N	NA	NA 10	1	NA
	Profenofos	1 (4)	NA	NA 1	.0 2	2 (69)	1-1640	134 (353)	10000	8 (30)	1.16-228	40.6 (81.2) 3000	1	0.583
	Quinalphos	10	NA	NA 1	0.	S	NA	NA	10	5 (19)	1.17-5.96	2.94 (1.79) 10	1	NA
	Triazophos	4 (15)	1.03-25.5	7.40 (12.1) 1	.0	5 (47)	1.16-685	65 (182)	10	1 (4)	NA	NA 10	1	0.201
	Chlorantraniliprole	\N	NA	NA 6	5 00	(16)	1.02-33.3	8.81 (13.9)	600	NN N	NA	NA 1000	1	NA
ACs	Imidacloprid	4 (15)	1.51-6.43	3.26 (2.23) 5	9 00	(19)	2.59-25.7	12.1 (9.02)	500	18 (67)	1.19-378	63.1 (108) 1000	1	0.281
	Carbendazim	21 (78)	1.21-154	19.7 (34) 5	00 3	2 (100)	1.45-337	50.3 (71.2)	300	22 (81)	1.11-95	20.1 (32.2) 100	1	0.002*
FUs	Metalaxyl	3 (11)	1.03-3.31	1.85 (1.27) 10	.0 2	1 (66)	1.54-88.6	19.6 (29.3)	300	8 (30)	1.08-439	62.1 (153) 500	1	0.086
	N-(2,6-dimethylphenyl)-N-	\N	NA	NA NA	IA 1	(3)	NA	NA	NA	2 (7)	1.32-2.32	1.82 (0.71) NA	1	
	(methoxyacetyl)alanine ⁿ	E												1
	Emamectin	1 (4)			/	S				NN V		NA 20		
INsB			NA	NA 2 ¹	0		NA	NA	20		NA		1	NA
Cod	e.													
"ш	' = Metabolites; "\" = <lod a<="" td=""><td>nd "NA"= Not a</td><td>pplicable; O</td><td>rganophosphat</td><td>tes, OPs</td><td>; acaricides, AC</td><td>s; fungicides</td><td>, FUs and ir</td><td>secticides</td><td>: of biological orig</td><td>çin, INsB.</td><td></td><td></td><td></td></lod>	nd "NA"= Not a	pplicable; O	rganophosphat	tes, OPs	; acaricides, AC	s; fungicides	, FUs and ir	secticides	: of biological orig	çin, INsB.			
MRI	Ls = Maximum residue level v	values set by EU	for the cori	responding pes	sticides .	and their metab	olites in veε	getables.						
Org.	anophosphates, OPs such as	phorate and or	ganochlorin	es, OCs such a:	as p, p'-C	DT; 0, p'-DDT; I	o, p'-DDD; p,	. p'-DDE; α-	Endosulfa	n; β-Endosulfan;	α-HCH and γ	-HCH were below	the LOD in	lla n

Table 3.1 Comparison of mean concentration of pesticides detected in vegetables and EU MRLs (µg/kg). The concentrations below the LODs (\) were not included when calculating

Pesticides group is based on Roberts and Reigart (2013).

samples and not included in the table.

* Significant at p < 0.05</p>

The concentration of pesticides and their metabolites with respect to the farming group, conventional farming and IPM, is shown in Table 3.2 and Figure S3.2. The total average concentration of pesticides in the vegetables from conventional farming was greater than the pesticides found on the vegetables farmed using IPM. The average concentration of few pesticide residues in vegetable samples from the two different farming systems differed significantly (p<0.05). In particular, the concentrations of imidacloprid, chlorpyrifos and 3,5,6-trichloro-2-pyridinol from conventional farming were significantly higher (p<0.05) than the residues detected in the vegetables from the IPM fields. However, concentrations of other pesticides did not vary based on farming techniques, which might be due to the small sample size.

Pesticides and metabolites	Farming group	Ν	Mean	Std.	Std. Error Mean	p-value
			concentration	Deviation		
3,5,6-trichloro-2-pyridinol ^m	IPM	3	3.05	0.31	0.18	0.014*
	Conventional	23	26	27.8	5.81	
Chlorpyrifos	IPM	13	3.62	2.63	0.73	0.012*
	Conventional	50	84.2	274	38.8	
Dichlorvos	IPM	1	NA	NA	NA	0.221
	Conventional	2	1.89	0.76	0.54	
Dimethoate	IPM	2	1.41	0.33	0.23	NA
	Conventional	0	NA	NA	NA	
Omethoate ^m	IPM	1	NA	NA	NA	0.317
	Conventional	1	NA	NA	NA	
Profenofos	IPM	4	25	42	21	0.906
	Conventional	27	118	321	61.9	
Quinalphos	IPM	2	2.54	0.13	0.1	1
	Conventional	3	3.21	2.47	1.43	
Triazophos	IPM	4	4.11	3.82	1.91	0.571
	Conventional	16	62.4	176	44	
Chlorantraniliprole	IPM	0	NA	NA	NA	NA
	Conventional	5	8.81	13.9	6.21	
Imidacloprid	IPM	8	8.59	18.1	6.41	0.005*
	Conventional	20	57.7	103	23	
Carbendazim	IPM	16	21.2	32.4	8.09	0.232
	Conventional	59	36	58.8	7.66	
Metalaxyl	IPM	7	21.4	30.7	11.6	0.305
	Conventional	25	30.5	88.2	17.6	
N-(2,6-dimethylphenyl)-N-	IPM	2	1.4	0.11	0.08	0.221
(methoxyacetyl)alanine ^m	Conventional	1	NA	NA	NA	
Emamectin	IPM	0	NA	NA	NA	NA
	Conventional	1	NA	NA	NA	
IPM ^a		19	39.5	54.2	12.4	0.008*
Conventional ^a		67	196	451	55.2	

Table 3.2 Statistics of pesticides detected in vegetables from the two different farming groups (μ g/kg). Only the positive samples were included in the calculation.

Code.

^a Total average concentration of pesticides; * Significant at p < 0.05; "NA" = Not applicable.

Overall, in only 2% of the vegetable samples no pesticide residues were detected. The number of residues of different pesticides and metabolites in tomato samples was higher than in chilli and eggplant samples (Figure 3.3). Vegetable samples from conventional farming were contaminated with up to 7 residues of pesticides.



Figure 3.3 The multiple residues of pesticides detected in vegetables from the two different farming groups.

3.3.2 Hazard characterisation of the detected pesticides

Mostly, aHQ and cHQ for single pesticide were quite low, implying no risk. The HQs of dietary risks from vegetables contaminated with chemical pesticides for adolescents and adults is presented in Table 3.3a-3.3c. The consumption of tomato was found to imply a short-term risk (for both adolescent and adult individuals) due to triazophos and chlorpyrifos residues (Table 3.3b). Of all the pesticides and vegetables tested, consumption of tomatoes could carry the highest short-term risks for both adolescents and adults due to the high amounts of triazophos and chlorpyrifos found in these vegetables.

Table	3.3a HQ and	' HI for dete	cted pestic	cides in egg	Iplant for	the consumer i	group of adoles	cents and adul	ts.					
					A	dolescent							Adult	
Group	Besticides	ARfD	ADI	ESTI	аНQ	ĒD	сНД	£	ESTI	аНО	EDI	сНQ	동	Source (ARfD, ADI)
		(mg kg-1	(mg kg-1	(mg kg-1		(mg kg-1 b.w.			(mgkg-1		(mg kg-1 b.w.			
		b.w.day-1)	b.w.day-	b.w.day-		day-1)			b.w.day-		day-1)			
			1)	1)					1)					
OPs	Triazophos	1.00E-03	1.00E-03	1.99E-04	1.99E-01	1.64E-07 (LB)	1.64E-04 (LB)	8.44E-04 (LB)	2.06E-04	2.06E-01	1.70E-07 (LB)	1.70E-04 (LB)	8.71E-04 (LB)	JMPR
						2.92E-07 (UB)	2.92E-04 (UB)	1.07E-03 (UB)			3.02E-07 (UB)	3.02E-04 (UB)	1.10E-03 (UB)	
	Chlorpyrifos	5.00E-03	1.00E-03	3.54E-04	7.08E-02	6.78E-07 (LB)	6.78E-04 (LB)		3.65E-04	7.31E-02	7.00E-07 (LB)	7.00E-04 (LB)		EFSA
						7.67E-07 (UB)	7.67E-04 (UB)				7.92E-07 (UB)	7.92E-04 (UB)		
	Profenofos	1.00E+00	3.00E-02	5.77E-05	5.77E-05	4.10E-08 (LB)	1.37E-06 (LB)		5.95E-05	5.95E-05	4.23E-08 (LB)	1.41E-06 (LB)		JMPR
						1.85E-07 (UB)	6.18E-06 (UB)				1.91E-07 (UB)	6.38E-06 (UB)		
ACs	Imidacloprid	8.00E-02	6.00E-02	5.02E-05	6.28E-04	7.23E-08 (LB)	1.21E-06 (LB)	NC (LB)	5.19E-05	6.48E-04	7.47E-08 (LB)	1.24E-06 (LB)	NC (LB)	EFSA
						2.00E-07 (UB)	3.34E-06 (UB)	NC (UB)			2.07E-07 (UB)	3.44E-06 (UB)	NC (UB)	
FUs	Carbendazim	2:00E-02	2.00E-02	1.20E-03	6.02E-02	2.30E-06 (LB)	1.15E-04 (LB)	NC (LB)	1.24E-03	6.21E-02	2.37E-06 (LB)	1.18E-04 (LB)	NC (LB)	EFSA
						2.33E-06 (UB)	1.16E-04 (UB)	NC(UB)			2.40E-06 (UB)	1.20E-04 (UB)	NC (UB)	
	Metalaxyl	5.00E-01	8.00E-02	2.59E-05	5.17E-05	3.09E-08 (LB)	3.86E-07 (LB)		2.67E-05	5.34E-05	3.19E-08 (LB)	3.99E-07 (LB)		EU
						1.64E-07 (UB)	2.05E-06 (UB)				1.70E-07 (UB)	2.12E-06 (UB)		
INsB	Emamectin	1.00E-02	5.00E-04	1.95E-05	1.95E-03	1.39E-08 (LB)	2.78E-05 (LB)	NC (LB)	2.02E-05	2.02E-03	1.43E-08 (LB)	2.87E-05 (LB)	NC (LB)	EFSA
						1.58E-07 (UB)	3.17E-04 (UB)	NC (UB)			1.63E-07 (UB)	3.27E-04 (UB)	NC (UB)	

•	-	
>	-	
- 5	ŝ	1
2	2	
- 7	5	i
	-	1
c	,	
4		
ċ	-	
2		
- 9	٤	ļ
		J
٠.	i	1
ē	١	í
-	5	
0	1	Ì
-	5	
•	ć	1
7	-	1
c	-	
*	2	
. (1
- 2		
_ 2	2	
-	5	
	-	
۰.	2	
- 5	-	
7		i
- 2	-	
5		
•	٢	i
2	1	
~	-	
- 3	-	
- 2	2	
ā	ĩ	í
2	4	ł
2	2	
•	2	1
č	1	i
	-	
۰.	١	1
	2	
-	-	
	-	
5		
	-	
	2	
s	2	
4		
4+		
4+0		
4 + tr		
ant for		
the for		
of toolo		
of tooloon		
of tooloon		
of turlung		
of taplant		
of tup lant fo		
n occupant fo		
in accordant fo		
- in occuration		
the according to		
tor in occupant fo		
dar in acculant fo		
idac in acculant fo		
cidor in occulant fo		
Hickdor in occulant for		
-ticidoc in occulant fo		
oticidor in occulout fo		
actividat in acculant fo		
pocticidor in occulant fo		
I pocticidor in occulont fo		
d porticidor in occulant fo		
ad pacticidae in acculant fo		
tod posticidos in occulant fo		
-tod norticidor in occulant fo		
acted porticides in paralant fo		
acted norticidar in acculant fo		
totad porticidar in acculant fo		
otoctod porticidor in paralant fo		
Antochod marticidar in acculant for		
dotoctod porticidor in occulant fo		
r dotoctod porticidor in occulant fo		
or dotoctod norticidar in againant fo		
for dotothad porticidar in acculant fo		
for dotoctod porticidor in occulant fo		
Il for dotortod porticidor in occulant fo		
Ul for dotoctod porticidor in pagelout fo		
Ul for dotoctod porticidor in page fo		
d LII for dotortod porticidor in occulant fo		
ad UI for dotoctod porticidar in acculant fo		
ad UI for dotoctod porticidor in occulant fo		
and Ul for dotothed porticider in page for		
and Ulfor detected meeticider in eccelent for		
D and UI for detected merticides in each for		
O and UI for detected merticides in eccentrat for		
TO and Ill for detected meeticides in secondary for		

				Adolescent					Adult					
	Pesticides	ARfD	ADI	ESTI	аНQ	EDI	сНQ	cHI	ESTI	aHQ [ĪĢ	cHQ	cHI	Source
Groups		(mg kg ⁻¹	(mg kg ¹											(arfd, adi)
		b.w.day	b.w.day	(mg kg ⁻¹		(mg kg ^{.1} b.w.			(mg kg ⁻¹	~	mg kg ⁻¹ b.w.			
		1)	¹)	b.w. day ⁻¹)		day ⁻¹)			b.w.day	J	lay ⁻¹)			
									¹)					
OPs	Omethoate	2.00E-03	3.00E-04	1.31E-04	6.54E-02	1.04E-06 (LB)	3.48E-03 (LB)	1.41E-01 (LB)	1.35E-04	6.75E-02	1.08E-06 (LB)	3.59E-03 (LB)	1.45E-01 (LB)	EFSA
						1.94E-06 (UB)	6.47E-03 (UB)	1.45E-01 (UB)			2.00E-06 (UB)	6.68E-03 (UB)	1.50E-01 (UB)	
	Dimethoate	1.00E-02	1.00E-03	7.69E-06	7.69E-04	4.91E-08 (LB)	4.91E-05 (LB)		7.94E-06	7.94E-04	5.07E-08 (LB)	5.07E-05 (LB)		EFSA
						9.77E-07 (UB)	9.77E-04 (UB)				1.01E-06 (UB)	1.01E-03 (UB)		
	Triazophos	1.00E-03	1.00E-03	3.21E-03	3.21E+00	2.92E-05 (LB)	2.92E-02 (LB)		3.31E-03	3.31E+00	3.01E-05 (LB)	3.01E-02 (LB)		JMPR
						2.97E-05 (UB)	2.97E-02 (UB)				3.07E-05 (UB)	3.07E-02 (UB)		
	Chlorpyrifos	5.00E-03	1.00E-03	8.31E-03	1.66E+00	1.05E-04 (LB)	1.05E-01 (LB)		8.57E-03	1.71E+00	1.09E-04 (LB)	1.09E-01 (LB)		EFSA
						1.05E-04 (UB)	1.05E-01 (UB)				1.09E-04 (UB)	1.09E-01 (UB)		
	Profenofos	1.00E+00	3.00E-02	7.69E-03	7.69E-03	8.82E-05 (LB)	2.94E-03 (LB)		7.94E-03	7.94E-03 5	9.10E-05 (LB)	3.03E-03 (LB)		JMPR
						8.85E-05 (UB)	2.95E-03 (UB)			5,	9.13E-05 (UB)	3.04E-03 (UB)		
ACs	Imidacloprid	8.00E-02	6.00E-02	1.20E-04	1.51E-03	2.18E-06 (LB)	3.63E-05 (LB)	NC (LB)	1.24E-04	1.55E-03	2.25E-06 (LB)	3.75E-05 (LB)	NC (LB)	EFSA
						2.96E-06 (UB)	4.93E-05 (UB)	NC (UB)			3.05E-06 (UB)	5.09E-05 (UB)	NC (UB)	
	Chlorantraniliprole	NA S	1.56E+00	1.56E-04	NC	1.32E-06 (LB)	8.45E-07 (LB)		1.61E-04	NA	1.36E-06 (LB)	8.72E-07 (LB)		EFSA
						2.13E-06 (UB)	1.36E-06 (UB)			, N	2.20E-06 (UB)	1.41E-06 (UB)		
FUs	Carbendazim	2.00E-02	2.00E-02	1.58E-03	7.90E-02	4.82E-05 (LB)	2.41E-03 (LB)	NC (LB)	1.63E-03	8.15E-02 4	1.98E-05 (LB)	2.49E-03 (LB)	NC (LB)	EFSA
						NA (UB)	NA (UB)	NC (UB)		-	VA (UB)	NA (UB)	NC (UB)	
	Metalaxyl	5.00E-01	8.00E-02	4.15E-04	8.31E-04	1.23E-05 (LB)	1.54E-04 (LB)		4.29E-04	8.57E-04	L.27E-05 (LB)	1.59E-04 (LB)		EU
						1.26E-05 (UB)	1.58E-04 (UB)				1.30E-05 (UB)	1.63E-04 (UB)		

Table 3.3b HQ and HI for detected pesticides in tomato for the consumer group of adolescents and adults.

Ľ£,
- 5
Ø
0
σ
5
0
ъ
2
, ju
2
نە_
6
Ø
_O
5
0
9
2
5
0
5
ų
5
2
ž
5
G
e
£
÷
.ō
4
1
Ē
G
2
(0
ຽ
ides
cides
sticides
esticides
pesticides
d pesticides
ed pesticides
cted pesticides
ected pesticides
stected pesticides
detected pesticides
r detected pesticides
or detected pesticides
I for detected pesticides
HI for detected pesticides
d HI for detected pesticides
nd HI for detected pesticides
and HI for detected pesticides
2 and HI for detected pesticides
HQ and HI for detected pesticides
: HQ and HI for detected pesticides
3c HQ and HI for detected pesticides
3c HQ and HI for detected pesticides
3.3c HQ and HI for detected pesticides
le 3.3c HQ and HI for detected pesticides
ble 3.3c HQ and HI for detected pesticides
able 3.3c HQ and HI for detected pesticides

				Adol	escent					Adult				
Groups	Pesticides	ARfD	ADI	ESTI a	q	EDI	cHQ	문	ESTI	aHQ	EDI	сНQ	сH	Source
														(ARfD, ADI)
		(mg kg-1	(mg kg-1	(mg kg-1		(mg kg-1 b.w.			(mg kg-1 b.w		(mg kg-1 b.w.			
		b.w. day-	b.w.day-	b.w.		day-1)			day-1)		day-1)			
		1)	1)	day-1)										
OPs	Dimethoate	1.00E-02	1.00E-03	9.22E-07 9	.22E-05	1.95E-09 (LB)	1.95E-06 (LB)	1.19E-03 (LB)	9.52E-07	9.52E-05	2.01E-09 (LB)	2.01E-06 (LB)	1.23E-03 (LB)	EFSA
						4.48E-08 (UB)	4.48E-05 (UB)	1.78E-03 (UB)			4.63E-08 (UB)	4.63E-05 (UB)	1.84E-03 (UB)	
	Triazophos	1.00E-03	1.00E-03	7.49E-06 7	.49E-03	1.58E-08 (LB)	1.58E-05 (LB)		7.73E-06	7.73E-03	1.63E-08 (LB)	1.63E-05 (LB)		JMPR
						5.87E-08 (UB)	5.87E-05 (UB)				6.06E-08 (UB)	6.06E-05 (UB)		
	Chlorpyrifos	5.00E-03	1.00E-03	3.84E-04 7	.67E-02	1.02E-06 (LB)	1.02E-03 (LB)		3.96E-04	7.92E-02	1.05E-06 (LB)	1.05E-03 (LB)		EFSA
						1.03E-06 (UB)	1.03E-03 (UB)				1.06E-06 (UB)	1.06E-03 (UB)		
	Profenofos	1.00E+00	3.00E-02	1.78E-04 1	.78E-04	5.36E-07 (LB)	1.79E-05 (LB)		1.84E-04	1.84E-04	5.53E-07 (LB)	1.84E-05 (LB)		JMPR
						5.67E-07 (UB)	1.89E-05 (UB)				5.86E-07 (UB)	1.95E-05 (UB)		
	Dichlorvos	2.00E-03	8.00E-05	2.16E-06 1	.08E-03	1.08E-08 (LB)	1.35E-04 (LB)		2.23E-06	1.11E-03	1.11E-08 (LB)	1.39E-04 (LB)		EFSA
						5.04E-08 (UB)	6.30E-04 (UB)				5.20E-08 (UB)	6.50E-04 (UB)		
ACs	Imidacloprid	8.00E-02	6.00E-02	2.95E-04 3	.69E-03	1.87E-06 (LB)	3.12E-05 (LB)	NC (LB)	3.05E-04	3.81E-03	1.93E-06 (LB)	3.22E-05 (LB)	NC (LB)	EFSA
						1.89E-06 (UB)	3.15E-05 (UB)	NC (UB)			1.95E-06 (UB)	3.25E-05 (UB)	NC (UB)	
FUs	Carbendazim	2.00E-02	2.00E-02	7.42E-05 3	.71E-03	7.30E-07 (LB)	3.65E-05 (LB)	NC (LB)	7.66E-05	3.83E-03	7.54E-07 (LB)	3.77E-05 (LB)	NC (LB)	EFSA
						7.39E-07 (UB)	3.69E-05 (UB)	NC (UB)			7.62E-07 (UB)	3.81E-05 (UB)	NC (UB)	
	Metalaxyl	5.00E-01	8.00E-02	3.43E-04 6	.86E-04	8.19E-07 (LB)	1.02E-05 (LB)		3.54E-04	7.08E-04	8.46E-07 (LB)	1.06E-05 (LB)		EU
						8.51E-07 (UB)	1.06E-05 (UB)				8.78E-07 (UB)	1.10E-05 (UB)		
Code (able 3.3).													
NA: No	t available; N	JC: Not ch	aracterize	.p										
LB: Low	er bound (re	sults < LO	D were re	eplaced wit	:h 0).									

JB: Upper bound (results < LOD were replaced with the value of LOD).</p>

The aHQ and aHI values in bold represent potential dietary risk of the pesticides in adolescents and adults.

ARfD (mg kg¹ b.w. day¹) and ADI (mg kg⁻¹ b.w. day⁻¹) values were extracted from the EU pesticides database (http://ec.europa.eu/food/plant/pesticides/eu-pesticidesdatabase/public/?event=homepage&language=EN), The concentrations of organophosphates, OPs such as phorate and organochlorines, OCs such as p, p'-DDT; o, p'-DDT; p, p'-DDD; p, p'-DDE; α-Endosulfan; β-Endosulfan; α-HCH and y-HCH were below the LOD in all vegetables (100% non-quantified) and not included in the calculation.

Data on ARfD and ADI of pesticides and metabolites such as quinalphos, 3,5,6-trichloro-2-pyridinol and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine was not available and not included in the calculation, therefore their potential health risks could not be ascertained. Chlorantraniliprole has only data on ADI, therefore the acute risk was not calculated. All tomato samples contained quantifiable carbendazim residues, therefore the UB was not available.

The body weight (b.w.) for adolescents and adults was 32 and 62 kg, respectively.

As data of Nepalese large portion consumption of the vegetables was not available, for acute risk estimation, the maximum consumption (g/day) value was assumed as: 500 tomato) for adult. For adolescent, the maximum and average vegetables intake (g/day) is assumed as half of adults' consumption. The acute HI was not a priority of this eggplant), 50 (chilli), and 300 (tomato) for adult. For chronic risk estimation, the average vegetables consumption (g/day) value is 9.6 (eggplant), 2.85 (chilli), and 61.3 research as the chance of having the highest residue in a sample for multiple pesticides is very low.

Pesticides group is based on Roberts and Reigart (2013).

3.3.3 Pesticides exposure

Our study detected mostly organophosphates in the investigated vegetable samples. A few of these samples exceeded the MRLs of these compounds set by Nepal and the EU. Chlorpyrifos exceeded the EU MRL in 19% of the tomato and chilli samples (Table 3.1). Likewise, chlorpyrifos exceeded the Nepalese foodstuff MRL (50 μ g/kg) in 25% of the tomato samples and 4% of the chilli samples. Similarly, triazophos exceeded the EU MRLs in 4% of the eggplant samples and 6% of the tomato samples. The concentration of omethoate exceeded the EU MRL in 3% of the tomato samples. Carbendazim exceeded the EU MRL in 3% of the tomato samples. Carbendazim exceeded the EU MRL in 3% of the tomato samples foodstuff MRL (500 μ g/kg). It is noteworthy to mention that all of the vegetable samples that were from the IPM fields had pesticide concentrations below EU MRLs. Measurements of omethoate, chlorpyrifos, triazophos and carbendazim exceeded the EU limits leading to unnecessary exposure to pesticides (Table 3.1).

3.3.4 Risk characterisation of the detected pesticides

Of all the pesticides groups and vegetables tested, the OPs that were detected in tomatoes posed the highest health risks. The chronic HIs for OPs in adults and adolescents were <1, indicating no risk of the vegetables consumption. HIs for other groups of pesticides could not be characterized, either due to their different modes of action or not applicable to characterize (Table 3.3a-3.3c). Pesticide-related health risks and diseases are presented in Table 3.4.

3.4 Discussion

The most frequently detected pesticides in the vegetables in this study were carbendazim and chlorpyrifos. Our vegetable samples tested negative for OCs and other pesticides banned in Nepal. However, previous studies (Bempah et al., 2016; Diop et al., 2016; Mtashobya, 2017) found OCs in vegetable samples (Table S3.2). One study carried out in Nepal (Koirala et al., 2007) detected OCs in vegetables that might have been due to past applications. For some pesticides, metabolites such as omethoate, N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, and 3,5,6-trichloro-2-pyridinol were detected in higher concentrations, probably due to their persistent nature (Xu et al., 2015). However, the occurrence of these pesticides in the vegetables varied (Table 3.1). Although farmers applied dimethoate to all vegetable crops (Bhandari et al., 2018), its metabolite (omethoate) was only detected in 6% of the tomato samples which might be due to the higher water solubility of this pesticide (Table S3.1). Farmers did not report the use of quinalphos in the sampled vegetables (Bhandari et al., 2018), but it was detected in 19% of the chilli samples. This might be due to the pesticide uptake from soil or spray drifting in from neighbouring farmers' fields or cross contamination by different vectors. Farmers apply dichlorvos the most often (Bhandari et al., 2018). However, its residue was detected in only 11% of the chilli samples, which might be due to higher water solubility and volatility of the compound. Along with its higher solubility in water, its shorter half-life time as well as its rapid disintegration in nature could also explain the values (Table S3.1). Further environmental factors such as rainfall, temperature, soil properties, and soil organisms also affect the occurrence of pesticides and their metabolites in different matrices (Bento et al., 2016). The detection frequency for fungicides was higher (Figure 3.2) which might have been due to its extreme use in vegetable farming in the study area (Bhandari et al., 2018). The multiple residues (resulting from the mixtures of different pesticides) occur due to plant uptake of pesticides (a mixture) of high persistence (Zhang et al., 2015), spray drift (Coronado et al., 2011) and poor agricultural practices. Farmers mix more than one kind of pesticide and sprayed it on tomato crops believing that the mixture would be more potent and more effectively kill the target pests (Bhandari et al., 2018). Therefore, we observed the highest number of residues in tomato. Tomato and green chillies are ready-to-eat foods and are consumed in salads, often without cooking. As a result, their detection suggest immediate action needs to be taken to minimize the risk of pesticide exposure. Previous studies showed that food grown conventionally had higher concentrations of pesticides than food grown on IPM fields (Mladenova and Shtereva, 2009; Singh et al., 2009). Our study also showed that most vegetables grown conventionally had higher levels of pesticide residues than food grown using IPM methods.

Nepalese farmers used pesticides on vegetables even when there was no disease or pests present (Pant et al., 2014). In our study, omethoate, chlorpyrifos, triazophos and carbendazim in the vegetables from conventional farming exceeded the EU MRLs, which further shows the poor agricultural practices of farmers leading to unnecessary exposure of consumers to pesticides. Farmers applied higher amounts of chlorpyrifos on tomato crops (Bhandari et al., 2018) and residue analysis of this crop revealed that 19% of the samples exceeded EU MRLs. This provides strong evidence of an indiscriminate use of pesticides in tomato crops. In our study, eggplant and tomato samples tested negative for dichlorvos (below LOD). However, Rawal et al. (2012) found the concentration of this compound in vegetables above EU limits. Of all the pesticides and vegetables examined in this current study, the consumption of tomatoes could pose highest potential human risks due to exposure to insecticides such as triazophos and chlorpyrifos. Acute pesticide poisonings from chlorpyrifos were also reported by hospitals in Nepal (Gyenwali et al., 2017; Lama, 2008). A study in Pakistan (Syed et al., 2014), however, estimated no risk of

chlorpyrifos from consuming tomatoes. HQs not only depend on pesticide concentration but also on average body weight and quantity of food consumed (Javed and Usmani, 2016), all of which ultimately affect an individual's level of risk from pesticides.

OPs are routinely applied to vegetables to control insects and disease (Quijano et al., 2016). In our study, the OPs that were detected indicated the highest potential for health risks. Similar pesticide risk was observed in a previous study (Liu et al., 2016a). In our study, the pesticides such as triazophos and chlorpyrifos contributed significantly to higher cHI values for the OPs, which might be due to the overuse of these pesticides from the organophosphate group in the study area. The HI could be reduced by gradually minimizing the use of OPs. Moreover, we observed higher HQs for insecticides such as chlorpyrifos and triazophos, and fungicides such as carbendazim. These pesticides are responsible for many health problems such as endocrine disruption, reproductive disorders and neurological effects (Table 3.4). Previous studies in Nepal observed acetylcholinesterase depression in farmers applying chemical pesticides (Atreya et al., 2012; Neupane et al., 2014) and increased numbers of cancer patients in hospitals (Poudel et al., 2017). A recent study from Nepal (Yadav et al., 2016) concluded that there was a higher cancer risk associated with pesticides. Along with cardiovascular diseases, the occurrence of renal failure, Alzheimer's disease, Parkinson's disease, and strokes are also increasing in Nepal. Whether or not these diseases are linked to dietary exposure to pesticides warrants further investigation.

Table 3.4 Pesticides detu	ected in v∈	etables	with the	ir correspond	ling health iss	ues ^A .					
Pesticides and	Carcinoger	Mutager	n Endocrin	e Reproductiv	e Cholinesteras	e Neurotoxicar	nt Respiratory tract	Skin	Skin	Eye	Phototoxicant
metabolites			disrupter	· effects	inhibitor		irritant	irritant	sensitiser i	irritant	
3,5,6-trichloro-2-	na	na	na	na	na	na	Ь	na	na	na	ar
pyridinol ^m											
Carbendazim	Ь	~	Ь	7	×	×	×	×	na	×	ы
Chlorantraniliprole	×	×	×	Ь	×	×	na	×	na	۵	ы
Chlorpyrifos	×	×	Ъ	~	~	~	×	×	Р	×	×
Dichlorvos	Ь	~	Ь	na	7	7	Ь	≻	٩	~	ы
Dimethoate	Ь	×	×	Ь	~	Ь	×	×	×	~	ы
Emamectin	×	×	na	Ь	×	Ь	na	×	٩	~	ы
Imidacloprid	×	Ь	na	~	×	Ь	×	Ь	na	۵	ы
Metalaxyl	×	×	×	×	×	×	×	×	P	×	ы
N-(2,6-dimethylphenyl)-N-	na	na	na	na	na	na	na	na	na	na	ы
(methoxyacetyl)alanine ^m											
Omethoate ^m	×	na	~	×	۲	×	na	≻	~	×	ы
Profenofos	×	na	na	×	7	۲	na	≻	na	۵	ы
Quinalphos	×	na	na	Ь	۲	۲	۲	≻	na	~	ы
Triazophos	×	na	na	na	۲	۲	٢	≻	, na	7	าล
Code. Y = Yes. known to cause a p	roblem: X =	= No. knov	vn not to c	ause a problen	n: P = Possibly. :	status not ident	ified: na = Data not a	ivailable			

A Health issues were adopted from a PPDB database (https://sitem.herts.ac.uk/aeru/ppdb/en/index.htm) and BPDB (https://sitem.herts.ac.uk/aeru/bpdb/Reports/1326.htm).

OPs and their metabolites may translocate, accumulate or be deposited into vegetable tissues (Jeong et al., 2012). For compounds such as omethoate, N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, and 3,5,6-trichloro-2-pyridinol, neither CODEX, EU or EPA have set MRLs for the vegetables in this study and the WHO has no set values for ARfD or ADI which leads to difficulties in evaluating their risks on human health.

The 23 prioritized pesticides and metabolites assessed in the vegetables correspond to <20% of the active ingredients used, indicating that the total amount of pesticides in Nepalese vegetables might even be higher than presented in this study and the actual dietary risks even greater. This study was limited to small sample size consisting of only 3 types of vegetables. The levels of residues varies with seasons (Mtashobya, 2017), which might be inadequate to assess a total exposure to pesticides. Only certain pesticides sharing a common mode of action are subject to cumulative risk assessment. We used hazard indices to present the risks that pesticides in vegetables pose for human health and the dietary risk estimates were based on aHI and cHI for OPs only. Using HI as an approach for the risk assessment further limits the current risk assessment as complexity in the mixture of pesticides present in our foods and their different mode of actions has increased shortcomings on summation effects (Reffstrup et al., 2010). A balanced diet differs with respect to gender and age (DFTQC, 2012), and there is no clear information available on vegetable consumption and marketing, especially along the porous India-Nepal border. Activities such as cooking, washing, and peeling have been found to reduce pesticide residues in vegetables (Keikotlhaile et al., 2010; Kumari et al., 2003; Reiler et al., 2015; Shabeer et al., 2015). Overall, our results indicated that there is a potential for increased health risk for adults and adolescents following acute single and cumulative exposure to some pesticides used in vegetable farming in Nepal.

3.5 Conclusion

This study quantified pesticide residues in vegetables which could potentially threaten people's health in Nepal. Risks were mainly associated with the residues of OP pesticides in vegetables. The HQ and HI estimations revealed a serious potential risk for consumers. Due to multiple pesticide residues exceeding the MRLs for single residue concentrations, the consumers are exposed to pesticides. ARfD and ADI for multiple residues do not exist and therefore, this risk should urgently be studied. The consumption of some of the vegetables posed an unacceptable risk to human health.

The current study recommends the following interventions in order to reduce pesticide risk associated with the consumption of vegetables in Nepal: (i) extensive IPM training for farmers currently using pesticides, focusing on good agricultural practices such as pesticide

spraying intervals and the importance of the waiting period; (ii) legal enforcement of pesticide residue limits to prevent, control and minimize health risks and (iii) decrease in the use of OPs, especially chlorpyrifos and triazophos.

Acknowledgements

We are grateful to the Netherlands Universities Foundation for International Cooperation (Nuffic) for funding this research (through NFP-PhD-5160957247). The authors thank local farmers for their cooperation and support during the sample collection. We would like to thank the students Dinesh Ghimire, Prashant Chaudhary, Santosh Gupta, and Ramesh Lodh for their support during the field work. We also thank Ron Hoogenboom, Ad Peijnenburg, Robin Palmer, Rita Boerrigter-Eenling, Joost Memelink, and Suryati Sunarto for their support. The comments received from the anonymous reviewers were also appreciated.

Supplementary Material

	Pesticides and	MF/MW(g mol-1)	aCASRN	Water	^b DT50	κd	dKoc	Vapour	^e GUS	^f BCF
Group	metabolites			solubility				pressure	index	
				(mg L-1)				(mPa)		
OPs	3,5,6-trichloro-2- pyridinol ^m	C ₅ H ₂ Cl ₃ NO/198.43	6515-38-4	81	/5	na	149	137.32	4.88	3.20
	Chlorpyrifos	C ₉ H ₁₁ Cl ₃ NO ₃ PS/350.59	2921-88-2	1.05	28	127	5509	1.43	3.63	1374
	Dichlorvos	$C_4H_7Cl_2O_4P/220.98$	62-73-7	18000	na	na	50	2100	0.69	<100
	Dimethoate	C ₅ H ₁₂ NO ₃ PS ₂ /229.26	60-51-5	25900	7	na	na	0.247	1.01	8
	Omethoate ^m	C ₅ H ₁₂ NO ₄ PS/213.19	1113-02-6	500000	14	0.53	41.3	19	-2.38	75
	Phorate!	C7H17O2PS3/260.4	298-02-2	50	63	na	1660	112	1.25	483
	Profenofos	$C_{11}H_{15}BrClO_3PS/373.63$	41198-08-7	28	7	na	2016	2.53	0.59	1186
	Quinalphos	C ₁₂ H ₁₅ N ₂ O ₃ PS/298.3	13593-03-8	18	na	na	1465	0.346	1.1	na
	Triazophos	$C_{12}H_{16}N_3O_3PS/313.31$	24017-47-8	35	9	na	358	1.33	2.38	300
OCs	o, p'-DDT!	C14H9Cl5/354.48	789-02-6	na	na	na	151000	na	-3.89	na
	p, p'-DDD!	C ₁₄ H ₁₀ C ₁₄ /320.04	72-54-8	0.09	na	na	131000	0.18	-2.46	na
	p, p'-DDE!	C14H8Cl4/318.02	72-55-9	0.12	5000	50000	na	na	na	1800
	p, p'-DDT!	C14H9Cl5/354.49	50-29-3	0.025	na	na	151000	na	-3.89	na
	α-Endosulfan!	C9H6Cl6O3S/406.93	959-98-8	0.32	86	na	11500	8.30	-0.1	2755
	α-HCH!	C ₆ H ₆ Cl ₆ /290.83	319-84-6	2	na	na	1888	5.99	1.62	na
	β-Endosulfan!	C ₉ H ₆ Cl ₆ O ₃ S/406.93	33213-65-9	0.45	na	na	na	na	na	na
	γ-HCH!	C ₆ H ₆ Cl ₆ /290.82	58-89-9	8.52	148	na	1270	4.40	3.95	1300
ACs	Chlorantraniliprole	C ₁₈ H ₁₄ BrCl ₂ N ₅ O ₂ /483.15	500008-45-7	0.88	204	3.18	362	6.3 X 10 ⁻ ⁰⁹	4.22	15
	Imidacloprid	$C_9H_{10}CIN_5O_2/255.66$	138261-41-3	610	174	na	na	4.0 X 10 ⁻ ₀7	3.74	0.61
FUs	Carbendazim	C ₉ H ₉ N ₃ O ₂ /191.21	10605-21-7	8	22	na	na	0.09	2.53	25
	Metalaxyl	C15H21NO4/279.33	58737-19-1	8400	39	na	162	0.75	2.79	7
	N-(2,6-dimethylphenyl)- N-	C ₁₄ H ₁₉ NO ₄ /265.3	87764-37-2	na	51	na	38	na	3.83	na
	(methoxyacetyl)alanine ^m									
INsB	Emamectin	C ₅₆ H ₈₁ NO ₁₅ /1008.3	155569-91-8	24	1	na	377000	0.004	na	80

Table S3.1 The	properties o	f the	pesticides	tested	and	their	metabolites	۸.
TUDIC 33.1 THE	properties	june	pesticiaes	lesteu	unu	uncin	metabomes	•

Code.

aCASRN, chemical abstracts service registry number; bDT50 (field), half-life time; cKd, soil sorption coefficient; dKoc, soil organic carbon-water partitioning coefficient; eGUS index, groundwater ubiquity score; fBCF, bio-concentration factor.

* = The properties of the pesticides were taken from PPDB (https://sitem.herts.ac.uk/aeru/ppdb/en/atoz.htm) and BPDB

(https://sitem.herts.ac.uk/aeru/bpdb/atoz.htm) (Retrieved on: 30 March, 2018).

MF = Molecular formula; MW = Molecular weight; ! = Banned pesticides; na = Not available; m = Metabolites.

Organophosphates, OPs; organochlorines, OCs; acaricides, ACs; fungicides, FUs and insecticides of biological origin, INsB.

Place country Destinides						References		
Flace, country	resticides	Eggnlant	No. of	Chilli	No of	Tomato	No. of	- Nelelences
		Egghiant	samplos	CIIIII	samplos	TOITIato	samples	
Hudorahad	Chlorpyrifes	24	10	22	samples	170	10	(Sinha at al. 2012)
India (m)	Triazonhos	24	10	lia	IId	2.01	10	(Sillia et al., 2012)
Andaman	Endoculfon	0.80	n 2	02	12	5.01	n 2	(Swarnam and
Islands India	Endosunan	na	IId	92	42	IId	IId	(Swarnann anu Volmurugan, 2012)
(m)								veimurugan, 2015)
(III) Croatia (h)	Dimothoata	100	20	50	53	22	64	(Knožović ot al. 2012)
Croatia (II)	Endoculfan	100	20	50	52	10	04	(KIIEZEVIC EL al., 2012)
Pogota	Carbondazim	na	-	na		40	400	(Ariac at al. 2014)
Dogota,	Dimothoato	lid	IId	lia	IId	20	400	(Allas et al., 2014)
	Diffectionate					20		
	Motoloxyd					10		
Turkey (n)	Carbondarin			11 210	02	10	177	(Delvinei et el 2014)
Turkey (I)	Chlorpyrinhos	lld	lld	11-210	00	20-1200	1//	(Dakirci et al., 2014)
	Childrey (av. 0)			10-400		10-55		
	Endosunan (d, p)			(230-430, 300-		lld		
	Imidacloprid			10 1240		12 00		
	Motologia			10-1240		12-88		
Equat (b)	Carbondarin			50-160	21	10-50	10	(Cad Alla at al
Egypt (n)	Carpendazim	na	na	210	31	70	19	(Gaŭ Alla et al.,
	Motology			310		50		2015)
	Dreferefee			10		10		
Chana (r)	Froienoios	22	-	4300	60	410	50	(Rompohietal 2016)
Griana (r)	enuosunan (u, p)	lld	lld	15-17, 0-10	60	15-19, 20-20	50	(Beilipali et al., 2010)
	p,p -DDD			9-13		25-31		
	p,p-DDE			8-21		20-24		
Nieze				20-26		23-29	F 7	(Diametral 2010)
Niaga,	2DD15 Chlorenuifee	na	na	na	na	15	57	(Diop et al., 2016)
Senegal (m)	Dimetheete					135		
Kunnait (r)	Dimetrioate	nd 00	14	nd 10	12	18 nd 510	16	(Jallow at al. 2017b)
Kuwait (I)	Motoloxyd	nu-90	14	nd-10	12	nd 200	10	(Jallow et al., 2017b)
	Drofonofor	na		nd 20		110-200 20, 200		
Burking Face	Chlorpyrifoc	na	-	na-30		20-390	17	/Lohmann at al
burkina rasu	Imidaeloprid	lld	lld	lld	lld	152	17	(Leilindiin et di.,
(11)	Drofonofor					155		2017)
	Triananhaa					74		
Algoria (r)	Motoloxyd	n 2	n 2	22	22	0.49	10	(Mobdoup of al
Algeria (I)	wetalaxyi	na	IId	IId	IId	4-412	10	(Mebuoua et al.,
Hluguru	a-HCH	na	na	na	n 2	2 20E-02	10	(Mtashohya, 2017)
Tanzania (m)	ß-endosulfan	IId	na	IId	na	0.10	15	(10103110098, 2017)
Tanzania (III)						1 70F-02		
Moru district	Chlorovrifos	na	na	na	n 2	833_603600	50	(Kariathi et al. 2016)
Tanzania (r)	chiorpymos	nu	na	na	nu	0000000	50	(Ranatin et al., 2010)
Pakistan (m r)	Chlorovrifos	250 (~90_570)	25	160(~00_270)	25	220(110_360)	25	(Latif et al. 2011)
r unstan (m, r)	a-Endosulfan	640 (<150-1570)	25	660(<150-1330)	25	300(<150-	25	(2011) 2011)
	a Endosunan	040 (<150 1570)		000(130 1330)		580)		
	B-Endosulfan	290 (<230-730)		1260(<230-1610)		330(<230-		
	F 2.1.600 and 1			1010/		870)		
Nepal (m)	Dichlorvos	6	3	4.20	3	11.4	3	(Rawal et al., 2012)
Rupandehi	OPS OCS ACS	- Table 3.1	-		-		2	This study
Nepal	FUs, and INsB							,,,,,,

Table S3.2 Comparison of pesticides level in vegetables in this study with previous studies across the world (µg/kg).

na = Not available and nd = Not detected.



Figure S3.1 Cropping calendar of vegetables.

Extraction and clean-up methodology

20g of each vegetable sample was made into a slurry using a mortar and pestle. Matrix extracts were based on QuEChERS (Quick, Easy, Cheap, Effective, Rugged and Safe) method (Anastassiades and Lehotay, 2003), according to which, 5g of a homogenised sample was extracted with 10mL acetonitrile containing 1% acetic acid in a 50 mL tube. The tubes were shaken in a mechanical shaker (end-over-end) for 30 min. Next, a salt mixture consisting of 1g sodium acetate and 4g magnesium sulphate were added to each tube and immediately vortexed for 30 seconds to induce phase separation and partitioning. The tubes were then centrifuged at 3500 rpm for 10 minutes and resulted in the formation of a clear liquid-liquid partitioning layer. The extract at the top layer was transferred into a 12mL tube for storage. For LC analysis, 250µL crude extract and 250µL water were pipetted into a vial with an integrated filter. The filter cap was placed onto each vial and the vials were vortexed. The cap was pushed through the liquid in the vial and the liquid was analysed using LC-MS/MS. For GC analysis, the extract was cleaned using dispersive SPE by pipetting 25µL PCB 198 (internal standard), 250µL extract and 250µL ACN into an eppendorf tube containing 50 mg primary secondary amine (PSA) and 150 mg MgSO₄. The dispersive SPE with PSA removes many polar compounds such as lipids and chlorophyll in vegetable extracts that might help to get rid of peak interferences (Meimaridou et al., 2013). After vortexing for 30 seconds and centrifuging for 5 minutes, 150µL of the upper layer extract was pipetted into a GC-vial containing an insert and then analysed using GC-MS/MS (details of chemicals and reagents and tools and equipment are provided in Table S3.3 and Table S3.4 respectively).

S/N	Chemicals and reagents, purity (%)	Remarks
1	Pure water	Milli- $Q^{\mathbb{C}}$ installation and with a minimum resistance of
		(Millipore Burlington, MA, USA)
2	Acetonitrile ULC-MS	UN 1648, Art. nr. 801022802 (Actu-All Chemicals b.v.,
		Oss, The Netherlands)
3	Acetic acid (HAc), 96%	EMSURE 1.00062.1000 (Merck KGaA, Darmstadt, Germany)
4	Magnesium sulphate (MgSO ₄),	291184P, GPR RECTAPUR (VWR International bvba,
	98.9%	Leuven, Belgium)
	Sodium acetate (C ₂ H ₃ NaO ₂)	EMSURE 1.06268.1000 (Merck KGaA, Darmstadt,
		Germany)
5	DisQuE [™] extraction 2ml tube	186004572 (Waters, Ireland)
	containing 150 MgSO4 and 50mg	
	PSA	
6	¹³ C-Caffeine, 99%	588598 (Sigma-Aldrich, Zwijndrecht, The Netherlands)
7	PCB 198	LGC Standards, Wesel, Germany
8	Formic acid, > 96%	251364 (Sigma-Aldrich, Zwijndrecht, The Netherlands)
9	Methanol, Ultra LC-MS	UN 1230, Art. nr. 813013802 (Actu-All Chemicals b.v.,
		Oss, The Netherlands)
10	Ammonium formate, >99.99%	516961 (Sigma-Aldrich, Zwijndrecht, The Netherlands)
11	Reference standards	Sigma-Aldrich (Zwijndrecht, Netherlands) & LGC
		Standards (Wesel, Germany)

Table S3.3 Chemicals and reagents.

Table S3.4 Tools and equipment.

S/N	Equipment	Remarks
1	Centrifuge (50 mL PP tubes)	SN 42251949, Thermo Electron LED GmbH, Langenselbold, Germany
2	Centrifuge (eppendorf epps)	SN 41693362, Thermo Electron LED GmbH, Langenselbold, Germany
3	Head over head shaker	Reax 2, Heidolph Instruments, Schwabach, Germany
4	LC-MS/MS system	A Waters TQS MS linked to a Waters Acquity UPLC system
		(Waters, Millford, MA, USA)
5	UPLC column	Waters Acquity HSS T3-C18 column (2.1 X 100 mm, 1.7 μ M particles);
		Waters part no. 186003539
_		
6	GC-MS/MS system	Agilent 7890B GC coupled to a Agilent 7010B MS system
		(Agilent, Santa Clara, CA, USA)
7	GC column	Restek ClPesticides (30m x 0.25 mm, 25 μ M film thickness)
		(Restek , Bellefonte, PA, USA)

Pesticide analytics

Pesticides were analysed using LC-MS/MS (parameters see Table S3.5) and GC-MS/MS (Table S3.6). GC-MS/MS was used for the more a-polar compounds that were difficult to analyse by LC-MS/MS. In the LC-MS/MS measurement positive ionisation was used, except for 3,5,6-trichloro-2-pyridinol which was analysed using negative mode.

LC flowrate was 0.4 ml/min, injection volume was 5 μ L. Eluent A consisted of water, eluent B of MeOH:H₂O 95:5, both eluents contained 5 mM ammonium formate and 0.1 % formic acid. A gradient was used, starting at 100% A, after 1 min. %B was increased to 45% in 1.5 min., then in 6 minutes to 100%B, which was held for 2.5 minutes. In 0.5 minutes the gradient was returned to 100% A and the column was equilibrated for 2 min. before the next injection.

The GC injector was used in PTV solvent vent mode, injection volume was 5 μ L. Temperature program of the GC oven: start 60°C, hold for 2 min, then with 20°C/min to 150°C, with 10 °C/min to 280°C and with 25°C/min to 320°C which was held for 5 min.

Compounds	Quantification	Qualifier transition	Dwell	Cone	RT
3 5 6-trichloro-2-nyridinol	196>35/15	198>37/15	0.1	30	43
Chlorpyrifos	350>198/ 17	352>200/ 17	0.05	20	10.56
Dichlorvos	220.90 > 108.90/ 18	220.90 > 126.90/ 15	0.1	25	5.78
Dimethoate	230>125/ 20	230>199/ 10	0.025	22	4.36
Omethoate	214>155/ 17	214>183/ 10	0.025	22	3.05
Profenofos	373>96.7/ 33	373>302.9/ 18	0.05	30	9.86
Quinalphos	299>147/ 20	299>163/ 20	0.05	25	8.78
Triazophos	313.9>118.9/ 35	313.9>162/20	0.025	25	8.16
Chlorantraniliprole	482>283.9/ 10	484>285.9/10	0.025	20	7.38
Imidacloprid	256>175/ 20	256>209/ 20	0.025	30	3.99
Carbendazim	192>131.9/ 32	192>159.9/ 15	0.025	25	3.76
Metalaxyl	280>192/ 17	280>220/ 15	0.025	30	7.07
N-(2,6-dimethylphenyl)-N- (methoxyacetyl)alanine	266>160/22	266>220/ 10	0.025	40	6.43
Emamectin	886.50>126/40	886.5>158/37	0.05	40	10.24

Table S3.5 Compound dependent LC-MS/MS parameters.

Table S3.6 Compound dependent GC-MS/MS parameters, top line was used for augntification for each compound

Compounds	Precursor Ion	Product Ion	RT	CE
o,p'-DDT	237	165	16.77	28
o,p'-DDT	235	165	16.77	28
p,p'- DDD (TDE)	165	17.07	28	28
p,p'- DDD (TDE)	165	17.07	28	28
p,p'-DDE	248	176	16.01	34
p,p'-DDE	246	176	16.01	34
p,p'-DDT	237	165	17.58	28
p,p'-DDT	235	165	17.58	28
Phorate	260	231	12.02	15
Phorate	260	75	12.02	2
α-Endosulfan	206	16.06	16	16
α-Endosulfan	160	16.06	8	8
α-HCH	183	12.07	6	6
α-HCH	145	12.07	16	16
β-Endosulfan	206	17.31	16	16
β-Endosulfan	160	17.31	8	8
ү НСН	183	12.71	6	6
ү НСН	145	12.71	16	16



Figure S3.2 (continued on next page)



Figure S3.2 (continued on next page)


metabolites in vegetables with respect to different farming groups (error bars: 95% Cl). Only measurements \geq LOD were considered in the graph.



4. Concentration and distribution of pesticide residues in soil: Non-dietary human health risk assessment

Soil contamination by pesticide residues is a primary concern because of the high soil persistence of pesticides and their toxicity to humans. We investigated pesticide concentration and distribution at 3 soil depths in 147 soil samples from agricultural land and assessed potential health risks due to non-dietary human exposure to pesticides in Nepal. About sixty percent of the soil samples had pesticides (25% of the soil samples had single residue, 35% of the soil samples had mixtures of 2 or more residues) in 39 different pesticide combinations. Pesticide residues were found more frequently in topsoil. Overall, the concentration of pesticides ranged from 1.0 μ g kg-1 to 251 μ g kg-1, with a mean of 16 μg kg-1. The concentration of the primary group, organophosphates (OPs), ranged from 1.23 μ g kg-1 to 239 μ g kg-1, with a mean of 23 μ g kg-1. Chlorpyrifos and 3,5,6-trichloro-2-pyridinol (TCP) were the predominant contaminants in soils. The ionic ratio of DDT and its degradation products suggested a continuing use of DDT in the area. Human health risk assessment of the observed pesticides in soil suggested negligible cancer risks and nealigible non-cancer risks based on ingestion as the primary route of exposure. The predicted environmental concentrations (PECs) of pesticides were higher than the values found in the guidance for soil contamination used internationally. Low concentrations of residues in the soils from agricultural farms practicing integrated pest management (IPM) suggest that this farming system could reduce soil pollution in Nepal.

Based on:

Bhandari, G., Atreya, K., Scheepers, P.T.J., Geissen, V., 2020. Concentration and distribution of pesticide residues in soil: Non-dietary human health risk assessment. Chemosphere 253, https://doi.org/10.1016/j.chemosphere.2020.126594

4.1 Introduction

Chemical pesticides have been used in agriculture for decades in the effort to reduce crop loss and to meet the world's growing food demands. About one-third of agricultural commodities are produced using chemical pesticides (Zhang et al., 2011). If farmers worldwide all of a sudden stopped using pesticides, crop losses to pests of fruits, veggies and grains would increase by 78%, 54% and 32%, respectively (Cai, 2008). Global production of pesticides increased by 11% annually, from 0.2 million tons in the 1950s to >5 million tons by 2000 (Carvalho, 2017). In 2012, on average, around 3.8 million tons of chemical pesticides were applied to agricultural land (FAO, 2020). This amounts to a value of >40 billion US dollars (Pimentel, 2009). As a consequence of pesticide use, over two million people, mainly residing in developing nations, are at an elevated health risks (Hicks, 2019). The rate of pesticide use varies across the globe, even within the same region. For instance, the average rate of pesticide use is observed highest in Asia, where 6.5 to 60 kg ha^{-1} insecticides are used (Carvalho, 2017). However, in the regions of Nepal, pesticide use is relatively low at <400 g ha⁻¹ (Sharma, 2015). One hundred and seventeen active ingredients of pesticides were registered and approved in Nepal, with annual imports of 410 tons of active ingredients consisting of 34% insecticides, 40% fungicides, 25% herbicides, 1.6% rodenticides, 0.03% biopesticides and 0.02% other botanical pesticides (PRMD, 2015).

Despite the benefits of using pesticides to improve food safety, intensive and widespread use of chemical pesticides can increase soil pollution, thereby increases environmental and health risks. Soil properties play a crucial role in the fate, behavior and dispersion of chemical pesticides (Lewis et al., 2016b) and has become the repository of pesticides used in agriculture. It adsorbs most pesticides and degradation products, which might negatively affect different food webs. Pesticides can ultimately reach to humans (Zhang et al., 2006), and are thereby subject to bio-amplification (Alamdar et al., 2014). Pesticides get washed away from soils by running water and thus find their way into water sources. Pesticides can also be emitted into the atmosphere through volatilization (Sweetman et al., 2005), which adversely affect air (Bidleman and Leone, 2004) and surface water quality (Mekonen et al., 2016). Runoff and flooding are two major pathways of movement of pesticides that may lead to unintentional diffusion and non-target contamination (Wong et al., 2017) that can ultimately negatively affect human health. The concentration of pesticides tend to increase with soil depth (Zhang et al., 2006) and thus, the concentrations found in the bottom soil layer may increase underground water pollution (Sankararamakrishnan et al., 2005). Different levels of pesticides were reported across the globe and have threatened humans and the environment (Houbraken et al., 2017; Sun et al., 2016). Along with analytical approaches, such as GC-MS and LC-MS, concentrations of pesticides are often estimated depending on their predicted environmental concentrations (PECs).

Farmers are exposed with pesticide-contaminated soils via different pathways such as dermal contact, direct ingestion, and inhalation (Li, 2018). The increasing probability of humans to develop cancer and non-cancer diseases as a result of such exposure was calculated by adopting USEPA standard models. Indices such as the hazard quotient (HQ) and the hazard index (HI) are used globally for health risk assessment and are mainly based on the concentration of pesticides (Hu et al., 2014; Pan et al., 2018; Sun et al., 2016). Likewise, risk of pesticides is often estimated based on their predicted environmental concentrations (PECs) (Silva et al., 2019; Vasickova et al., 2019), however, studies on PEC in the risk assessment of pesticide use are limited in literature. Further, monitoring pesticide residues in soils and examining their potential health risks is also scanty in Nepal. The application of hazardous insecticides and fungicides is increasing in the areas of Nepal where vegetables are cultivated for markets (Atreya et al., 2011; Bhandari et al., 2018). More than 80% of the pesticides applied in Nepal are used in vegetable farming (Adhikari, 2017). Vegetable farming is becoming increasingly more popular and is a good income option for farmers in Nepal. Adolescents and adults who work in the vegetable fields can be exposed to pesticides via different pathways such as non-dietary ingestion, dermal exposure and soil particle inhalation. In other parts of the globe, there is a growing evidences of human health risk due to pesticide use in agricultural (Ritz and Yu, 2000; Samsel and Seneff, 2013; Schreinemachers, 2003; Shelton et al., 2014). The dietary risks from pesticide ingestion is at the higher end, and thus considered unacceptable (Bhandari et al., 2019). In Nepal, studies have observed an emergence of higher cancer and non-cancer risks associated with pesticide use (PPDB, 2019; Yadav et al., 2016). Hazardous pesticides that are banned for use in the EU (pesticides use status in bold, Table S4.1, Supplementary information) are still used in Nepal. In addition, a recent study (Bhandari et al., 2018) showed that persistent and toxic pesticides are frequently applied to vegetable fields. This combined with the fact that farmers in Nepal are less likely to follow safe work practices and do not have access to personal protective equipment, resulting the pesticides exposure risks even larger. This study thus conducted to analyze pesticide residues in the soil of vegetable fields, and to estimate potential health risk for humans due to non-dietary exposure to pesticides in soil.

4.2 Materials and methods

Supplementary information presents information in support of the materials and methods.

4.2.1 Study area

Gaidahawa Rural Municipality (GRM) (27° 35.429' N and 83° 19.215' E) of Nepal was selected for the study (Figure 4.1). The area and population of the municipality is 96.79 sq. km and 47,565, respectively. About eighty-one percent (approx. 7900 ha) of the land is agricultural. The mean annual rainfall is about 1400 mm. The land is flat and the soil consists mainly of silt, clay and sand. The temperature may reach 42.4°C and 8.7°C in summer and winter, respectively (GRM, 2018).



Figure 4.1 Map of the study area.

4.2.2 Farming practice and selection of pesticides

Bhandari et al. (2018) reported that farmers in the study area practice commercial vegetable farming. They grow many types of vegetables such as eggplant, chilli pepper, coriander, tomato, bean, onion, sponge gourd, pumpkin, broccoli, bitter gourd, fenugreek, cauliflower, spinach, okra, radish, cucumber, cabbage, fennel, bottle gourd, and pea. Application rate of pesticides differs by crops. A few farmers practice integrated pest management (IPM) techniques for controlling pests. IPM techniques included were the use of insect traps, animal dung and urine, ash, alcohol, and tobacco. The existing farming system is heavily dependent on the application of chemical pesticides at significantly higher than recommended doses, however IPM fields were less likely to receive high amount of chemical pesticides. The frequency of applications is also higher than recommended, indicating poor agricultural practices. Farmers use thirty litre lever-operated Knapsack sprayers. Of all the pesticides used, the general trend in application rate (kg ha⁻¹) of top five pesticides was: mancozeb > dichlorvos > chlorpyrifos > profenofos > triazophos (Bhandari et al., 2018). Farmers even apply these pesticides as cocktails.

The present study accounted commonly used following pesticide groups: 7 organophosphates (OP), 1 anthranilic diamide (AD), 1 neonicotinoid (NND), 1 benzimidazole (BD), 1 phenylamide (PA), 1 micro-organism derived (MOD) and 2 unclassified degradation products (UDP). In addition to these compounds, one organophosphate and 8 organochlorines (OC), which are banned in the EU and Nepal due to their high soil persistence and toxicity (Table S4.1, Supplementary information) were also considered. Due to analytical limits, namely poor recoveries (< 70%) as well as logistic and financial limitations, some compounds were excluded.

4.2.3 The soil samples

At the time of soil sampling, farmers had not sprayed their fields with pesticides for 7 days. Soil samples were collected following the principals laid out in the EU guidelines (Theocharopoulos et al., 2001). The municipality consists of 9 wards, which is the smallest administrative unit of government. Each ward consists of a number of villages. The soil samples were collected from randomly selected villages in the 6th and 7th wards where most farmers were involved in vegetable farming following either IPM or intensive farming practices. The pesticide groups in the areas represented the use and pollution of pesticides in Nepal. Furthermore, the soil sampling areas were the same from where vegetable samples were selected for a study (Bhandari et al., 2019). The soil samples were taken from a total of 49 farmers' standing vegetable fields at 3 depths (0-5 cm, 15-20 cm and 35-40 cm).

There were 27 samples from eggplant fields, 36 samples from chilli fields and 84 samples from tomato fields, altogether 147 soil samples were collected during the vegetable growing season in 2017. An auger was used to collect soil samples. Ten samples from each sampling field were collected and mixed thoroughly after foreign materials such as stones, leaves, pebbles, gravel and roots were removed. A composite and representative sample was then collected by quartering and compartmentalization. Furthermore, a twenty gram of sample from each sampling field was kept in a separate plastic bag with a zipper and labelled with a unique sample identity. Sterile gloves were used to prevent contamination during the whole process. Final samples were kept in the refrigerator at -20 °C until complete analysis. Collected soil samples were grouped and labelled one of two ways: i) fields where intensive/ conventional farming was practiced (n = 114) and ii) fields where IPM was practiced (n = 33).

4.2.4 Chemical determination and quality control

Soil samples, labelled as clean by Wageningen Food Safety Research and which did not contain any of the 23 tested pesticides examined (list of pesticides including LODs, see Table 4.1) were considered as blanks. The analytical determination, chemicals and reagents, instrumentation, and quality assurance of the method are described in Bhandari et al. (2019). The chemical analysis and the quality control were performed as per the European Commission guidelines (SANTE/11813/2017) (EC, 2017). In the case where different LODs of the pesticide were observed, the highest LOD of the pesticide was used as the reporting limit and was considered as the final LOD of the pesticide. The calibration curves showed linearity within the range of 70-120%, with a regression coefficient (R²) > 0.99. Relative standard deviations were <10%, indicating reliability and accuracy of the method.

4.2.5 Human health risk assessment

The selected pesticides have been found to induce either cancer or non-cancer diseases (PPDB, 2019) (Table S4.2, Supplementary information). DDT was categorized as a probable carcinogenic for humans by the International Agency for Research on Cancer (IARC). In 2015, the IARC found positive associations between exposure to DDT and diseases such as non-Hodgkin lymphoma, testicular cancer, and liver cancer (IARC, 2015). In order to assess health risks posed to humans, we adopted USEPA models that have been proven successful and adopted worldwide (Liu et al., 2013b). The threshold values of the models and the concentrations of pesticides found in our soils were used to assess human health risks (for

the threshold values used in the risk assessment, see Table S4.3, S4.4 and S4.5 in Supplementary information, page 5).

To facilitate the process, the risk assessment process for humans is divided into 3 steps: i) hazard identification (Hal), ii) exposure assessment (ExA), and iii) risk characterization (RiC). Hal involves the identification of pesticides in the study area that can have health risks to humans. The persistence and toxicity of pesticides that have risks were identified based on PPDB (2019). ExA was done for pesticides that posed potential health risks. Cancer and noncancer risks from pesticides exposure were estimated following USEPA models and hazard quotient (HQ) and hazard index (HI) indices, respectively. The relationship between pesticides concentration and the resulting incidence of impacts was based on mathematical models to determine risks to humans via different exposure pathways such as ingestion, inhalation and dermal contact. RiC incorporates information derived from Hal and ExA. It involves the estimation of health risk due to single and multiple pesticides. Total cancer risk (TCR), HQ and HI were used to characterize risks to human health.

4.2.5.1 Assessment of cancer risk (CR) for OCs

Exposure to DDTs is potentially linked with CR (Band et al., 2011; Wong et al., 2015). Considering the incidental ingestion, inhalation and dermal contact with contaminated soil, (USEPA, 2018), the chronic (average) daily non-dietary intake (CDI, mg kg⁻¹ day⁻¹) of compounds (pp'-DDT, pp'-DDE and pp'-DDD) in adolescents and adults can be estimated using the following equations.

$CDI_{ing} = \frac{C_{soil} \times EF \times ED \times IR_{ing}}{AT \times BW} \times CF$	(4.1)
$CDI_{der} = \frac{C_{soil} \times SA \times SAF \times ABS \times EF \times ED}{AT \times BW} \times CF$	(4.2)
CDI _{inh} = $\frac{C_{soil} \times EF \times ED \times IR_{inh}}{DEF \times ED \times IR_{inh}}$	(4.3)

where CDI_{ing}, CDI_{der} and CDI_{inh} are the average daily doses via ingestion, dermal contact and inhalation (mg kg⁻¹ day⁻¹), respectively.

 C_{soil} (mg kg⁻¹) = concentration of pesticides in soil

PEF×AT×BW

The details of other parameters/exposure factors such as IRing, EF, ED, BW, CF, AT, SA, SAF, ABS, IR_{inh}, and PEF are listed in Table S4.3. The incremental lifetime CR denotes the increasing possibility of humans to get cancer during their lifespan via exposure to a carcinogenic compound. In our study, the CR in adolescents and adults over their lifetime exposure to DDT and its degradation products was calculated following USEPA (2001) and Yadav et al. (2016).

$CR = CDI \times SF$	4 4	4)	
$CIC = CD1 \wedge 01$		τı	

where, SF = carcinogenicity slope factor (details in Table S4.4) CDI = estimated average chronic daily non-dietary intake (Table S4.6) If multiple carcinogenic compounds are present, the total CR (TCR) from all of the compounds and possible routes is calculated following Yadav et al. (2016).

 $TCR = \sum_{k=1}^{n} CRi....(4.5)$

where, i = different exposure pathways

In general, TCR values between 1×10^{-6} and 1×10^{-4} are considered to be acceptable, while those exceeding 1×10^{-4} are considered to constitute a lifetime carcinogenic risk to humans. A risk factor $<1 \times 10^{-6}$ is regarded as negligible or no risk (USEPA, 1989).

4.2.5.2 Assessment of non-cancer risk (NCR)

The NCR was calculated following USEPA (2019), which was also adopted in a past study (Pan et al., 2018). The NCR for the pesticides of interest for a specific exposure route can be expressed as Hazard Quotient (HQ). HQ of a pesticide is the ratio of CDI, and RfD (reference dose) of the pesticide.

 $HQ = \frac{CDI}{RfD}.$ (4.6)

The total NCR of pesticides belonging to OPs via two primary routes such as soil ingestion and dermal contact can be denoted as HI, which was estimated by following Equation 4.7.

 $HI = \sum_{k=1}^{n} HQi....(4.7)$

where, HQ_i = hazard quotient of exposure pathway i

RfD (mg kg⁻¹ day⁻¹) = daily maximum permissible concentration of OPs, including the reference doses for exposures such as ingestion (RfD_{ing}) and dermal contact (RfD_{der}). The RfD_{ing} = RfD and RfD_{der} = $RfD \times ABS_{GI}$. ABS_{GI} is the gastrointestinal absorption factor (dimensionless). Pesticides belonging to OPs have their common mode of action (Table S4.1), therefore HI was estimated for only the OPs.

The details of parameters such as RfD and ABS_{GI} used in NCR assessment are listed in Table S4.5. HQ or HI greater than one shows potential NCR, while HQ or HI \leq 1 means negligible or no risk. Our study could not estimate the potential NCR via inhalation because reference concentration (RfC) values for the pesticides of interest were not available in the PPDB (2019).

4.2.6 Predicted environmental concentration (PEC)

PEC is an indicator of the expected pesticide concentrations in soil, taking into account the default values (EFSA et al., 2017). The PEC was estimated with the default values (otherwise stated) using the Equations 4.8 to 4.12. For multiple applications of chemical pesticides, the maximum time-weighted average (TWA) concentrations for exposure days of 1, 2, 4, 7, 14, 28, 50, and 100 were estimated using a moving time-frame (MTF) Excel spreadsheet. For a given exposure period, the spreadsheet calculates the TWA concentrations for period starting times ranging from day of first application to day of last application (MTF), and scans for the highest value (EC, 2004).

$PEC_{s,ini} = \frac{A(1-f_{itn})}{100 DEPTH_{soli} bd_{soli}} \dots (4.8)$
$PEC_{s,act} = PEC_{s,ini}e^{-k_{soil}t}.$ (4.9)
$PEC_{s,twa} = PEC_{s,ini} \frac{(1 - e^{-k_{soil}t})}{k_{soil}t} $ (4.10)
$PCEC_{s,ini,n} = \frac{PEC_{s,ini,1}(1-e^{-nk_{soil}i})}{(1-e^{-k_{soil}i})}(4.11)$
$PEC_{s,act,n} = PEC_{s,int,n}e^{-k_{soil}t}.$ (4.12)
where, A = application rate (g/ha); f_{int} = fraction intercepted by crop cover; $DEPTH_{soil}$ =
depth of soil (cm); $bd_{soil} =$ bulk density of soil (g cm ⁻³);
$PEC_{initial}$ = initial PECs after one application: n = number of applications: i = application

 $PEC_{s,ini,1}$ = initial PECs after one application; n = number of applications; i = application interval (d); k_{soil} = degradation rate in soil (d⁻¹) = ln(2)/half-life

PECs of pesticides are estimated for the different farms at three different time points (after pesticide application) for each pesticide prediction: the initial PECs (immediately), the shortterm PECs (1-4 days) and the long-term PECs (7-100 days). Since the farmers in our study hadn't applied pesticides for a week prior to our first sample measurement, predicted concentrations (PEC_{s,act,7} days after pesticide application) of pesticides would be suitable for making possible comparisons with their measured environmental concentrations (MECs). Likewise, since farmers followed poor agricultural practices, we used the initial PECs of pesticides in order to compare measurements with the international guidelines of soil quality standard. All the tested pesticides had a 90% degradation time under a year, thus the background concentrations and the PECs that accumulated were not considered in this study. Human health risk (HR) was evaluated by comparing the PECs of the pesticides in soils with international guidelines of soil guality such as pesticide soil regulatory guidance values (PSRGVs), the maximum concentrations of pesticides present in soil without hampering the environmental balance (Li and Jennings, 2017) and Chinese Soil Quality Standard (GB 15618-2018) (MoEP, 2018). Past studies (Qu et al., 2015; Wang, 2007) also followed similar methods for the evaluation of the health risk.

4.2.7 Statistical analysis

The data analysis was performed using Canoco 5 and SPSS 23. To avoid underestimating soil concentrations of pesticides, only pesticide residue concentrations \geq LODs were considered in data analysis. Concentrations below the LOD were not included in the analysis (Sun et al., 2016). The normality of the data was tested by the Kolmogorov–Smirnov test. In the cases of normal distribution, a one-way ANOVA was used to compare the number of pesticide residues and the total pesticide concentrations in soil between conventional and IPM farming practices, vegetable farms (eggplant, chilli and tomato), and three depths of soil. In case of significant effects at the 95% confidence level (p < 0.05), the Bonferroni post hoc test was conducted. Principal component analysis (PCA) and Pearson correlation coefficients were used to study relationships between the pesticide concentrations and the pesticide properties.

4.3 Results

4.3.1 Number of pesticides in soil

Pesticide residues analysis in soils revealed the presence of a variety of pesticide combinations. Thirty-nine pesticide combinations were detected in soils. One single pesticide residue was detected in 25% of the soil samples, while multiple residues were present in 35% of the soil samples (Figure 4.2A). The number of residues varied significantly with soil depths, vegetable fields, and farming practices (Figure 4.2B). Pesticide residues were found less frequently in the depth 35-40 cm and more frequently in the top soil (0-5 cm) (p=0.001). A large number of pesticide residues (up to 7 residues) was detected in 2% of the tested top soils.

The number of pesticide residues detected in soils from eggplant fields was significantly lower than tomato fields (p=0.025; Figure 4.2B). Seventy-three percent of soil samples from tomato fields and half of the soil samples from chilli fields contained detectable pesticide residues.



Number of residues of different pesticides and their degradation products at different depths

Figure 4.2A Number of residues of different pesticides and their degradation products at different soil depths and the frequency (in percentage of soil samples).





Soils from IPM fields had a significantly smaller number of observed pesticide residues >LOD than the soils from conventional fields (p < 0.01; Figure 4.2B and LODs in Table 4.1). Only fifteen percent of the soil samples from the IPM fields contained detectable pesticide residues. On the other hand, about seventy-three percent of soil samples from the conventional fields contained detectable residues of pesticides.

Table 4.1 Comparing mean concentrations of pesticides detected in soils from different farming systems during the growing season ($\mu g \ kg^{-1}$). The concentrations <LODs (na) were excluded when calculating the average values. Dichlorvos, dimethoate, omethoate, phorate, α - β -endosulfan and α - γ -HCH had concentrations <LOD (i.e. <1 $\mu g \ kg^{-1}$) and do not appear in the table. Abbreviations of pesticides are shown in parenthesis. NA = not applicable.

Chemical	Pesticides and	Type of far	ming					LODs	p-value
group	degradation	IPM (N=33))		Convention	nal (N=114)		-	
	products	NPS (%)	Min-Max	Mean (SD)	NPS (%)	Min-Max	Mean (SD)		
UNC	3,5,6-trichloro-2- pyridinol ^m (TCP)	NA	NA	NA	36(32)	2.63-57.4	10.4(9.98)	2.5	NA
OPs	Chlorpyrifos (CHLPY)	NA	NA	NA	11(10)	10.5-177	40.8(49.35)	10	NA
	Profenofos	NA	NA	NA	4 (4)	1.09-3.37	1.75(1.09)	1	NA
	Quinalphos	NA	NA	NA	3(3)	1.06-2.47	1.59(0.77)	1	NA
	Triazophos (TRIZO)	NA	NA	NA	6(5)	1.05-8.12	3.28(2.73)	1	NA
OCs	o, p'-DDT!	NA	NA	NA	3(3)	1.60-4.28	2.85(1.35)	1	NA
	p, p'-DDD!	NA	NA	NA	3(3)	1.95-11.1	7.11(4.69)	1	NA
	p, p'-DDE!	NA	NA	NA	18(16)	1-13.9	3.31(3.16)	1	NA
	p, p'-DDT!	NA	NA	NA	10(9)	1.05-78.4	12.1(24.5)	1	NA
AD	Chlorantraniliprole (CHLNITR)	NA	NA	NA	35(31)	1.08-14.2	3.17(2.62)	1	NA
NND	Imidacloprid (IMDA)	2(6)	1.02-1.17	1.10 (0.11)	26 (23)	1.16-31.6	5.52(6.52)	1	0.354
BD	Carbendazim	NA	NA	NA	18(16)	1.03-6.45	2.12(1.54)	1	NA
PA	Metalaxyl (MA)	3(9)	1.22-3.80	2.19 (1.41)	12(11)	1.12-8.97	3.25(2.4)	1	0.482
UNC	N-(2,6- dimethylphenyl)- N- (methoxyacetyl)ala nine ^m (MMB)	1(3)	NA	NA	2(2)	1.11-1.56	1.34(0.32)	1	0.207
MOD	Emamectin	NA	NA	NA	1(1)	NA	NA	1	NA

Notation.

Chemical group: UNC = Unclassified; OPs = organophosphates; OCs = organochlorines; AD = Anthranilic diamide; NND = Neonicotinoid; BD = Benzimidazole; PA = Phenylamide and MOD = Micro-organism derived.

NPS = Number of positive samples; m = degradation product; ! = banned pesticides.

The concentrations of residues of dichlorvos, dimethoate, omethoate, phorate, α - β -endosulfan and α - γ -HCH were < LOD and do not appear in Figure 4.3A.



Figure 4.3A Distribution of pesticide concentrations and pesticide degradation products in soils.



75-

50

Concentration (µg/kg)

25-

9

Figure 4.3B Box plots for comparison of mean concentration of pesticide residues at different soil depths detected in different vegetable fields of two farming types and the p-values.

4.3.2 Types of pesticide residues and their combinations

Fifteen different pesticide residues (approximately 65% of the tested pesticides) were observed in the soils (Table 4.1). Residues of dichlorvos, dimethoate, omethoate, phorate, α - γ -HCH and α - β -endosulfan were below LOD (< 1 µg kg⁻¹). The six most common pesticide mixtures (for abbreviations of pesticides, see Table 4.1) detected in soils were:

- (i) CHLNITR + IMDA
- (ii) CHLNITR + CHLPY + TCP + TRIZO
- (iii) MA + TCP
- (iv) CHLNITR + p,p'-DDE + p,p'-DDT
- (v) IMDA + p,p'-DDE and
- (vi) o,p'-p,p'-DDTs + p,p'-DDE + p,p'-DDD

Chlorantraniliprole and imidacloprid residues were the most prominent with 11% and 5% of the total soil samples with detectable residue levels, respectively, and found frequently at the soil depth 15-20 cm. The most common pesticide, CHLNITR, was found in 17% of the soil samples from tomato fields. In IPM fields, most soil samples (85%) were residues free however, residues of MA, IMDA, and MA + MMB were detected.

Of all the detected pesticides in our samples, about 60% of the pesticides were nonpersistent or moderately persistent compounds. Persistent and very persistent compounds represented about 13% and 27% of the detected pesticides, respectively. Four of the compounds such as o,p'-p,p'-DDTs; p,p'-DDE and p,p'-DDD, detected in soils were degradation products of active substances that are currently banned in Nepal. Overall, eight percent of the soils contained quantifiable residues of DDT and its degradation products.

4.3.3 Hazard identification

Our study found 4 organophosphates (chlorpyrifos, profenofos, quinalphos and triazophos), 4 organochlorines (o,p'-DDT, p,p'-DDT, p,p'-DDD and p,p'-DDE), 1 anthranilic diamide (chlorantraniliprole), 1 neonicotinoid (imidacloprid), 1 benzimidazole (carbendazim), 1 phenylamide (metalaxyl), 1 micro-organism derived (emamectin) and 2 unclassified degradation products [TCP and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine] in soil samples (Table 4.1). The mean concentration of pesticides in soil samples was 16.05 μ g kg⁻¹ with a range of 1.02-251.28 μ g kg⁻¹. Total pesticide concentrations in soils differed according to the farming practices, vegetables cultivated, and soil depths as seen in Table 4.1 and Figure 4.3B. A significant correlation existed between total pesticide concentration and farming practices, vegetables cultivated, and soil depth (p < 0.05). The identified hazards are the most commonly detected pesticides in soils such as carbendazim, imidacloprid, chlorantraniliprole, p,p'-DDT, p,p'-DDE, metalaxyl, chlorpyrifos, and TCP (Table 4.1). Except chlorantraniliprole, concentrations of the other pesticides in soils showed relationship with each other, indicating positive or negative correlation (Table 4.2).

Table 4.2 Correlations between the most common pesticides and degradation products detected in soils (positively correlated in bold font).

<u> </u>		, ,							
Pesticides and	Ν	CARBE	IMDA	CHLNITR	p,p'-DDT	p,p'-DDE	MA	CHLPY	ТСР
degradation products									
CARBE	18	1							
IMDA	28	-0.53	1						
CHLNITR	35	-0.05	0.42	1					
p,p'-DDT	10	na	-1.00**	0.20	1				
p,p'-DDE	18	1.00**	0.07	0.04	0.32	1			
MA	15	-0.28	-0.97	0.91	na	na	1		
CHLPY	11	0.99*	-1.00**	-0.95	na	na	na	1	
ТСР	36	0.95**	0.08	0.67	na	0.41	0.72*	0.89**	1

Notation.

CARBE = Carbendazim, IMDA = Imidacloprid, CHLNITR = Chlorantraniliprole, MA = Metalaxyl, CHLPY = Chlorpyrifos and TCP = 3,5,6-trichloro-2-pyridinol.

"*" and "**" represented significant correlation at the levels 0.05 and 0.01 (both 2-tailed), respectively.

"N" = Number of positive soil samples, "na" = Cannot be calculated because at least one of the variables is constant.

The average concentration of the pesticides was weakly correlated with their properties such as H₂O solubility, DT50, adsorption and mobility, vapour pressure, GUS index, and BCF; while the frequency of detection of the pesticides had a significant positive correlation with GUS index (Table 4.3). Principal Component Analysis (PCA) was performed to make the results of the table more informative and easily interpretable (Figure 4.4). Except p,p'-DDE, all the DDTs made a similar contributions to the OCs and suggested a similar source of origin.

		.,		, e.e. j e.r.e,						
Parameters of pesticides	Ν	WS	DT ₅₀	Kd	Koc	VP	GUS	BCF	FREQ	AVGC
WS (mg L ⁻¹)	13	1								
DT ₅₀	11	-0.13	1							
K _d	3	-0.99	1.00*	1						
Koc	12	-0.19	-0.31	1.00**	1					
VP (mPa)	11	-0.10	0.05	1.00**	-0.18	1				
GUS index	13	0.14	0.59	-1.00**	-0.92**	0.41	1			
BCF	10	-0.26	0.65*	0.69	-0.24	-0.21	-0.46	1		
FREQ	15	0.04	0.09	-0.24	-0.39	0.54	0.59*	-0.35	1	
AVGC	15	-0.13	-0.11	-0.50	-0.11	0.09	0.12	0.39	0.06	1

Table 4.3 Correlations between the parameters, frequency and concentration of pesticides detected in soils (positively correlated in bold font).

Notation.

WS (mg L⁻¹) = Water solubility, DT_{50} = Half-life, K_d = Soil distribution Coefficients, K_{oc} = Soil adsorption Coefficient, VP (mPa) = Vapour pressure, GUS = Groundwater Ubiquity Score, BCF = Bio-concentration factor, FREQ = Frequency, and AVGC = Average concentration.

"N" = Number of pesticides corresponding to the parameters.

"*" and "**" represented significant correlation at the levels 0.05 and 0.01 (both 2-tailed), respectively.

TCP and chlorantraniliprole contributed the most to the total frequency of detection, while chlorpyrifos and p,p'-DDT had the highest pesticide concentration in soils with a maximum

concentration of 177 and 78.4 μ g kg⁻¹, respectively. These two compounds also pose the highest health risks. The compounds that had lower concentrations such as carbendazim and p,p'-DDE were comparable to those of metalaxyl and chlorantraniliprole, respectively. For all the DDTs, the concentration of p,p'-DDT (40 μ g kg⁻¹) and p,p'-DDD (11.1 μ g kg⁻¹) was higher in the depth 35-40 cm. Likewise, the concentration of p,p'-DDE (5.70 μ g kg⁻¹) and o,p'-DDT (4.28 μ g kg⁻¹) was higher in the depth 15-20 cm.



Figure 4.4 Loading plots of PCA. Loading plot of PCA (A) showed loadings of the frequency of detection (FREQ) and average concentration of detected pesticides (AVGC) in soil with the pesticide properties such as DT50-soil half-life time (days); K_d -soil distribution coefficient (mL g⁻¹); Koc-organic carbon-water partition coefficient (mL g⁻¹); Sw-H₂O solubility at 20°C (mg L⁻¹); Vp-vapor pressure at 25°C (mPa); GUS-leaching potential index; BCF-bio-concentration factor (I kg⁻¹). Loading plot of PCA (B) showed the distribution of sources of different pesticides observed in soil (N = 15).

Predicted concentrations (PEC_{s,act,7 days after pesticide application}) of pesticides in soils from tomato, eggplant and chilli farms are presented in the Supplementary information, Table S4.7. PEC_{s,act,7} of most of the pesticides were much higher than their measured environmental concentrations (MECs). Carbendazim, dichlorvos, imidacloprid and profenofos showed higher initial PECs of pesticides than their global pesticide soil regulatory guidance maximum values, indicating that farmers might be at greater risks from the pesticides (Table S4.7 and Table S4.8).

4.3.4 Exposure assessment for farm workers

The non-dietary chronic daily intake from exposure to p,p'-DDE, p,p'-DDT and p,p'-DDD in soils via non-dietary ingestion, dermal contact, and inhalation are presented in Table S4.9. Similarly, the total average cancer risks (CR) resulting from exposure to OCs are presented in Table 4.4. The CR posed by p,p'-DDE, p,p'-DDT, and p,p'-DDD in soils for adults were 1.30E-09, 4.75E-09, and 1.97E-09, respectively, which were slightly lower than those for adolescents (1.38E-09, 5.03E-09, and 2.09E-09, respectively). Likewise, the carcinogenic effects of p,p'-DDT in adults and adolescents was comparable and was higher than that of p,p'-DDE and p,p'-DDD in adolescents and adults, respectively.

pesticiae	s exposur	e în solis.						
Pesticides	CR- (adol-	CR- (adol-	CR- (adol-	CR- (adul-	CR- (adul-	CR- (adul-	TCR(adol)	TCR(adul)
	ing)	der)	inh)	ing)	der)	inh)		
p,p'-DDD	1.02E-09	1.06E-09	1.33E-13	1.13E-09	8.38E-10	1.46E-13	2.09E-09	1.97E-09
p,p'-DDE	6.75E-10	7.03E-10	8.79E-14	7.47E-10	5.54E-10	9.61E-14	1.38E-09	1.30E-09
p,p'-DDT	2.46E-09	2.56E-09	3.21E-13	2.73E-09	2.02E-09	3.51E-13	5.03E-09	4.75E-09

Table 4.4 The cancer risks (CR) of adolescents (adol) and adults (adul) resulting from the pesticides exposure in soils.

4.3.5 Risk characterization for farm workers

The estimated CR of three OCs such as p,p'-DDE, p,p'-DDT and p,p'-DDD in adolescents and adults via different pathways were found below 1×10^{-6} (Table 4.4) and showed no CR due to the exposure to the pesticides in soils. Even after considering the TCR as cumulative, the CR for adolescents and adults were below the USEPA bench mark (1×10^{-6}) .

The average HI for OPs was < 1 and posed negligible NCR. The HI (mean±SD) estimated for adolescents via dermal contact exposure routes and soil ingestion was 8.02E-05±1.04E-04 and 1.10E-04±1.43E-04, respectively. Likewise, the HI (mean±SD) estimated for adults via dermal contact exposure routes and soil ingestion was 2.95E-05±3.83E-05 and 5.69E-

05±7.38E-05, respectively. The total HIs for adolescents and adults were 1.90E-04 and 8.64E-05, respectively, showing a negligible NCR.

The non-cancer risks of the OPs to farmers via dermal contact pathways and soil ingestion are presented in Figure S4.1 and Figure S4.2. The HQ and HI (of the total OPs) were found <1 for both adolescents and adults, indicating negligible NCR.

4.4 Discussion

4.4.1 Pollution assessment

In general, top soils (0-5 cm) contained higher concentrations and numbers of pesticide residues. However, DDT and its degradation products were less frequently found in the top soils. According to the Chinese standard (GB 15618-2018), risk screening value for the total DDT is 100 μ g kg⁻¹. Likewise, Ma et al. (2016) classified soil into (i) negligible contamination, with DDT concentration <50 μ g kg⁻¹; (ii) lower contamination, with DDT concentration 50-500 μ g kg⁻¹; (iii) medium contamination, with DDT concentration 500-1000 μ g kg⁻¹, and (iv) higher contamination, with DDT concentration >1000 μ g kg⁻¹. Although DDT has been banned in Nepal since 2001, we found DDT concentration <50 μ g kg⁻¹ in 99% of the total soil samples.

Farmers applied most of the pesticides with higher application rates, greater numbers of applications and shorter application intervals (Bhandari et al., 2018). This might have contributed to higher predicted environmental concentrations (PECs) of pesticides (see Table S4.7 in Supplementary information). The PECs after 7 days of pesticide application (PEC_{s.act.7}) for most pesticides, except for dichlorvos and emamectin, were much higher than their measured concentrations (Table 4.1 and Table S4.7, Supplementary information). In the same study, they applied dimethoate and dichlorvos at higher than their recommended levels. However, soils were free from pesticide residues such as dimethoate and its degradation product (omethoate), dichlorvos, phorate, α - β -endosulfan and α - γ -HCH (<LOD). Dimethoate, omethoate and dichlorvos have higher water solubility (Table S4.1) and phorate, α - β -endosulfan and α - γ -HCH have been banned for many years which may explain why their residues were absent in the soils. Soils from conventional systems had significantly higher numbers of pesticide residues than the soils from IPM farming. In the IPM farms, about 85% of the soil samples were clean, and the remaining samples had pesticide concentrations close to their corresponding LODs (Table 4.1). In an another study, we observed higher residues of pesticides in vegetables from conventional farms than that of IPM farms (Bhandari et al., 2019).

The fate of pesticides in soils is determined by their various factors: mobility, persistence, and volatility. Furthermore, other pesticide properties such as phosphorus and nitrogen levels, organic carbon content, and soil pH affect distribution and occurrence (Gong et al., 2004; Pan et al., 2018). Few of the pesticides detected in soils have lower soil organic carbon-water partitioning coefficients (Koc) and thus, moderate leaching potential which suggests a risk of ground water pollution. The sorption of chemical pesticides was the highest for the soils with greatest OC content (Zbytniewski and Buszewski, 2002).

Conventional fields contain less OC than IPM fields and this might enhance the mobility of pesticides and could thus increase groundwater pollution (Sánchez-González et al., 2013). About ninety percent of total inhabitants in the study area drink water from tube wells adjacent to their vegetable fields (GRM, 2018) which could increase their HR. Pesticides may contaminated groundwater and make it unsuitable for drinking, which is the case for Nigeria (Sosan et al., 2008) and the Philippines (Castaneda and Bhuiyan, 1996). The mean concentrations of the pesticides in our soil samples are at lower end in comparison to other countries (Table 4.5).

Table 4.5 Comparison of pesticide levels in soils from conventional farming in this study with past studies
across the globe. To find the relevant literature, Web of Science database was considered by using the
search phrase pesticide and soil and *concentration*. Hyphens indicate that no information was
available. Pesticides concentration in μ g kg ⁻¹ . The mean concentration of the most pesticides in this study
is lower than the other studies abroad.

is lower than the other st			
	Abroad		Nepalese
Pesticides			agricultural soil
	Place, country/land use/mean	Reference	Mean concentration
	concentration		(this study)
3,5,6-trichloro-2-pyridinol	-	-	10.4
Chlorpyrifos	Okara, Pakistan/cotton, wheat/1393	(Rafique et al., 2016)	40.8
	China/persimmons and jujubes/17.15	(Liu et al., 2016a)	
	China/nuts/42.2	(Han et al., 2017)	
	Dormaa West, Ghana/cocoa/30	(Fosu-Mensah et al., 2016)	
Profenofos	Okara, Pakistan/cotton, wheat/89.79	(Rafique et al., 2016)	1.75
	Dormaa West, Ghana/cocoa/30	(Fosu-Mensah et al., 2016)	
Quinalphos	-	-	1.59
Triazophos	Okara, Pakistan/cotton, wheat/99.74	(Rafique et al., 2016)	3.28
o,p'-DDT	Nagaon, India/paddy fields, tea gardens and others/150	(Mishra et al., 2012)	2.85
	Hong Kong/different types of land use/0.05	(Zhang et al., 2006)	
	Shanghai, China/agriculture/1.66	(Jiang et al., 2009)	
	Beijing, China/school yards/42.38	(Wang et al., 2008)	
	Moldavia, Romania/forest/0.7	(Tarcau et al., 2013)	

Pesticides	Abroad	Reference	Nepalese
	Place, country/land use/mean		agricultural soil
	concentration		Mean concentration
			(this study)
p,p'-DDD	Limuru, Kenya/rural and semi	(Sun et al., 2016)	7.11
	urban areas/1.71	(1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	
	Nagaon, India/paddy fields, tea gardens and others/73	(Mishra et al., 2012)	
	Hong Kong/farmland/0.05	(Zhang et al., 2006)	
	Shanghai, China/agriculture/4.56	(Jiang et al., 2009)	
	Beijing, China/school yards/6.47	(Wang et al., 2008)	
	Moldavia, Romania/forest/1.2	(Tarcau et al., 2013)	
p,p'-DDE	Limuru, Kenya/rural and semi urban areas/0.97	(Sun et al., 2016)	3.31
	Nagaon, India/paddy fields, tea gardens and others/276	(Mishra et al., 2012)	
	Hong Kong/farmland/1.73	(Zhang et al., 2006)	
	Shanghai, China/agriculture/16.14	(Jiang et al., 2009)	
	Beijing, China/school yards/27.29	(Wang et al., 2008)	
	Moldavia, Romania/forest/10	(Tarcau et al., 2013)	
p,p'-DDT	Limuru, Kenya/rural and semi urban areas/11.76	(Sun et al., 2016)	12.1
	Nagaon, India/paddy fields, tea gardens and others/351	(Mishra et al., 2012)	
	Hong Kong/farmland/0.02	(Zhang et al., 2006)	
	Shanghai, China/agriculture/3.26	(Jiang et al., 2009)	
	Beijing, China/school yards/17.54	(Wang et al., 2008)	
	Moldavia, Romania/forest/8.1	(Tarcau et al., 2013)	
Chlorantraniliprole	-	-	3.17
Imidacloprid	Okara, Pakistan/cotton, wheat/548.7	(Rafique et al., 2016)	5.52
Carbendazim	Basrah, Iraq/agricultural soil/1259	(Raheem et al., 2017)	2.12
Metalaxyl	Spain/agricultural areas/ 3.82	(Sánchez-González et al., 2013)	3.25
N-(2,6-dimethylphenyl)-N-	-	-	1.34
(methoxyacetyl)alanine			
Emamectin	-	-	-

Depending on the date of pesticides application, the PEC_{s,act,7} were correct for only dichlorvos and emamectin. However, the PECs of other pesticides in different fields (see Table S4.7 in Supplementary information) were much higher than their MECs (measured environmental concentrations or the mean concentration) (Table 4.1) and pesticide soil regulatory guidance values (Table S4.8, Supplementary information). The farmers' pesticide use behaviours such as the application rates were self-reported and observed higher than recommended (Bhandari et al., 2018). The differences between MECs and PECs are several orders of magnitude that might be due to the estimation of PECs from the realistic worst-case scenario. Since PSRGVs might be risk-based, the values could more accurately reflect the potential environmental and health risks which are worth consideration. In addition, immediate predicted environmental concentrations (PEC_{s,act,0}) of contaminated soils and their effects on human health should not be neglected.

Our predicted environmental concentration in soil (PEC_s) for dimethoate on tomato after multiple applications was much higher than their respective values in the draft assessment report (DAR) of the European Commission (EC, 2004). The possible reason for higher PEC_s for dimethoate might be due to the fact that the pesticide was applied at level higher than its recommended dose (Bhandari et al., 2018). Likewise, the PEC_s for other pesticides from eggplant and chilli farms could not be compared due to unavailability of their DARs.

4.4.2 Source identification of DDT

DDT is a mixture of its degradation products: 15% o,p'-DDT and 85% p,p'-DDT (Zheng et al., 2009), and the half-life has been estimated >15 years in the environment. Parent DDT disintegrates to DDE and DDD, more stable compounds than their parent. The ratio o,p'-DDT/p,p'-DDT is used to differentiate dicofol from DDT. The ratio between 0.2 and 0.3 corresponds to the occurrence of technical DDT, while the ratio between 1.9 and 9.3 or higher corresponds to the presence of dicofol (Qiu et al., 2005). In our study, the ratio ranged between 0.03 and 0.17, except in one sample, signifying the application of technical DDT. One sample showed the ratio comparatively higher which corresponds with dicofol use in the area. The ratio (p,p'-DDE+p,p'-DDD)/p,p'-DDT assesses the time and degree of disintegration of p,p'-DDT in soil (Qiu et al., 2004). Ratios greater than one indicate aged mixtures, while ratios < 1 indicate fresh applications of the parent DDT in soil. In our study, the ratios ranged from 0.21 to 2.20. The ratio was less than one in two samples, specifying the ongoing use of DDT and the ratio was > 1 in another sample, indicating its historical use (Dhimal et al., 2014). The current use of DDT might be due to conventional farming and/or expansion of diseases such as malaria fever and dengue (Awasthi et al., 2017; Shah et al., 2012). Similar findings have been reported from other regions of Nepal (Yadav et al., 2017; Yadav et al., 2016). Potential source analysis indicated that DDT and related compounds mainly originated from a recently applied DDT, possibly due to: (i) the illegal entry due to the porous India-Nepal border; (ii) inadequate execution of the ban and/or (iii) application of DDT for dengue control.

DDT disintegrates to DDD from anaerobic degradations while it changes to DDE from aerobic degradations. The ratio DDD/DDE indicates whether DDT is degraded aerobically or anaerobically. In our study, the ratio DDD/DDE ranged from 0.30 to 2. The ratio was less than one in a higher number of samples, indicating higher percentages of DDE than DDD and thus, DDT was aerobically degraded. Our results were different from those in soils from China (Ma et al., 2016), where the ratio was > 1 in a higher number of samples, indicating higher percentages of DDD than DDE and thus, DDT was anaerobically degraded. The disintegration of DDT-DDE-DDD can occur directly or indirectly (Wenzel et al., 2002). The

ratios of DDE:DDT, DDD:DDE, and DDD:DDT decide dechlorination paths in soils. DDT to DDE was the major disintegration route, as the ratios were: DDE:DDT (1.73) > DDD:DDE (0.96) > DDD:DDT (0.33). These results coincide with Zhang et al. (2006), but differed with Ma et al. (2016). This differences can be explained by dissimilarities in the precipitation, temperature, humidity, soil moisture, soil texture, microbes, CEC, and OM, which affect the conversion of DDT into DDE-DDD (Aislabie et al., 2010; Chattopadhyay and Chattopadhyay, 2015).

PCA estimates the source and disintegration behaviour of pesticides (Yang et al., 2012). In our study, pesticides belonging to the groups such as organophosphates, anthranilic diamide, neonicotinoid, benzimidazole, phenylamide and unclassified degradation products were aligned together indicating similar source and degradation behaviour, while OCs were separated, suggesting different source and fate of the pesticides.

4.4.3 Carcinogenic and non-carcinogenic risk

TCR and CR via dermal and ingestion pathways of exposure to DDTs for adolescents and adults were below 1x10⁻⁶, indicating negligible cancer risk (Table 4.4). Adolescent and adult exposure to single non-carcinogenic pesticides (HQ) and multiple pesticides (HI) was < 1, suggesting no appreciable non-cancer health risk. Likewise, HIs of pesticides via ingestion and dermal exposure for adolescents and adults were also negligible (< 1) (Figure S4.1 and Figure S4.2). However, other pathways of exposure such as inhalation could still exist in Nepal and cannot be excluded for a non-cancer risk assessment. The risk via inhalation was not considered in this study because essential parameters were unavailable. Furthermore, metabolism and excretion of pesticides in humans were excluded from this study. All soil samples came from farmers' fields close to their houses thus, children may have had direct contact with these soils on a daily basis.

Overall, the cancer and non-cancer risks of pesticides for adolescents were relatively higher than those for adults. Previous studies (Landrigan and Goldman, 2011; Pan et al., 2018) also indicated relatively higher risks for children than adults. The possible reason for higher risks for adolescents might be due to their higher exposure to given doses of OCs and OPs. The soil ingestion was the main pathway of OP exposure and added to 58% and 66% of the total risks in adolescents and adults, respectively [Figure S4.1 (b) and Figure S4.2 (b), Supplementary information]. Even though the soil samples of GRM were contaminated with pesticides, a negligible health risk from the exposure to the pesticide contaminated soil was observed in this study.

This study considered the worst case scenario (only positive samples and their total average concentrations): replacing the non-detects with 0 (Yadav et al., 2016) would even further decrease the CR, HQ and HI values. However, children are more likely to unintentionally ingest significant amounts of contaminated soil because of their childish behavior such as putting contaminated hand or fingers in their mouths (Rasmussen et al., 2001). Henceforth, this study warrants further research to investigate the implications of exposure for children through all of the possible pathways.

4.4.4 Limitations and future recommendations

Pesticide residues could move from neighbouring fields via water and wind and be deposited in surrounding environments (Silva et al., 2018) and accumulate in higher concentrations on the topsoil (1-2 cm) than deeper soil (Yang et al., 2015). Future research should consider soil samples and the distribution of residues in the topmost surface layer. We used conservative risk assessment methods that are generally used for risk assessment of contaminated sites and their applications in farmland needs further research. The PECs of pesticides in soils were based on information related to the pesticide application history in our earlier study (Bhandari et al., 2018) and thus, the results may not be representative of other areas and the latest pesticides use statistics. The twenty-three prioritized pesticides and degradation products assessed in Nepalese soil correspond to <20% of the active ingredients imported for use, indicating that the total pesticides in soils might even be higher than detected in our study and the pesticide mixtures may even be more complicated. Reference concentrations (RfC) of the pesticides and degradation products for the estimation of inhalation exposure were not available thus, risks due to the exposure to pesticides could not be estimated. Worldwide PSRGVs (Table S4.8) were not calculated comprehensively in humans (Li and Jennings, 2017) and comparison of the PECs with the global values for pesticides in soil may be inadequate for the assessment of HR. Further, whether or not the PECs are reasonable to evaluate the risk of pesticides in the area compared with the PSRGVs could not be answered. Despite such limitations, we have used widely accepted models and indices for the risk assessment.

4.5 Conclusion

Pesticides applied to vegetables farming in Nepal pollutes soils. Adoption of IPM techniques could reduce pesticide pollution in soils, as this study showed a notably smaller number of pesticide residues and their minimum concentration in the soil samples collected from IPM fields, compared to conventional farming. OCs concentration were sufficiently low in most

soil samples (<LOD). However, DDTs were detected with p,p'-DDE being the predominant compound. There is no appreciable health risk from pesticides residues in soils, based on direct dermal contact and/or ingestion in adults or adolescents. The focus should be placed on DDT pollution and the recommendations from the United Nations treaty, the Stockholm Convention should be implemented. A few pesticides detected in soils have a potential of leaching thus, there is a risk of ground water pollution.

Predicted environmental concentrations (PEC_s) for most of the frequently applied pesticides used on vegetables in Nepal did not appear in the European Commission (EC) draft assessment reports thus, the estimated PECs is of minimal use. The PECs 7 (PEC_{s,act,7 days after pesticide application}) for almost all of the pesticides were much higher than their measured environmental concentrations (MECs). The initial PECs of carbendazim, dichlorvos, imidacloprid and profenofos were much higher than their guidance values in soil. The PECs scenario based on the poor agricultural practices is insufficient to claim an increasing health risk of farm workers which warrants future research on PECs and health risk from pesticides in soils from other locations.

Acknowledgements

We are thankful to the Nuffic for supporting this work through project No. PhD-5160957247. We would like to thank the farmers for their help in fields. We thank the students who supported researchers carrying out the sample collection: Dinesh Ghimire, Ramesh Lodh, Prashant Chaudhary, and Santosh Gupta. We would also like to thank Robin Palmer, Ad Peijnenburg, Ron Hoogenboom and Paul Zomer for their support of this study, reviewing the manuscript and helping with the laboratory work, respectively.

Table	S4.1 The prope	rties of the pesticides	tested an	d their degradation pro	$ducts^{\wedge}$.										
Group	Pesticides and degradation	Mode of action	Use status	Major pests controlled	MF/MW(g ^a mol ⁻¹)	CASRN	WS	^b DT50	'Log P	dKoc	٨P	°GUS index	^f HLC	gKd	hBCF
	products/type		(EU/NE)												
UNC	3,5,6-trichloro-2- pyridinol ^m /ins	NA	NA	na	C ₅ H ₂ Cl ₃ NO 6 /198.43	515-38-4	81	75	3.21	149	137.32	4.88	1.10	na	3.20
OP	Chlorpyrifos/ins	NSY with contact, inhalation and stomach action; AChE inhibitor	AP/AP	Scale; Wooly aphid; Leaf roller; Caterpillars; Worms; flies; Ants; Cockroaches; Beetles; Termites	C ₉ H ₁₁ Cl ₃ N 2 O ₃ PS/350.5 9	921-88-2	1.05	28	4.7	5509	1.43	3.63	0.478	127	1374
	Dichlorvos/ins	Respiratory; contact and stomach action, ChE inhibitor	NP/AP-	Beetles; Weevils; Borer; Flies; Mosquitoes; Cockroaches; Aphids; Thrins	C4H7Cl2O4 6 P/220.98	2-73-7	18000	na	1.9	50	2100	0.69	2.58x10 ⁻²	na	<100
	Dimethoate/ins	SYS with contact and stomach action, AChE inhibitor	AP/AP	Aphids; Thrips; Flies; Grasshoppers; Jassids; Spidermites; Leaf and plant hoppers	C ₅ H ₁₂ NO ₃ P 6 S ₂ /229.26	0-51-5	25900	٢	0.75	na	0.247	1.01	1.42x10 ⁻⁶	na	~
	Omethoate ^m /ins	SYS with contact and stomach action, ChE inhibitor	NP/NP	Aphids; Mites; Caterpillars; Beetles; Mealvbugs: Scale	C ₅ H ₁₂ NO4P 1 S/213.19	113-02-6	50000	14	-0.9	41.3	61	-2.38	4.62x10 ⁻⁹	0.53	75
	Phorate!/ins	SYS with contact and stomach action, AChE inhibitor	NP/NP	Leaf miners; Beetles; Flies; Nematodes; Mites; Rootworms	C ₇ H ₁₇ O ₂ PS 2 ₃ /260.4	.98-02-2	50	63	3.86	1660	112	1.25	5.90x10 ⁻¹	na	483
	Profenofos/ins	NSY with contact and stomach action, AChE inhibitor	NP/AP	Aphids: Bugs; Worms; Leaf webber; Whitefly; Spidermites; Caterpillars; Cotton leaf- nerforator	C ₁₁ H ₁₅ BrCl 4 0 ₃ PS/373.6 3	-1198-08-7	28	~	1.7	2016	2.53	0.59	1.65x10 ⁻³	na	1186
	Quinalphos/ins	NSY with contact and stomach action, AChE inhibitor	NP/AP	Caterpillars; Aphids; Mealybugs; Mites; Bollworms; Leaf honners: Borers	C ₁₂ H ₁₅ N ₂ O ₃ 1 PS/298.3	3593-03-8	18	na	4.44	1465	0.346	1.1	4.70x10 ⁻³	na	na
	Triazophos/ins	NSY, broad spectrum with contact and stomach action, ChE inhibitor	NP/AP-	Aphids, Thrips, Midges, Beetles, Sawflies, Apple suckers, Spidermites, Scale insects, Fruit flies	C ₁₂ H ₁₆ N ₃ O ₃ 2 PS/313.31	4017-47-8	35	6	3.55	358	1.33	2.38	4.90x10 ⁻³	na	300

Supplementary Material

Tabl	e S4.1 The prop	<i>perties of the pestic</i>	ides	(continued)											
Grou	 Pesticides and degradation products/type 	Mode of action	Use status (EC/NE)	Major pests controlled	MF/MW(g mol ⁻¹)	^a CASRN	SW	^b DT50	'Log P	dKoc	VP	¢GUS index	,HLC	gKd	hBCF
8	o, p'-DDT!/ins	s Central nervous system (CNS) stimulant	NP/NP	Mosquitoes; Houseflies; Body lice; Beetles; Gypsy moths	C ₁₄ H ₉ Cl ₅ /35 4.48	789-02-6	na	na	na	151000	na	-3.89	na	na	na
	p, p'- DDD!/ins	NSY stomach and contact action	NP/NP	Mosquitoes; Lice	C ₁₄ H ₁₀ C ₁₄ /3 20.04	72-54-8	0.09	na	6.02	131000	0.18	-2.46	4.0x10 ⁻⁶	na	na
	p, p'-DDE!/ins	s CNS stimulant; A weak androgen receptor antagonist	NA/NP	_	C ₁₄ H ₈ Cl ₄ /31 8.02	72-55-9	0.12	5000	6.51	na	na	na	na	50000	1800
	p, p'-DDT!/ins	s CNS stimulant; GABA-gated chloride channel antaoonist	NP/NP	Mosquitoes; Houseflies; Body lice; Beetles; Gypsy moths	C ₁₄ H ₉ Cl ₅ /35 4.49	50-29-3	0.025	na	6.91	151000	na	-3.89	na	na	na
	α - Endosulfan!/i ns	NSY with contact and stomach action	NP/NP	Mites; Ticks; Tstetse fly; Beetle; Aphids; Leaf hoppers	C ₉ H ₆ Cl ₆ O ₃ S /406.93	959-98-8	0.32	86	4.74	11500	8.30	-0.1	1.48	na	2755
	α-HCH!/ins	NA	NP/NP	Soil-dwelling and plant eating insects	C ₆ H ₆ Cl ₆ /290 .83	319-84-6	5	na	3.82	1888	5.99	1.62	na	na	na
	β- Endosulfan!/i ns	NSY with contact and stomach action	dN/dN	Mites: Ticks; Tstetse fly; Beetle: Aphids; Leaf homers	C ₉ H ₆ Cl ₆ O ₃ S /406.93	33213-65-9	0.45	na	3.83	na	na	na	na	na	na
	γ-HCH!/ins	CNS stimulant with contact and stomach action	dN/dN	Soil-dwelling and plant eating insects	C ₆ H ₆ Cl ₆ /290 .82	58-89-9	8.52	148	3.50	1270	4.40	3.95	1.48x10 ⁻⁶	na	1300
AD	Chlorantranili prole/ins	Ryanodine receptor	AP/AP	Cabbage loopers; Corn borers; Armyworms; Cutworms	C ₁₈ H ₁₄ BrCl ₂ N ₅ O ₂ /483.15	500008-45-7	0.88	204	2.86	362	6.3 X 10 ⁻⁹	4.22	3.2x10 ⁻⁹	3.18	15
DNN	Imidacloprid/i ns	SYS with contact and stomach action	AP/AP	Plant hopper; Aphids; Termites; Beetle; Fleas; White grups, Craneflies; Ants; Crickets	C9H10CIN5O 2/255.66	138261-41-3	610	174	0.57	na	4.0 X 10 ⁻⁷	3.74	1.7x10 ⁻¹⁰	na	0.61
BD	Carbendazim/ fun	SYS with curative and protectant activity	NP/AP	Husk spot; Grey mould; Crown rot	C9H5N3O2/1 91.21	10605-21-7	×	22	1.48	na	0.09	2.53	3.6x10 ⁻³	na	25

Tanne	THE FIRE PLACE WAY	manned and for a		(man)												ŝ
Group F	esticides and	Mode of action	Use status	Major pests controlled	MF/MW(g	^a CASRN	SW	^b DT50	cLog	dKoc	VP ^e	GUS f	LC g	Kd h	BCF	
q	legradation		(EC/NE)		mol ⁻¹)							ndex				
d	products/type															
PA N	Metalaxyl/fun	SYS with curative and protective	AP/AP	Downy mildew; Foliar and tuber blight;	C _{1s} H ₂₁ NO ₄ : /279.33	58737-19-1	8400	39	1.75	162	0.75 2	.79 1.	6x10 ⁻⁵ n	a 7		
	701	action	NIA	Damping-orr		6 26 87220		14	5	00	, ,	60	21	5		
	۲-(۲/۱۰)- himethvlnhenvl)-N-			Па	/265.3	7-10-40/10	па	10	па	00	c c	6	+	=	-	
, ∵.=	methoxyacetyl)alan ne ^m /fun	_														
MOD E	Emamectin/ins	NSY, acts by	AP/AP	Lepidoptera species;	C ₅₆ H ₈₁ NO ₁	155569-91-8	24	1	5	377000	0.004 n	a 1.	7x10 ⁻⁴ n	a 8	0	
		causing insect		Mites; Thrips;	₅ /1008.3											
		paralysis; Chloride		Leafminers; Emerald ash												
		channel activator		borer; Caterpillars												
Notation.																
aCASRN	', chemical abstracts	s service registry numl	ber; WS (mg	t L ⁻¹), water solubility [L–L	ر0S <50) مەر), M-Moderate	: (WS: 5() - 500), H	High (WS >500	[
bDT50 (f	ield days), half-life	time [NP-Non-Persis	tent (DT50 -	<30), MP-Moderately Pers	istent (DT50:	30-100), P-Pe	ersistent (DT50: 1	00-365)	, VP-Vei	v Persist	tent (DT	50 > 365)	_		
cLog P, C	Octanol-water Partit	tion Coefficient [LB-L	ow bioaccur	mulation (Log P<2.7), MB-	-Moderate bio	accumulation (Log P: 2	2.7-3), HI	3-High b	ioaccum	ilation (I	.og P: >	3)]			
dKoc (ml	L/g), soil organic ca	arbon-water partitionin	ng coefficien	t [VM-Very mobile (Koc <	<15), M-Mobil	le (Koc: 15-75), MM-N	Ioderatel	v mobile	(Koc: 75	-500), SI	M-Sligh	tly mobil	e (Koc		
500-4000), NM-Non mobile	(Koc: >4000)]	6	-	2					,		0		_		
VP (mPa)), vapour pressure [1	L-Low volatility (VP	<5.0), M–M	Ioderately volatile (VP: 5.0	−10.0), H−Hig	ghly volatile (V	/P >10.0	[(
eGUS inc	lex, groundwater ub	biquity score [L-Low .	leachability	(GUS <1.8), T-Transition	state (GUS: 2.	.8-1.8), H-Hig	th leacha	bility (G	US >2.8)							
fHLC (Pa	a m ³ mol ⁻¹), Henry's	s Law Constant [HV-F	Highly volati	ile (HLC >100), MV-Mode	stately volatile	s (HLC: 0.1-10	0), NV-1	Von volat	ile (HLŐ	[(10)]						
gKd, soil	sorption coefficient	t														
hBCF, bi	o-concentration fact	tor [LW-Lower potent	tial (BCF<1(00), THC-Threshold for co.	ncern (BCF: 5	5000-100), HP-	Higher p	otential (BCF>5([(00						
Status is l	based on EC regulat	tion 1107/2009 and Pt	esticide Regi	istration and Management]	Division, Gove	ernment of Nej	pal									
$^{\wedge} = Pestic$	side properties were	c retrieved from the PF	⁹ DB and the	BPDB databases which we	ere maintained	1 by the Univer	sity of H	lertfordsh	iire, UK	(Retrieve	d on: 23	Noveml	er, 2019)	_		
$AP = Ap_1$	proved; NP = Not al	pproved; "" = still in	n use but in ti	he process to ban; MF = M	lolecular form	ula; MW = Mc	olecular v	veight; !	= Banne	d pesticid	es; na =	Not ava	lable; m	Ш		
Degradati	ion products; NA =	Not applicable														
Group: U	Inclassified, UNC; (Organophosphates, OF	P; organochle	orines, OC; Anthranilic dia	unide, AD; Ne	conicotinoid, N	ND; Ber	nzimidazo	ole, BD;	Phenylan	nide, PA	; Micro-	organism	derive	d,	
MOD. Tł	ne group is based on	1 PPDB (2019)														
NSY/SY:	S= Non-systemic/sy	vstemic; AChE/ChE =	Acetylcholi	nesterase/Cholinesterase												

Table S4.1 The properties of the pesticides......(continued)

Table S4.2 Detected resid	dues in so	ils with their corres	sponding hui	nan health cor	sequences#.						
Pesticides and	Cancer	Genotoxicity	Endocrine	Reproductive	Cholinesterase	Nervous	Respiratory	Skin	Skin	Eye	Phototoxicity
degradation			disruption	disorders	inhibition	system	problems	irritation	sensitization	problems	
products						disorders					
3,5,6-trichloro-2-pyridinol	DU	A3; B3; C3; D0; E0	DU	PSNI	DU	DU	DU	DU	DU	DU	DU
Carbendazim	PSNI	A2; B3; C3; D0; E1	PSNI	U	NC	NC	NC	NC	DU	NC	DU
Chlorantraniliprole	NC	A3; B0; C3; D0; E3	NC	PSNI	NC	NC	DU	NC	DU	PSNI	DU
Chlorpyrifos	NC	A3; B3; C3; D0; E3	PSNI	U	U	U	NC	NC	PSNI	NC	NC
Emamectin	NC	A3; B0; C0; D0; E3	DU	PSNI	NC	PSNI	DU	NC	PSNI	U	DU
Imidacloprid	NC	A3; B3; C3; D3; E2	DU	C	NC	PSNI	NC	PSNI	DU	PSNI	DU
Metalaxyl	NC	A3; B0; C0; D0; E3	NC	NC	NC	NC	NC	NC	PSNI	NC	DU
N-(2,6-dimethylphenyl)-N-	DU	A0; B0; C0; D0; E0	DU	DU	DU	DU	DU	DU	DU	DU	DU
(methoxyacetyl)alanine											
o,p'-DDT	U	A0; B0; C0; D0; E0	U	C	NC	U	DU	J	DU	DU	DU
p,p'-DDD	U	A0; B0; C0; D0; E1	U	U	DU	DU	U	C	DU	NC	DU
p,p'-DDE	U	A0; B0; C0; D0; E2	U	C	DU	U	DU	J	DU	NC	DU
p,p'-DDT	U	A0; B0; C0; D0; E1	PSNI	U	DU	U	DU	C	DU	PSNI	DU
Profenofos	NC	A3; B0; C0; D0; E0	DU	NC	U	U	DU	J	DU	PSNI	DU
Quinalphos	NC	A0; B0; C0; D0; E0	DU	PSNI	U	C	C	C	DU	C	DU
Triazophos	NC	A3; B0; C0; D0; E0	DU	DU	C	J	c	C	DU	C	DU
Notation.		:	-		-	-		-			
C = Known to cause health c	Onseguence	ces: NC = Known not :	to cause healt	h consequences:	PSNI = Possibly.	status has n	ot been identif	ied: DU = C	ata unavailabl	٩	

Human health consequences were retrieved from the PPDB and the BPDB databases which were maintained by the University of Hertfordshire, UK (Retrieved on: 23 November, 2019) # Human health consequences were retrieved from the PPDB and the BPDB databases which were maintained by the University of Hertfordshire, UK (Retrieved on: 23 November, 2019) Genotoxicity: A, Chromosome aberration; B, DNA damage/repair; C, Gene mutation; D, Genome mutation; E, Unspecified genotoxicity type (Data sources: EFSA database and others)

0: Data not available; 1: Positive; 2: Mixed/ambiguous results; 3: Negative

Exposure factors	Unit	Adolescent (ado)	Adult (adu)
Average life span (AT)	days	LTx365	LTx365
Body weight (BW)	kg	32	62
Dermal absorption factor (ABS)	unit less	0.13	0.13
Dermal adherence factor (SAF)	mg cm ⁻²	0.2	0.07
Exposure duration (ED)	years	14	30
Exposure frequency (EF)	days yr ⁻¹	350	350
Ingestion rate (IR _{ing})	mg day⁻¹	100	100
Inhalation rate (IR _{inh})	m³ day⁻¹	17.7	17.5
Lifetime (LT)	years	70	70
Particle emission factor (PEF)	m³ kg⁻¹	1.36x10 ⁹	1.36x10 ⁹
Surface area (SA)	cm ² day ⁻¹	2800	5700
Conversion factor (CF)	kg mg ⁻¹	10 ⁻⁶	10 ⁻⁶

Table S4.3 Constant parameters and their values for the cancer and non-cancer risk estimation (USEPA, 2002; Wang, 2007).

Notation.

BW is based on WHO (2011); for non-cancer risk estimation, AT = ED x 365

Table S4.4 The carcinogenicity slope factor (SF) $[(mg kg \cdot 1 day \cdot 1)^{-1}]$ of pesticides through different exposure routes (Qu et al., 2015).

Pesticides	CSFingestion	CSF _{dermal}	CSFinhalation
p,p'-DDD	2.40E-01	3.43E-01	2.40E-01
p,p'-DDE	3.40E-01	4.86E-01	3.40E-01
p,p'-DDT	3.40E-01	4.86E-01	3.40E-01

Table S4.5 Pesticides with their corresponding reference dose (RfD) and gastrointestinal absorption factor (ABS_{GI}) values for cancer and non-cancer risk estimation (USEPA, 2018).

Pesticides	RfD (mg kg ⁻¹ dav ⁻¹)	ABS _{GI} (unit
3.5.6-trichloro-2-pyridinol ^m	na	na
Carbendazim	0.02	1 ^{dv}
Chlorantraniliprole	1.56	1 ^{dv}
Chlorpyrifos	0.001	1
Emamectin	0.0005	1 ^{dv}
Imidacloprid	0.06	1 ^{dv}
Metalaxyl	0.08	1
N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine ^m	na	na
Profenofos	na	na
Quinalphos	0.0005	1
Triazophos	0.001	1 ^{dv}

RfC values (mg/m³) of the pesticides were not available (na) in Integrated Risk Information System (IRIS) developed by US EPA, hence assessment of non-cancer risk via inhalation is excluded from this study. Furthermore, ABS_{GI} values of some pesticides did not appear in the system therefore, a default value (dv) of 1 was used for the pesticides (except profenofos and other degradation products, m) in the calculations. Profenofos and other degradation products are excluded in the calculations as their RfD values did not appear.

Pesticides	CDI-ing(ado)	CDI-der(ado)	CDI-inh(ado)	CDI-ing(adu)	CDI-der(adu)	CDI-inh(adu)
p,p'-DDD	4.26E-09	3.10E-09	5.55E-13	4.71E-09	2.44E-09	6.06E-13
p,p'-DDE	1.99E-09	1.45E-09	2.59E-13	2.20E-09	1.14E-09	2.83E-13
p,p'-DDT	7.25E-09	5.28E-09	9.43E-13	8.02E-09	4.16E-09	1.03E-12

Table S4.6 The average CDI valu	es (ma ka-1 dav 1) of pesticides	for adolescents (ado) and adults (adu)

Table S	4.7 Predicted e	nvironment cor	ncentrations	[(PECs) µg/kg	l of pe	sticide	s (in so	oil) tha	t farm	ers apt	olied ir	diffe	rentj	Fields.
		arApplication	^{np} No. of	ai Application	PECs, a	act (day	s after _I	pesticid	e applic	cations)				
Fields	Pesticides	rate (g/ha)	applications	interval (days)										Reference
					0	1	2	4	7	14	28	50	100	
	Carbendazim	1770	17	15	3130	3030	2940	2760	2510	2010	1300	650	0	
	Chlorpyrifos	11590	19	15	24900	24300	23700	22500	20900	17600	12400	7200	0	
	Dichlorvos	23120	19	15	15400	4900	1500	200	0	0	0	0	0	
Tomato	Dimethoate	3380	25	10	1430	1300	1180	970	720	360	60	10	0	
	Emamectin	770	21	15	510	260	130	30	0	0	0	0	0	
	Imidacloprid	1080	15	15	7350	7320	7290	7230	7150	6950	6570	6020	0	
	Metalaxyl	1730	20	15	4900	4820	4730	4570	4330	3820	2980	2020	0	
	Profenofos	6830	15	15	5890	5330	4830	3960	2940	1470	370	40	0	This study
	Triazophos	2330	11	30	1720	1600	1480	1270	1010	590	200	40	0	
	Chlorantranilipr	ole 2100	17	15	6530	6510	6490	6440	6380	6230	5940	5510	0	
Eggplant	: Dimethoate	8720	24	15	7510	6810	6160	5060	3760	1880	470	50	0	
	Imidacloprid	440	14	30	2120	2110	2100	2080	2060	2000	1890	1730	0	
	Profenofos	13600	10	30	9560	8660	7840	6430	4780	2390	600	70	0	
Chilli	Dimethoate	1790	8	30	1260	1140	1030	850	630	310	80	10	0	
	Profenofos	4140	13	15	3570	3230	2930	2400	1780	890	220	30	0	
Notation														
Except fc	or dimethoate (to	mato) and chlora	ntraniliprole (chilli), 80%, the	fractio	interce	epted b	y the ve	getable	cover 1	or othe	r pesti	cides	
is 50%; E	hepth of soil = 5 ci	m and bulk densi	ty = 1.5 g cm ⁻²	³ (EFSA, 2002); a	ar, np ai	nd ai we	ere base	io uo pa	ur previ	ous stu	dy (Bha	ndari e	et al.,	
2018). In	the respective fie	elds, few pesticide	es in bold shov	wed higher PECs	s (days a	after pe	sticides	applica	tion, 0)	than th	eir glok	al pest	ticide	
soil regu	latory guidance n	naximum values,	indicating gre	ater risks for fa	rm wor	kers.								

st study	
om a pa	
domly fr	
sted ran	arison.
r ⁻¹)] seled	le comp
s) (µg kg	or possib
[(PSRGV	idered fo
e values	ere cons
guidanc	values w
gulatory	aximum
le soil rei	s, the m
l pesticia	PSRGV
eir globa	multiple
s and the	des with
esticides	or pestici
studied p	2017). Fc
3 List of s	nnings, 2
able S4.8	'Li and Je

	Pasticidas		PSRG//c				cupdup.				
-	Carbendazim	Belarus, Georgia, Ukraine	100								
2	Chlorpyrifos	Australia Florida (US), Bahamas Alberta BC (Canada)	160000-	250000	49000-	180	100000	180000	61000	200000	200
		Lithuania Malaysia New Zealand Spain Ukraine	340000		95000						
ŝ	Dichlorvos	Belarus, Armenia, Georgia Florida (US), Bahamas Malaysia,	30	300	1700	17	1000	100			
		New Zealand BC (Canada) China Moldova, Tatarstan, Russia									
4	Dimethoate	China Armenia, Belarus Florida (US),	2000	100	13000	5.5-5.8	12000	50			
		Bahamas Alberta Malaysia, New Zealand Vietnam									
ъ	Imidacloprid	Georgia	40								
9	Metalaxyl	Armenia, Georgia, Ukraine BC (Canada) Malaysia, New Zealand	50	3700	3700000						
7	o,p'-DDT	Bulgaria BC (Canada) China, Moldova, Tatarstan, Russia	100	15	100						
∞	p,p'-DDD	Andorra Bulgaria Georgia (US) BC	700	100	660	15	500	100	10	700	100
		(Canada) China Ecuador Montenegro New Zealand Russia									
6	p,p'-DDE	Andorra Bulgaria Georgia (US) BC	600	100	660	15	500	100	10	700	100
		(Canada) China Ecuador Montenegro New Zealand Russia									
10	p,p'-DDT	Andorra Bulgaria Georgia (US) BC	200	100	660	15	500	100	10	700	100
		(Canada) China Ecuador Montenegro New Zealand Russia									
11	Phorate	New Zealand Florida (US), Bahamas Alberta Malaysia BC	12210	16000	75-140	12000	12				
		(Canada)									
12	Profenofos	Georgia Ukraine	100	1							
13	Quinalphos	BC (Canada) Arizona, New Zealand, Malaysia	31	31000							
14	Triazophos	Georgia (US)	25000								
15	α-Endosulfan	Bahamas, Florida (US) Ecuador Panama Victoria	450000	100	500	1000					
16	α-HCH	Serbia, Italy, Hungary, Ecuador, Czech Republic,	10	1	0.9	100	200	10000	2000	400	50000
		Andorra Bulgaria BC (Canada) China, Croatia Estonia,									
		Singapore GermanyLatvia Lithuania EU									
17	β-Endosulfan	Bahamas, Florida (US) Ecuador Panama Victoria	450000	100	500	1000					
18	ү-нсн	Serbia, Italy, Hungary, Ecuador, Czech Republic,	10	1	0.01	100	200	10000	2000	400	50000
		Andorra Bulgaria BC (Canada) China,									
		Croatia Estonia Germany Latvia Lithuania EU									
Notati	on.										
PSRGV	's of omethoate	, 3,5,6-trichloro-2-pyridinol, chlorantraniliprole, N-(2,6-dimethylph	henyl)-N-(r	nethoxya	cetyl)alanir	e and em	amectin w	rere not n	oticed an	d did not	apear in
the list											

Table S4.9 The non-dietary chronic daily intake for adolescents (ado) and adults (adu) derived from exposure to the pesticides in soils. Only the positive samples were included in the calculation. CDI values in ma ka⁻¹ day 1.

Pesticides	CDI-	CDI-	CDI-	CDI-	CDI-	CDI-	CR-	CR-	CR-	CR-	CR-	CR-
	(ado-ing)	(ado-der)	(ado-inh)	(adu-ing)	(adu-der)	(adu-inh)	(ado-ing)	(ado-der)	(ado-inh)	(adu-ing)	(adu-der)	(adu-inh)
p,p'-DDE	5.99E-10	4.36E-10	7.80E-14	6.63E-10	3.44E-10	8.53E-14	2.04E-10	2.12E-10	2.65E-14	2.25E-10	1.67E-10	2.90E-14
	6.83E-10	4.97E-10	8.89E-14	7.56E-10	3.92E-10	9.72E-14	2.32E-10	2.42E-10	3.02E-14	2.57E-10	1.90E-10	3.31E-14
	3.85E-09	2.80E-09	5.01E-13	4.26E-09	2.21E-09	5.48E-13	1.31E-09	1.36E-09	1.70E-13	1.45E-09	1.07E-09	1.86E-13
	8.33E-09	6.06E-09	1.08E-12	9.21E-09	4.78E-09	1.19E-12	2.83E-09	2.95E-09	3.69E-13	3.13E-09	2.32E-09	4.03E-13
	3.33E-09	2.42E-09	4.33E-13	3.68E-09	1.91E-09	4.73E-13	1.13E-09	1.18E-09	1.47E-13	1.25E-09	9.27E-10	1.61E-13
	6.59E-10	4.80E-10	8.58E-14	7.29E-10	3.78E-10	9.38E-14	2.24E-10	2.33E-10	2.92E-14	2.48E-10	1.84E-10	3.19E-14
	8.87E-10	6.46E-10	1.15E-13	9.81E-10	5.09E-10	1.26E-13	3.02E-10	3.14E-10	3.92E-14	3.34E-10	2.47E-10	4.29E-14
	6.53E-10	4.76E-10	8.50E-14	7.22E-10	3.75E-10	9.30E-14	2.22E-10	2.31E-10	2.89E-14	2.46E-10	1.82E-10	3.16E-14
	6.29E-10	4.58E-10	8.19E-14	6.96E-10	3.61E-10	8.96E-14	2.14E-10	2.23E-10	2.78E-14	2.37E-10	1.75E-10	3.04E-14
	1.40E-09	1.02E-09	1.82E-13	1.54E-09	8.01E-10	1.99E-13	4.75E-10	4.94E-10	6.18E-14	5.25E-10	3.89E-10	6.76E-14
	1.32E-09	9.60E-10	1.72E-13	1.46E-09	7.56E-10	1.88E-13	4.48E-10	4.66E-10	5.83E-14	4.96E-10	3.68E-10	6.38E-14
	2.76E-09	2.01E-09	3.60E-13	3.06E-09	1.58E-09	3.93E-13	9.39E-10	9.78E-10	1.22E-13	1.04E-09	7.70E-10	1.34E-13
	2.85E-09	2.07E-09	3.70E-13	3.15E-09	1.63E-09	4.05E-13	9.68E-10	1.01E-09	1.26E-13	1.07E-09	7.94E-10	1.38E-13
	2.57E-09	1.87E-09	3.34E-13	2.84E-09	1.47E-09	3.65E-13	8.72E-10	9.08E-10	1.14E-13	9.65E-10	7.15E-10	1.24E-13
	1.01E-09	7.33E-10	1.31E-13	1.11E-09	5.78E-10	1.43E-13	3.42E-10	3.56E-10	4.46E-14	3.79E-10	2.81E-10	4.87E-14
	1.82E-09	1.33E-09	2.37E-13	2.02E-09	1.05E-09	2.59E-13	6.19E-10	6.45E-10	8.06E-14	6.85E-10	5.08E-10	8.82E-14
	1.82E-09	1.33E-09	2.37E-13	2.02E-09	1.05E-09	2.59E-13	6.19E-10	6.45E-10	8.06E-14	6.85E-10	5.08E-10	8.82E-14
	5.99E-10	4.36E-10	7.80E-14	6.63E-10	3.44E-10	8.53E-14	2.04E-10	2.12E-10	2.65E-14	2.25E-10	1.67E-10	2.90E-14
p,p'-DDT	2.28E-09	1.66E-09	2.97E-13	2.53E-09	1.31E-09	3.25E-13	7.76E-10	8.08E-10	1.01E-13	8.59E-10	6.37E-10	1.10E-13
	1.53E-08	1.11E-08	1.99E-12	1.69E-08	8.77E-09	2.17E-12	5.20E-09	5.41E-09	6.76E-13	5.75E-09	4.26E-09	7.39E-13
	4.70E-08	3.42E-08	6.12E-12	5.20E-08	2.70E-08	6.69E-12	1.60E-08	1.66E-08	2.08E-12	1.77E-08	1.31E-08	2.27E-12
	6.29E-10	4.58E-10	8.19E-14	6.96E-10	3.61E-10	8.96E-14	2.14E-10	2.23E-10	2.78E-14	2.37E-10	1.75E-10	3.04E-14
	1.89E-09	1.37E-09	2.46E-13	2.09E-09	1.08E-09	2.69E-13	6.42E-10	6.68E-10	8.35E-14	7.10E-10	5.26E-10	9.13E-14
	2.02E-09	1.47E-09	2.63E-13	2.23E-09	1.16E-09	2.87E-13	6.87E-10	7.15E-10	8.94E-14	7.59E-10	5.63E-10	9.77E-14
	8.99E-10	6.54E-10	1.17E-13	9.94E-10	5.16E-10	1.28E-13	3.06E-10	3.18E-10	3.98E-14	3.38E-10	2.51E-10	4.35E-14
	1.16E-09	8.42E-10	1.51E-13	1.28E-09	6.64E-10	1.65E-13	3.93E-10	4.09E-10	5.12E-14	4.35E-10	3.22E-10	5.60E-14
	6.59E-10	4.80E-10	8.58E-14	7.29E-10	3.78E-10	9.38E-14	2.24E-10	2.33E-10	2.92E-14	2.48E-10	1.84E-10	3.19E-14
	6.77E-10	4.93E-10	8.81E-14	7.49E-10	3.89E-10	9.64E-14	2.30E-10	2.40E-10	3.00E-14	2.55E-10	1.89E-10	3.28E-14
p,p'-DDD	1.17E-09	8.51E-10	1.52E-13	1.29E-09	6.70E-10	1.66E-13	2.80E-10	2.92E-10	3.65E-14	3.10E-10	2.30E-10	3.99E-14
	4.96E-09	3.61E-09	6.46E-13	5.49E-09	2.85E-09	7.06E-13	1.19E-09	1.24E-09	1.55E-13	1.32E-09	9.76E-10	1.69E-13
	6.65E-09	4.84E-09	8.66E-13	7.36E-09	3.82E-09	9.47E-13	1.60E-09	1.66E-09	2.08E-13	1.77E-09	1.31E-09	2.27E-13


Figure S4.1 Box plots showing non-cancer risks of pesticides for adolescents (a) and comparison of exposure pathways (b). Only the positive samples were included in the calculation of HQ and HI.



Figure S4.2 Box plots showing non-cancer risks of pesticides for adults (a) and comparison of exposure pathways (b). Only the positive samples were included in the calculation of HQ and HI.

5. Ecological risk assessment of pesticide residues in soils from vegetable production areas: A case study in S-Nepal

Pesticides pose a serious risk to ecosystems. In this study, we used European Food Safety Authority methods, such as risk quotient (RQ) and toxicity exposure ratios (TER), to assess the potential ecological risks of 15 pesticide residues detected in agricultural soils in the Gaidahawa Rural Municipality of Nepal. The mean and maximum concentrations of the detected pesticide residues in the soil were used for risk characterization related to soil organisms. RQmean, TERmean and RQmaximum, TERmaximum were used to determine general and the worst-case scenarios, respectively. Of all the detected pesticides in soils, the no observed effect concentration (NOEC) for 27% of the pesticides was not available in literature for the tested soil organisms and their TER and RQ could not be calculated. RQ threshold value of ≥ 1 indicates high risk for organisms. Similarly, TER threshold value of \geq 5, which is acceptable trigger point value for chronic exposure, indicates an acceptable risk. The results showed that the worst-case scenario (RQ_{maximum}) indicated a high risk for soil organisms from chlorpyrifos $[RQ_{maximum} > 9 \text{ at depths (cm) of } 0-5, 15-20 \text{ and } 35-40 \text{ soil}$ layer]; imidacloprid (1.78 in the 35-40 cm soil layer) and profenofos (3.37 in the 0-5 cm and 1.09 in the 35-40 cm soil layer). Likewise, for all the soil depths, the calculated TER for both the general and worst-case scenarios for chlorpyrifos ranged from 0.37 to 3.22, indicating chronic toxicity to F. candida. Furthermore, the risk of organophosphate pesticides for soil organisms in the sampling sites was mainly due to chlorpyrifos, except for two study sites where the risk was from profenofos. Ecological risk assessment (EcoRA) of the pesticide use in the study area indicated that the EFSA soil organisms were at risk at some of the localities where farmers practiced conventional farming. Farmer awareness and training related to pesticide effects on ecological entities and the safe disposal behaviour for pesticide containers including leftover pesticides are urgently needed.

Based on:

Bhandari, G., Atreya, K., Vašíčková, J., Yang, X., Geissen, V., 2020. Ecological risk assessment of pesticide residues in soils from vegetable production areas: A case study in S-Nepal. Submitted to the Environmental Pollution.

5.1 Introduction

Over 4 million tons of pesticides are used annually worldwide (FAO, 2017a). Unfortunately, this number is only expected to increase due to the burgeoning world population demanding more food from shrinking agricultural lands that suffer from declining soil quality. As if that wasn't enough, climate change and the emergence of new pests and diseases are throwing all kinds of new challenges into the mix (Brain and Anderson, 2019; Delcour et al., 2015; Xu et al., 2008). Modern farming methods rely on chemical pesticides to control insects and diseases, thereby improving food quantity. However, both the abundant use and in some cases, misuse of pesticides have contributed to soil pollution (Tsaboula et al., 2016). Research has discovered that pesticides can bioaccumulate and become biomagnified in soil, leading to even greater possible risks for the environment (Haj-Younes et al., 2015; Yuantari et al., 2015). There should be systematic monitoring of pesticide levels in soil that should include an evaluation of pesticide toxicity as well as an ecological risk assessment (EcoRA).

Many ecotoxicological studies have stated that pesticides can induce DNA injury, disturb hormone activity, decrease growth and survival rates, affect reproduction, alter individual food consumption, and diminish the density of earthworm communities (Jager et al., 2007; Uwizeyimana et al., 2017; Wang et al., 2019). Pesticides in soils induced behavioural changes in some organisms thus affecting the environmental system and impairing predator-prey interactions (Dinh Van et al., 2014). A number of toxicological studies (Table S5.1) have confirmed that pesticides are harmful to soil fauna. According to the Regulation SANCO/10329/2002, the European Food Safety Authority (EFSA) recommended risk assessment methods such as risk quotient (RQ) and toxicity exposure ratios (TER) for soil organisms (EFSA et al., 2017). With these laws in place, ecotoxicological testing and systematic monitoring of pesticides in soil are being carried out in Europe and are slowly emerging from other parts of the world. Unfortunately, this testing is not yet a common occurrence in Nepal.

Pesticide application in Nepal has been increasing with the annual import going from 404 tons in 2012 to 635 tons in 2018 (CBS, 2019). The government of Nepal estimated the average application of pesticides to be about 396 g of active ingredients per ha in 2014 (PPD, 2014). About 80% of the imported pesticides were applied to vegetable fields (Adhikari, 2017). There were 169 types of active ingredients in pesticides registered and approved for use in agriculture in 2019 (GC and Neupane, 2019). Earlier studies have shown misuse and overuse of pesticides in agriculture (Aryal et al., 2014; Sharma et al., 2012), mainly in vegetable farming (Atreya et al., 2011; Chhetri et al., 2014). Unsustainable agricultural practices in Nepal expose soils to a mixture of pesticides that could decrease the country's

rich biodiversity, which currently includes 17,097 known fauna species (MoFE, 2018). Soil biota such as bacteria, fungi, nematodes, earthworms, enchytraeids, microarthropods (springtails and mites), and insect larvae along with several other organisms help to maintain soil quality: structure and properties, pivotal functions and major ecosystem services. Soil health is of crucial importance as it determines the quality and quantity of food production, biodiversity, and resilience to climate change. Increased use of pesticides carries a greater risk to soil health that may destroy ecological cycles, including the breakdown of organic material, sequestration of carbon, cycling of nutrients, pest suppressiveness of soil and soil fertility (Keesstra et al., 2016; Lavelle et al., 2006).

For the sustainable management and responsible application of pesticides, ecological risk assessment (EcoRA) is necessary. The EFSA's risk assessment procedure mainly involves an assessment of exposure and a characterization of risk (Figure S5.1). The risks posed by pesticides depend on the exposure concentrations and intrinsic (eco) toxicity, expressed as Toxicity-Exposure- Ratio (TER). The TER is interpreted using trigger standards as defined in the EU number 546/2011 (EC, 2002). TERs are identified for single assessments and single organisms. The TER explains the toxicity of a pesticide and provides an impression of the exposure estimates for each species separately. A trigger value equal to 5 represents "safety factors" for earthworms and other soil organisms. TER values <5 and \geq 5 indicate high risk (unacceptable) and low risk (acceptable), respectively. The ratio of a measured soil concentration (MSC) or a predicted soil concentration (PSC) to a predicted no-effect concentration (PNEC) is used to calculate RQ (EC, 2002; Palma et al., 2014; Vasickova et al., 2019) which represents no risk (RQ<0.01), lower risk ($0.01 \le RQ \le 0.1$), moderate risk $(0.1 \le RQ \le 1)$ and higher risk (RQ \ge 1). Earlier studies conducted elsewhere used both the TER and the RQ to define an EcoRA for pesticides (Thomatou et al., 2013; Vasickova et al., 2019; Wee and Aris, 2017). However, there are scant scientific studies examining the ecological risks of pesticides on soil organisms in Nepal.

This study aims 1) to investigate the potential risk posed by pesticide residues following the EFSA's guidelines for soil organisms [including earthworms (*Eisenia fetida*), enchytraeids (*Enchytraeus crypticus*), springtails (*Folsomia candida*), and mites (*Hypoaspis aculifer*) as well as nitrogen and carbon mineralization microorganisms, which are the recommended invertebrate subjects for ecotoxicological studies (EFSA et al., 2017; Jänsch et al., 2006)]; and 2) to compute the correlation between pesticide risk and a farmer's knowledge about pesticide use and behavior in the environment. This study provides the first evidence of the ecological risk of exposure to the most commonly applied pesticides in Nepal. The findings can be useful in developing effective pesticide risk mitigation strategies and national pesticide policy.

5.2 Materials and Methods

5.2.1 Pesticide residues in the study area

In a previous study (Bhandari et al., 2020) we studied pesticide residues in soils (3 depths) from 11 integrated and 38 conventional vegetable farms of the Gaidahawa Rural Municipality in the Rupandehi district, Nepal. Of the 23 pesticides analysed in our previous study, residues of 15 different pesticides were detected frequently and heavily in soils from conventional farms (Bhandari et al., 2020). Details of these pesticide residues are included here in Table S5.2. In this study we focus on the assessment of ecological risk of these residues. Detailed descriptions of the study area including the soil sampling points and the residues detected are stated in our previous paper (Bhandari et al., 2020). To the best of our knowledge, a residual limit for pesticides in soil has not yet been developed in Nepal. Therefore, we compared the measured concentration of pesticide residues with the guidance values established for different countries (Li and Jennings, 2017). The guidance values are the maximum concentration of individual pesticide residues present in soils posing no ecological risk.

5.2.2 Risk Assessment

Pesticide EcoRA included an assessment of exposure and ecotoxicity (effects) (Figure S5.1). We used two common methods: a) the TER for 4 selected species of the EFSA soil organisms (EC, 2002), and b) the risk quotient (RQ) for each pesticide residue (Renaud et al., 2018). When the no observed effect concentration (NOEC) value for a pesticide was available via systematic review, the RQ and TER were estimated to assess chronic EcoRA. When the value was not known, the risk could not be assessed. Additionally, we compared pesticide concentrations in soils with pesticide soil regulatory guidance values (PSRGVs) to see if findings using the TER and the RQ methods corresponded with the guidance values. The concentrations of banned pesticides such as DDT and its principal metabolites in soils were compared with the existing threshold values for soils. A farmer's field was denoted as a "site". The risk of organophosphate in pesticide mixtures was estimated by adding up all the individual pesticide risks with a common mode of action (Damodaran, 2019) based on the concentration addition (CA) technique (Bundschuh et al., 2014). The risk of pesticide mixtures for other chemical groups could not be computed because ecotoxicity data was not available.

5.2.2.1 Assessment of exposure

The concentration of pesticide residues detected in the 3 depths (0-5, 15-20 and 35-40 cm) (Bhandari et al., 2020) was used for the risk assessment. The geometric mean and maximum pesticide levels detected at the studied sites were used as the mean measured soil concentration (MSC_{mn}) and the maximum measured soil concentration (MSC_{mn}) and the maximum measured soil concentration (MSC_{mn}). In the first case, the value gives a general scenario (GS) (TER_{mn} or RQ_{mn}) and, in the second case, the value gives the worst-case scenario (WS) (TER_{mx} or RQ_{mx}) (Palma et al., 2014).

5.2.2.2 Assessment of toxicity

The toxicity assessment was based on the available ecotoxicological data from i) the pesticide properties database (PPDB) https://sitem.herts.ac.uk/aeru/ppdb/; ii) the draft assessment reports (DARs) from the EFSA; and iii) a basic literature search using the Web of Science Core Collection and Scopus databases. The search terms included:

- (i) Pesticide AND soil AND *toxic*
- (ii) Pesticide AND soil AND organism

A systematic review of literature was conducted based on the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) statement (Moher et al., 2009) for the assessment of toxicity including the effect of pesticides on soil organisms (Supplementary material, Figure S5.2 and Table S5.1).

In order to assess the ecological risks of pesticides, there are several ecotoxicological aspects that can be examined such as the no observed effect concentration (NOEC) and/or the lethal concentration at which 50% of the examined organisms exhibit mortality (LC_{50}), as well as a median effective concentration (EC_{50}) for organisms such as the earthworm (E. fetida), the enchytraeid (E. crypticus), the springtail (F. candida), the mite (H. aculifer) and the nitrogen and carbon mineralization microorganisms. These are organisms that need to be included in any study assessing pesticides for approval by the EFSA for the European Union. Data from past studies that were in agreement with the Organisation for Economic Co-operation and Development (OECD) standardized procedures for the organisms was considered for the current study. Since the toxicological dose descriptors listed above differed with their corresponding ecotoxicological output, they could not be compared. Although pesticides detected in soil have their LC₅₀ as well as EC₅₀, the present study was based on the available NOEC endpoints. If multiple NOEC values were available for a single organism, the geometric mean was considered and used. Of the 15 pesticides (Table S5.2), the NOEC for DDD, DDE, quinalphos and triazophos were not available for the tested species thus, the TER and RQ for these compounds could not be calculated.

Using the NOEC value, we derived the predicted no-effect concentration for the most sensitive species (PNEC_{mss}). To overcome issues such as insufficient toxicity data, errors and

inaccuracy related to the conservative approach, the PNEC_{mss} value was estimated as the lowest long-term NOEC divided by the assessment factor (AF). The most susceptible organism for each pesticide was selected to obtain the PNEC_{mss} with an AF to account for potential chronic risks. The selection of the AF was based on the guidance document of the EU (EC, 2002) and could range from 10 to 1000: (i) an AF of 1000 was used in a case where at least one LC₅₀ at one ecological level was available; (ii) an AF of 100 was used in a case where data from a long term assay was available; and (iii) an AF of 50 and 10 were used in the cases where two and three or more NOECs were available, respectively. In the present study, based on available long-term NOECs, we used an AF of 100, 50 and 10 (Table 5.1).

Table 5.1 Ecotoxicology (NOEC and NSDE in $\mu g/kg$) of pesticides for E. fetida (earthworm), E. crypticus (enchytraeid), F. candida (springtail), H. aculeifer (mite), and N and C mineralization organisms extracted from different sources (see details in footnote). Degradation products of chlorpyrifos and metalaxyl such as 3,5,6-TCP and N-alanine indicated 3,5,6-trichloro-2-pyridinol and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, respectively. The N-alanine was referenced as CGA 62826 in the EFSA document and hence its NOEC was used in the calculation. PNEC = Predicted no-effect concentration; NOEC = No observed effect concentration; AF = Assessment factor.

Group	Compound	Ε.	Ε.	F. candida,	H. aculeifer,	NSDE for N C	Critical	PNEC	PNECmss
		fetida,	crypticus,	NOEC	NOEC	mineralization	concentration	AF	
		NOEC	NOEC			microorganisms			
AD	Chlorantraniliprole	1000000d	na	390d	100000d	700ad 700ad	390	10	39
BD	Carbendazim	1000a,d	100b0	na	na	4800d^ 4800d^	100	10	10
MOD	Emamectin	2000d	na	na	na	400d^ na	2000	50	40
NND	Imidacloprid	178a	1000g	1250d	≥2670d	na	178	10	17.8
OC	DDT	280000ah	na	176000ah	na	na	176000	50	3520
OP	Chlorpyrifos	12700a	5000e	65c	na	na 4800ad	65	10	6.5
	Profenofos	na	na	<50f	na	250^ na	50	50	1
PA	Metalaxyl	40000ad	na	125000d#	na	1350d^ 1350d^	40000	10	4000
UNC	3,5,6-TCP	4600a,d	na	na	na	4150d^ 4150d^	4600	50	92
	N-alanine	500000d	na	na	na	na	500000	100	5000

Note.

Data stated as "≥"number or "<"number, the given number was used in a calculation.

a Lewis et al. (2006).

b Novais et al. (2010); \diamond = Data for *Enchytraeus crypticus* was not found in literatures, hence NOEC of carbendazim is used for *Enchytraeus albidus*.

c (Herbert et al., 2004); d Data collected from the EFSA documents that are available online; e (Carniel, 2019).

f (Liu et al., 2012); △= nitrification rate; g (de Lima et al., 2017).

h (RIVM, 2015).

PNECmss = the lowest long-term NOEC of the most susceptible species/AF.

na = Information on toxicity was not available in the refereed databases.

NSDE = No significant adverse effect. ^ = <25% effect considered as NSDE.

In the case of metalaxyl, NSDE data for ridomil gold was used. # = Toxicity data of metalaxyl to Folsomia candida was not available, hence the data of ridomil was used in the calculation.

Pesticides group: AD = Anthranilic diamide; UNC = Unclassified; OP = Organophosphate; NND = Neonicotinoid; OC = Organochlorine; BD = Benzimidazole; PA = Phenylamide; MOD = Micro-organism derived.

5.2.2.3 Risk characterization of a single pesticide

The most commonly used methods for assessing ecological risk are the Toxicity-Exposure-Ratio (TER) (EC, 2002) and the Risk Quotient (RQ) (Renaud et al., 2018). TERs based on NOECs for single test organisms, which included *E. fetida, E. crypticus, F. candida, H. aculifer* as well as nitrogen and carbon mineralization microorganisms were considered. Based on EC (2009), the TER approach relates toxicity and exposure. As mentioned earlier, in the cases denoted as " \geq value" or "< value", the given value was used. The TER for each pesticide was estimated by using the TER for the test organisms (TER_{species}) and the following Equation 5.1.

TER_{species}= $\frac{NOEC_{species}}{MSC_{maximum or mean}}$(5.1)

where, NOEC = No observed effect concentration and MSC = Measured pesticide concentration in soil.

The EC (2002) defined cut-off (trigger point) values of 5 and 10 for chronic and acute toxicity for soil organisms, respectively. Pesticide risk was considered negligible if the TER exceeded the cut-off values. TER values of \geq 10 or \geq 5, which are acceptable trigger point values for acute and chronic exposure, respectively, indicated an acceptable risk for the organisms (Jaabiri Kamoun et al., 2017).

The risk quotient of a pesticide i (RQ_i) provided an index for the risk of a single pesticide and was calculated as described in Equation 5.2.

 $RQ_i = \frac{MSC_{soil}}{PNEC_{mss}}....(5.2)$

where, MSC = Measured pesticide concentration in soil and $PNEC_{mss}$ = Predicted no-effect concentration for the most sensitive species.

The classification of the risk quotient was based on the previous existing studies (Sánchez-Bayo et al., 2002; Vryzas et al., 2011): no risk (RQ<0.01), lower risk ($0.01 \le RQ < 0.1$), moderate risk ($0.1 \le RQ < 1$) and higher risk (RQ ≥ 1).

5.2.2.4 Risk characterization of pesticide mixtures-concentration using addition model

The widely accepted concentration addition (CA) approach was used to calculate the toxicity of pesticide cocktails (Vasickova et al., 2019; Wee and Aris, 2017). Multi-pesticide exposures can lead to additive actions. The mixture RQ (RQ_{mix}) of organophosphates (OP) was estimated by adding up the individual RQ_i of each pesticide that belongs to the OP group. Furthermore, the total risk of multiple pesticide residues of a site (ΣRQ_{site}) was estimated using the concentration addition (CA) based on the mixture risk assessment method (Bundschuh et al., 2014). CA, the most suitable model to use in ecotoxicological studies (Chen et al., 2014), is based on the assumption that all pesticides in a cocktail have the same mode of action and can be stated as in Equation 5.3.

$$\sum (RQ_{site} \text{ or } RQ_{mix}) = \sum_{k=1}^{n} RQ_{i} = \sum_{k=1}^{n} (MSC_{i}/PNEC_{i})....(5.3)$$

where, RQ_{site} = Risk quotient of a site; RQ_{mix} = Risk quotient of pesticide mixtures; RQ_i = Risk quotient of a pesticide i; MSC_i = Measured soil concentration of a pesticide i; $PNEC_i$ = Predicted no-effect concentration of a pesticide i; n = number of pesticides.

The classification of RQ_{mix} was based on Sánchez-Bayo et al. (2002) and Vasickova et al. (2019) as mentioned for the RQ above.

To correlate the classified RQ at sites where the pesticide knowledge and behaviour of farmers was known, we developed a summative score based on farmer's replies to the survey questions (Table 5.2).

Table 5.2 Variables (n=6) for farmers pesticide use KNB score. Results of the questionnaire survey among the farmers was based on our previous study (Bhandari et al., 2018). Commercial vegetable farmers from different villages were selected for the survey conducted in 2017. Of the 183 farmers, this study included the scores of 49 farmers that were selected randomly.

,-		
Variables	Description	KNB values (1 or 0)
ANIMALS	Do pesticides negatively affect animals?	0=no, 1=yes
BIRDS	Do pesticides negatively affect birds around you?	0=no, 1=yes
FISHES	Do pesticides negatively affect fishes?	0=no, 1=yes
HONEYBEES	Do pesticides negatively affect honeybees?	0=no, 1=yes
PESCONT	Do you throw pesticides container at field after use?	0=yes, 1=no
UNUSEDPES	Do you throw unused/leftover pesticides at field?	0=yes, 1=no

The contribution of each RQ_i to RQ_{site} or RQ_{mix} was derived following Equation 5.4. % contribution = $\left(\frac{RQ_i}{\sum (RQ_{site} \text{ or } RQ_{mix})}\right)$(5.4)

where, RQ_i = Risk quotient of a pesticide i; RQ_{site} = Risk quotient of a site; RQ_{mix} = Risk quotient of pesticide mixtures.

5.2.3 Statistical analysis

Data analysis on the concentration of pesticides \geq LODs was performed. Data entries where pesticide concentrations were <LODs were left empty and excluded from the study (Sun et al., 2016). We used the Spearman's rho correlation coefficient to calculate the linear correlation of the risk quotient (RQ_{site}) calculated for the soil depths (0-5, 15-20 and 35-40 cm) and the knowledge and behaviour (KNB) of farmers. A p-value less than 0.05 was considered significant. The correlation was also used to demonstrate if there was a positive relationship between the risks of pesticides at different sites and the farmers' pesticide KNB scores. The score was based on the sum of values from each variable listed in Table 5.2.

5.3 Results

5.3.1 Risk assessment

5.3.1.1 Ecological risk based on the TER approach

Data from ecotoxicological tests conducted following the ISO/OECD procedures were considered for this research. Most ecotoxicological information about pesticides included data for only a single organism. Toxicological information for several detected pesticides was unavailable and thus assessment of their potential ecological risk was not possible. Data on the NOEC for *E. fetida* was available for 60% of the 15 detected pesticides (Table 5.1). In Nepal, ecotoxicological studies have not been carried for many of the pesticides that have been approved for agricultural use. However, ecotoxicological studies from other countries have been used (Table S5.1).

The toxicity exposure ratio (TER) for a general scenario (GS) and a worst-case scenario (WS) are presented in Table 5.3a-5.3c. For both scenarios, chronic risk for E. fetida, E. crypticus, and H. aculifer was negligible for all assessed pesticides at all the depths of soil, except chlorpyrifos. Both TER_{mx} and TER_{mn} calculated for different depths were above the cut-off value of 5 for chronic toxicity. Pesticides such as carbendazim, chlorantraniliprole, DDT, emamectin, metalaxyl and N-alanine notably showed high TERs at different depths. Of all pesticides, chlorantraniliprole showed the highest TER (TER=494430) at 35-40 cm for E. fetida under the GS.

maximum and n p'-DDT concentr	nean con ations. "	icentrati NA" = no	ion of pestic	, cides in soils e. Dearadati	(in μg/kg), ι ion products	espectively such as 3.5	v. DDT repre	esented the N-alanine	sum of p, p indicated 3.	'-DDT and o, 5.6-trichloro-
2-pyridinol and	N-(2,6-d	imethyl	ohenyl)-N-(r	nethoxyace	tyl)alanine, r	espectively	·.			,
Pesticides	MSC _{mx}	MSC _{mn}	E. fetida Chronic		E. crypticus Chronic		F. candida chronic		H. aculeifer chronic	
			TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)

Table 5.3a Maximum and mean toxicity exposure ratios (TER _{mx} and TER _{mn}) for soil organisms at 0-5 cm. TER values <5 (in bold)
and >5 indicated risk and no risk, respectively. WS = worst scenario and GS = general scenario. MSC_{mx} and MSC_{mn} indicated the
maximum and mean concentration of pesticides in soils (in µg/kg), respectively. DDT represented the sum of p, p'-DDT and o,
p'-DDT concentrations. "NA" = not applicable. Degradation products such as 3,5,6-TCP and N-alanine indicated 3,5,6-trichloro-
2-pyridinol and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, respectively.

			TER _{mx} (WS)	TER _{mn} (GS)						
3,5,6-TCP	57.4	8.69	80.1	529	NA	NA	NA	NA	NA	NA
Carbendazim	6.45	1.62	155	617	15.5	61.7	NA	NA	NA	NA
Chlorantraniliprole	14.2	3.23	70423	309723	NA	NA	27.5	121	7042	30972
Chlorpyrifos	177	32.5	71.8	391	28.3	154	0.37	2.00	NA	NA
DDT	5.41	3.53	51756	79330	NA	NA	32532	49865	NA	NA
Emamectin	3.30	3.30	606	606	NA	NA	NA	NA	NA	NA
Imidacloprid	13.8	3.94	12.9	45.2	72.5	254	90.6	317	193	678
Metalaxyl	6.49	2.44	6163	16382	NA	NA	19260	51194	NA	NA
N-alanine	2.49	2.49	200803	200803	NA	NA	NA	NA	NA	NA
Profenofos	3.37	1.74	NA	NA	NA	NA	14.8	29	NA	NA

acteetea at the a	cpui, ii	chec un	и пос аррса	i nere.						
Pesticides	MSC _{mx}	MSCmn	E. fetida		E. crypticus	;	F. candida		H. aculeife	r
			Chronic		Chronic		Chronic		Chronic	
			TER _{my} (W/S)	TERma (GS)	TER _{my} (W/S)	TERma (GS)	TFR _{my} (W/S)	TERma (GS)	TER _{mx}	TER _{ma} (GS)
				121(11) (03)		1210111 (03)		1210111 (03)	(WS)	121(1)(05)
3,5,6-TCP	15.9	7.26	289	633	NA	NA	NA	NA	NA	NA
Carbendazim	5.55	2.01	180	498	18.0	49.8	NA	NA	NA	NA
Chlorantraniliprole	4.22	2.06	236967	485546	NA	NA	92.4	189	23697	48555
Chlorpyrifos	68.4	20.2	186	629	73.1	248	0.95	3.22	NA	NA
DDT	29.8	7.98	9402	35088	NA	NA	5910	22055	NA	NA
Imidacloprid	7.57	2.06	23.5	86.6	132	486	165	608	353	1299
Metalaxyl	4.23	2.67	9456	14961	NA	NA	29551	46752	NA	NA

Table 5.3b The TER at 15-20 cm. The pesticide concentration in μ g/kg. N-alanine and profenofos residues were not detected at the depth, hence did not appear here.

Table 5.3c The TER at 35-40 cm. The pesticide concentration in μ g/kg.

Pesticides	MSC _{mx}	MSC _{mn}	E. fetida Chronic		E. crypticus Chronic		F. candida Chronic		H. aculeifer Chronic	
			TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)	TER _{mx} (WS)	TER _{mn} (GS)
3,5,6-TCP	31.1	7.26	148	634	NA	NA	NA	NA	NA	NA
Carbendazim	3.41	2.23	293	448	29.3	44.8	NA	NA	NA	NA
Chlorantraniliprole	6.52	2.02	153374	494430	NA	NA	59.8	193	15337	49443.02
Chlorpyrifos	60.6	26.4	210	481	82.5	189	1.07	2.46	NA	NA
DDT	81.1	13.5	3454	20734	NA	NA	2171	13033	NA	NA
Imidacloprid	31.6	3.35	5.63	53.2	31.7	299	39.6	373	84.5	797.68
Metalaxyl	8.97	2.41	4459	16582	NA	NA	13935	51818	NA	NA
N-alanine	1.56	1.32	320513	379967	NA	NA	NA	NA	NA	NA
Profenofos	1.09	1.09	NA	NA	NA	NA	45.9	45.9	NA	NA

The calculated TER of chlorpyrifos for GS and WS at all depths of soil was lower than the corresponding trigger value which indicated potential risk (TER <5) to non-target soil organisms. It has been observed that MSC_{mx} and MSC_{mn} of chlorpyrifos posed a risk to *F. candida* based on TER_{mx} and TER_{mn}, respectively. Furthermore, chlorpyrifos showed the lowest TER (TER=0.37) at 0-5 cm for *F. candida* under the WS.

5.3.1.2 Ecological risk based on RQ approach

The risk quotient (RQ) values under GS and WS for the studied pesticides are shown in Table 5.4. Chlorpyrifos in all the depths showed RQs >1 for both MSCs geometric mean and MSCs maximum. Profenofos showed RQs >1 for both MSCs at 0-5 cm and 35-40 cm depths. In addition to chlorpyrifos and profenofos, our results suggested a potential risk of imidacloprid to soil organisms at 35-40 cm under the WS. Due to the higher RQ values at all depths and scenarios, the risk posed by chlorpyrifos was worrisome. The top 3 pesticides ranked in decreasing order of their toxicity to the in-soil organisms were: chlorpyrifos > profenofos > imidacloprid, indicating the highest toxicity for organophosphates (OPs).

Table 5.4 The Risk Quotient (RQ) for the most sensible organisms at different depths of soil. RQ values for an individual pesticide at the depth calculated as a ratio of the measured soil concentrations $[(MSC_{mx \ or \ mn}) \ \mu g/kg]$ divided by the PNEC_{mss} $\mu g/kg$ (reported in Table 5.3a-5.3c and Table 5.1). WS = worst scenario and GS = general scenario. The calculated values of RQ were categorised into 4 risk levels: no risk (RQ<0.01), lower risk (0.01≤RQ<0.1), moderate risk (0.1≤RQ<1) and higher risk (RQ≥1). RQ values in bold indicated higher risk. DDT represented the sum of p, p'-DDT and o, p'-DDT concentrations. "NA" = not applicable. Degradation products such as 3,5,6-TCP and N-alanine indicated 3,5,6-trichloro-2-pyridinol and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, respectively.

Destisides			Risk (Quotient (RQ)		
Pesticides	0-5 cm		15-20 cm		35-40 cm	
	RQ _{mx} (WS)	RQ _{mn} (GS)	RQ _{mx} (WS)	RQ _{mn} (GS)	RQ _{mx} (WS)	RQ _{mn} (GS)
3,5,6-TCP	0.62	0.09	0.17	0.08	0.34	0.08
Carbendazim	0.65	0.16	0.56	0.20	0.34	0.22
Chlorantraniliprole	0.36	0.08	0.11	0.05	0.17	0.05
Chlorpyrifos	27.23	4.99	10.52	3.10	9.32	4.06
DDT	<0.01	<0.01	0.01	<0.01	0.02	<0.01
Emamectin	0.08	0.08	NA	NA	NA	NA
Imidacloprid	0.78	0.22	0.43	0.12	1.78	0.19
Metalaxyl	<0.01	<0.01	<0.01	<0.01	<0.01	<0.01
N-alanine	<0.01	<0.01	NA	NA	<0.01	<0.01
Profenofos	3.37	1.74	NA	NA	1.09	1.09

Furthermore, 84% of the sites represented no risk ($\sum RQ_{site} < 0.01$) and 16% showed a higher risk ($\sum RQ_{site} \ge 1$). The highest risk based on RQ_{site} was 30.60 for a conventional farmer's site with the identification code F53 and was indicated for the depth of 15-20 cm (Table S5.3). All IPM sites such as F4, F5, F7, F32, F36, F56, F62, F102, F109, F143 and F158 showed no risk (RQ<0.01). The contribution for the RQ was 98% and 2% for chlorpyrifos and profenofos, respectively. Table S5.3 presents pesticide contributions measured at 8 sites with a higher risk ($\sum RQ_{site} \ge 1$) to the overall risk. Although multiple pesticide residues were observed at a single site, this doesn't infer that pesticides contributed equally to the overall risk posed by the pesticide cocktail. Table S5.3 shows that chlorpyrifos contributed higher than profenofos to the overall risk at one site ($\sum RQ_{site}$). The RQ of OPs at depths (cm) 0-5, 15-20 and 35-40 ranged from 0-11, 0-31 and 0-9, respectively (Table 5.5).

Table 5.5 Correlation matrix between risk quotient (RQsite) of organophosphates (OPs) and knowledge and behaviour (KNB) score at different depths (cm) of soil (n=49). Descriptive statistics of the variables in parenthesis. Values in bold represented either positively or negatively correlated.

· · ·	/	/ 3	/		
Variables (Min-Max; Mean±SD)	RQ _0-5	RQ _15-20	RQ _35-40	KNB	
RQ _0-5 (0-11; 0.34±1.57)	1	0.57*	0.27	-0.44*	
RQ _15-20 (0-31; 0.96±4.45)		1	0.62*	-0.60*	
RQ _35-40 (0-9; 0.25±1.36)			1	-0.38*	
KNB (1-6; 3.63±1.81)				1	

* Correlation is significant at the 0.01 level (2-tailed). OPs included in the analysis were profenofos and chlorpyrifos.

Overall, both the RQ and TER methods seemed conservative as the $PNEC_{mss}$ applied an assessment factor of 10-1000 due to the scarcity of NOEC values and thus considered as the worst-case scenarios. For example, in this study, we considered the RQ_{mn} with geometric

mean of positives only, and the RQ_{mx} with maximum measured soil concentration. Among the 15 compounds that were detected (Table S5.2), a higher risk ($RQ_{mx}>1$) was observed for imidacloprid, profenofos and chlorpyrifos, while a higher risk for sites ($\Sigma RQ_{site}>1$) was posed by chlorpyrifos and profenofos only (Table S5.3).

5.3.2 Perspectives on pesticide risk and farmers' knowledge and behaviour (KNB)

Data on farmers' knowledge and behaviour originated from our previous study (Bhandari et al., 2018). The knowledge score (mean±SD) of farmers about the effect of pesticides on animals, birds, fishes and honeybees was 0.84±0.37, 0.51±0.51, 0.59±0.50 and 0.51±0.51, respectively. Likewise, the behaviour score (mean±SD) of farmers concerning pesticide waste management and correct application was 0.45±0.50 and 0.73±0.45, respectively. The knowledge score ranged from 1 to 4 and the behaviour score ranged from 1 to 2. About 45% of the farmers managed their pesticide packets/containers by burning the waste at a designated area, while 73% of the farmers kept the leftover pesticides and reused them. During a visit to a farmer's field, packets and containers of pesticides were observed simply discarded on the ground, which was unsafe and indicated poor hygiene in fields (Figure 5.1). The total KNB score (mean±SD) for farmers was 3.63±1.81, ranging from 1 to 6. The score indicated a level of awareness of farmers related to pesticide effects and waste management.



Figure 5.1 Unsafe disposal of pesticide packets and containers at fields.

The correlation between the KNB score and the risk quotient (RQ) of pesticides at different farm sites is shown in Table 5.5. The pesticide risk at sites at all depths was negatively correlated (p<0.01) with the KNB score. A significant positive correlation was observed among RQs of organophosphate at depths (cm) 0-5 and 15-20 as well as 15-20 and 35-40.

5.4 Discussion

5.4.1 Pesticide residues in the soils

Although mathematical equations have been used for estimating predicted pesticide environmental concentrations (PEC), the use of real measured concentrations in the agricultural fields (MSC) instead of modelled (PEC) for the pesticide EcoRA provides significant benefits since it gives an accurate measurement of pesticides and includes the inherent heterogeneity of ecosystems (ECOFRAM, 1999). The use of modelled PEC data in the pesticide evaluation process has also been criticized due to its limitation on reflecting the dissipation of pesticides in the environment (Vasickova et al., 2019). Due to higher application rates of pesticides, PEC of most pesticides were much higher than their MSC, indicating that the risk of pesticides could be overestimated due to the use of PEC in the risk assessment processes (Bhandari et al., 2020).

Of all pesticides, chlorpyrifos showed the highest concentration at all the depths. DDT, an organochlorine insecticide has been banned for use in Nepal since 2001 (PQPMC, 2019), however its residues were detected at all the depths as they degrade very slowly in the environment (Boul, 2010). Similar studies (Tan et al., 2020; Xu et al., 2020) done elsewhere showed the highest concentration of carbendazim. Residues of HCH and DDT were detected at higher concentrations in bottom soils from Hong Kong (Zhang et al., 2006). At the 35-40 cm depth, residues of N-alanine were more frequently detected. Risk was due to higher concentrations of DDT and endosulfan, the frequently occurring residues in soil (Yadav et al., 2016). Most frequently detected pesticides in Chinese agricultural topsoil were imidacloprid and emamectin benzoate (Tan et al., 2020). The presence of several pesticides in soil increased the ecotoxicity and caused several effects (Table S5.1), including the death of non-target soil organisms (Cang et al., 2017; Tiwari et al., 2019).

5.4.2 Ecological risk based on TER, RQ including threshold and guidance values

Ecotoxicological data is a prime requisite when performing studies on EcoRA (Frampton, 2000). The ecotoxicity was especially high due to chlorpyrifos residues in soils (Table 5.3a-5.3c). Thomatou et al. (2013) demonstrated a non-acceptable ecological risk due to chlorpyrifos methyl in Greece. In the same study, pesticides with the highest ecotoxicity were organophosphates (OPs). The major contributors to ecotoxicity in many different studies were triazoles (Vasickova et al., 2019), OPs (Wee and Aris, 2017), triazines and OPs

(Palma et al., 2014), and triazoles, carbamates and neonicotinoids (Xu et al., 2020). In our study, OPs most notably contributed to the ecological risk, while the other groups didn't.

The ecological risk based on TER under general scenario (GS) and worst-case scenario (WS) is shown in Table 5.3a-5.3c. Of all pesticides and soil organisms, chlorpyrifos exhibited higher toxicity under both scenarios for *F. candida* due to its higher toxicity persistence in a tropical climate (Watts, 2012). The higher toxicity might be due to higher concentrations of chlorpyrifos and its lower NOEC. Chlorpyrifos was indicated as one of the most acutely as well as chronically toxic pesticides for soil organisms (USEPA, 2009). For imidacloprid, the TER_{mx} value of *E. fetida* at 35-40 cm under the WS was observed closer to its trigger point value which might be due to its exposure to the pesticide. In a multi-level ecotoxicological study, Wang et al. (2019) also demonstrated a toxicity of imidacloprid for *E. fetida*. Of all pesticides and organisms examined in a previous study (de Lima et al., 2017), imidacloprid was found to be the most toxic compound to F. candida, which is known to be very susceptible to pesticides and is one of the most susceptible among soil invertebrates (Fountain and Hopkin, 2005). Although chlorantraniliprole was applied in higher doses than recommended on vegetables such as chillies (Bhandari et al., 2018), it showed greater TER values (Table 5.3a-5.3c), indicating negligible risk at all depths of soil. However, the risk assessment of the pesticide applied to a fruiting vegetable by EFSA presented a chronic risk to F. candida (TER=3) (EFSA, 2013a). In the same study, the ecotoxicity of chlorantraniliprole was observed for *F. candida* in fruits such as grapes (TER=1.9) and pomes (TER=1.4).

Our study presented a contradictory trend for chlorpyrifos and profenofos. Both the pesticides demonstrated higher risks (RQs > 1) for both scenarios: general and worst-case. In a previous study on risk assessment, chlorpyrifos presented a similar trend (Wee and Aris, 2017). The major contribution to the higher risk for 6 sites was only for chlorpyrifos and ranged from 98 to 100%. A similar study (Chen et al., 2020b) indicated that chlorpyrifos and butachlor were the main pollutants. In the arable soils of the Czech Republic, Vasickova et al. (2019) identified that 11% of sites had no risk and 35% of sites had higher risk. One study reported a higher risk in 29% of the sites, mainly due to the use of chemical pesticides that have been banned in Europe (Iturburu et al., 2019). In our study, chlorpyrifos significantly contributed to the overall risk of a site ($\sum RQ_{site}$), a finding similar to a previous study (Wee and Aris, 2017). Chlorpyrifos, profenofos and imidacloprid have an ability to persist in soil (Lewis et al., 2006) and were used in higher doses than the recommended (Bhandari et al., 2018); all of which might have caused moderate to higher ecological risks (RQ). Li et al. (2018) also estimated lower to moderate ecological risks due to simazine. As described by Montuori et al. (2016), the RQ values may be less relevant in other regions because of the variation in seasons, agricultural practices, contamination levels, and distribution of pollutants. For all the pesticides that showed higher risk (RQ>1), a future site-specific risk

assessment is required to better understand the risks of specific pesticides (Pivato et al., 2017). It is noteworthy to mention that all of the sites that were from the integrated farming had no risk ($\sum RQ$ site <0.01); however, 21% of the sites from conventional farming had high risk ($\sum RQ$ site ≥ 1).

The threshold value of DDT in soil (μ g/kg) for the safety of soil organisms is 10 (Jongbloed et al., 1996). The Dutch ecological limit for DDT as well as DDD and DDE in soil is 10 μ g/kg (RIVM, 2015), the maximum permissible concentration in soil based on direct ecotoxicology. The DDT (o, p'-DDT+p, p'-DDT) mean concentration in 35-40 cm from our study area was above 10 μ g/kg (Table 5.3c), indicating the higher ecotoxicity associated with DDT. Furthermore, the mean concentration of p, p'-DDD at the same depth was slightly greater than the Dutch standard for DDD in soil (Table S5.2), which might affect soil organisms. However, the mean concentration of p, p'-DDE in soils did not exceed the standard. The high TER and low RQ values for DDT in Table 5.3a-5.3c and Table 5.4, respectively, further supported the finding. Our previous study (Bhandari et al., 2020) indicated that fungicides such as carbendazim and metalaxyl and insecticides such as chlorpyrifos and imidacloprid had concentration in soils below the guidance values (PSRGVs). However, the current EcoRA showed moderate to high risks to soil organisms from the aforementioned pesticides and did not converge with the guidance values, except for metalaxyl which showed negligible risks. Different methods and databases used in the EcoRA may preclude the effective comparison among the estimated risks (Wang et al., 2009), henceforth our results are less consistent. In the PSRGVs, Ukraine considered $1 \mu g/kg$ as a benchmark for the concentration of profenofos in soil. The profenofos concentrations in 3% of our soil samples were above this benchmark, indicating that ecological risk associated with profenofos pollution should be considered. Furthermore, the PSRGVs for 3,5,6-TCP, N-alanine and chlorantraniliprole were not found, hence their estimated risk (TER and RQ) could not be compared.

5.4.3 Risk quotient and its relationship with farmers' knowledge and behaviour (KNW)

A significant correlation (p<0.01) existed between pesticide risk (RQ) at a farm site and the farmer's KNW related to the effects and management of pesticides. Farmers who were aware of the ecological effects of pesticides and the appropriate waste management measures that should be taken had a negligible pesticide risk at their sites. Although demonstrating no relationship between pesticide risk and knowledge (Lekei et al., 2014a), many other studies have highlighted the significance of a farmer's knowledge regarding pesticide effects and management (Mohanty et al., 2013; Wang et al., 2017a; Wang et al., 2017b; Yang et al., 2014). Farmers were at high risk due to the lack of knowledge and

training related to pesticides (Akter et al., 2018; Atreya et al., 2012). A few farmers randomly disposed of pesticide containers in fields after carrying out pesticide applications (Figure 5.1). A similar disposal method for pesticide packaging was common among farmers in Tanzania (Lekei et al., 2014a) and Ghana (Okoffo et al., 2016). Furthermore, unacceptable levels of exposure to organophosphates caused high ecological risk (RQ>1) in Malaysia (Wee and Aris, 2017). In Costa Rica, an RQ>1 was observed and believed to be due to exposure to pyrethroids (Fournier et al., 2018) and in China, the same can be said for organochlorines (Chen et al., 2020a). Organophosphate pesticides are commonly used in Nepal (Aryal et al., 2014). This study demonstrated that there was a high risk of organophosphate exposure at farm sites and the risk was linked to poor knowledge and behaviour related to chlorpyrifos and profenofos exposure and waste management.

5.4.4 Uncertainty and variability related to risk assessment

In this section, uncertainty and variability regarding the EcoRA performed are recognized and discussed. For every EcoRA, uncertainty and variability is inevitable and the risk cannot be estimated with absolute certainty (USEPA, 2004). Of course, the EcoRA rests on and is limited by the availability of data and handling (i.e. strength and excellence) (Wee and Aris, 2017). Pesticides were not detected in most of the soil samples (due to the limits of detection) (Table S5.2). In our study, data on pesticide concentrations <LOD were excluded which could have led to an overestimation of the mean concentration of pesticides and their associated risks. However, the TER_{mx} and RQ_{mx} of pesticides (values in bold of Table 5.3a-5.3c and Table 5.4) allows us to say that, considering the worst-case scenario, the ecological risk of several pesticides is notably high. Variability in endpoint data and risk assessment models accelerate the uncertainty (Chen, 2005). Uncertainty related to ecotoxicity data and models can be anticipated since we used the NOEC data from studies with known ISO/OECD procedures and the models for risk assessment were based on EFSA methods.

For reducing uncertainties due to variations in soil quality, Dutch EC₅₀ values were corrected for location-specific differences in samples, considering the organic content and clay material in soils (Rutgers et al., 2008). Therefore, the use of NOEC values in this study without any correction for differences in backgrounds such as with Nepalese soil properties, pesticide application practices and meteorological parameters has brought a higher level of uncertainty. Furthermore, the use of international pesticide soil regulatory guidance values for comparisons with measured pesticide concentration (MSC) in our soil samples without corrections could also increase uncertainty. The aforementioned properties, practises and parameters may vary within the country. Generalization of results from the site-specific research conducted in previous studies may not represent the current scenario of risks, thus conclusion should be drawn with caution when comparing findings to other areas of Nepal. Nevertheless, information from this study provides a baseline of pesticide EcoRA for policy makers of Nepal.

Ecotoxicological information about chemical pesticides and their degradation products are not always available and should be incorporated in future risk assessments. For instance, the EcoRA of N-alanine should be considered more in testing protocols because of its persistence in soils. Furthermore, the ecotoxicity of the degradation product of chlorpyrifos was higher than chlorpyrifos itself (Baskaran et al., 2003). Higher tier risk assessment methods directed at improving risk assessments should also be considered to better understand pesticide risk rising from the current (first tier) risk assessment. For comprehensive EcoRA, future studies should also consider acute toxicity which can be derived from using toxicity data such as EC_{50} and an assessment factor of 1000 (Wang et al., 2020). Our study dealt with a few pesticides and limited ecotoxicological information (Table 5.1). It isn't likely that it provided a complete assessment of pesticide risks in the environment. The present risk assessment depends upon deterministic methods and conservative approaches. Various elements including the test organisms, the regional meteorology and hydrology, and the local soil qualities can directly influence pesticide ecotoxicity (Huguier et al., 2015; Jegede et al., 2017). Consequently, studies need to be conducted with recent soil samples and the findings of these studies need to be integrated into Nepalese pesticide risk assessments after developed models are validated.

5.5 Conclusions

- 1. Based on TER and RQ methods under two scenarios: general and worst-case, our results identified chlorpyrifos as the main soil pollutant and revealed the highest ecological risks at all depths, particularly for *Folsomia candida*.
- 2. Pesticide residues in soils revealed the presence of multiple residues (pesticide cocktails) in 35% of the soil samples, including a diversity of pesticide combinations (Bhandari et al., 2020). We considered the additive effect for the pesticides with a similar mode of action, however their interaction in a cocktail may have been the result of either antagonistic or synergistic effects rather than the additive, signifying that the resulting risks due to such effects are unknown.
- 3. Generally, the TER values are only assigned for single compounds, thus only applicable when estimating a potential ecological risk associated with a single compound, hence other methods should be explored.

- To reduce the ecological risk of pesticides, programs such as awareness and capacity building training related to pesticide effects and waste management should be provided to farmers.
- 5. Ecotoxicological research is limited in Nepal. Our findings may be used to argue for a comprehensive EcoRA in future research as well as setting national benchmarks for pesticide concentrations in soil.
- 6. All the IPM sites revealed no ecological risk.

Acknowledgements

We are grateful to the Nuffic for providing the funds to conduct this study (through Project No. 5160957247). We would like to thank Jakub Hofman and Paul van den Brink for providing comments and suggestions during the manuscript preparation. We thank Robin Palmer for reviewing the English language.

Supplementary Material



Figure S5.1 The framework of ecological risk assessment [modified Ying (2018)].

Assessment of hazard and pesticides effect

Literature published from 2000 to 2019 was considered for the assessment. Pesticides and their degradation products in Table S5.2 and organisms such as E. fetida, E. crypticus, F. candida, H. aculifer and N as well as C mineralization microorganisms were considered for the search 1 and 2 separately (Figure S5.2). For 1 and 2, article inclusion criteria were that the titles should contain *toxic* and *effect*, respectively.



129



Figure S5.2 The PRISMA flowchart based on (Moher et al., 2009).

Pesticides	Test organisms/species	Effects	References
3,5,6-trichloro-2-	Earthworms	Low risk	(EFSA, 2005)
Carbendazim	Earthworms/Eisenia fetida	Impact on the weight and reproduction	(Yasmin and D'Souza, 2007)
	Earthworms	DNA damage	(Huan et al., 2016)
	Enchytraeid	Impaired the antioxidant system	(Novais et al., 2014)
	Earthworms	Population of earthworms decreased at higher concentrations	(Burrows and Edwards, 2002)
	Earthworms/Eisenia fetida	Harmful effects on embryo development and genetic factor	(Huan et al., 2016; Rico et al., 2016)
Chlorantraniliprole	Earthworms/Eisenia fetida	Inhibition of growth, reproduction and destruction of biomacromolecules	(Liu et al., 2018)
	Springtails/Folsomia candida	Affect locomotor abilities	(Lavtizar et al., 2016)
Chlorpyrifos	Earthworms/Eisenia fetida	Alterations of SOD activity	(Wang et al., 2012)
	Earthworms/Eisenia fetida	Morphological abnormalities and inhibition of AChE	(Venkateswara Rao et al., 2003)
	Springtails/Folsomia candida	AChE inhibition and oxidative damage	(Jager et al., 2007)
	Mites/Hypoaspis aculeifer	Affect survival and reproduction	(Jaabiri Kamoun et al., 2017)
	Earthworms/Eudrilus eugeniae	Changes in activities of AChE, enzymes related to oxidative stress, contents of LPO, GSH and GST including changes in morphology	(Tiwari et al., 2019)
	Earthworms/Pheretima peguana	Decreases neutral-red retention time (NRRT) for the coelomocyte and inhibits AChE	(Muangphra et al., 2016)
	Earthworms/Eisenia fetida	DNA damage in earthworms	(Casabé et al., 2007)
	Earthworms/Eisenia fetida	Growth retardation	(Zhou et al., 2011)
	Earthworms/A. caliginosa	Inhibited and depressed CbE activity and AChE activity, respectively	(Sanchez-Hernandez et al., 2014)
	Earthworms/L. terrestris	Chlorpyrifos inhibited the activity of cholinesterases and AChE	(Sanchez-Hernandez et al., 2018)
	Earthworms/Eisenia fetida	Effect on development and fecundity in earthworm	(Shi-ping et al., 2007)
	Earthworms/P. excavatus	Effect on growth, development and reproduction	(De Silva et al., 2010)
DDT	Earthworms/Eisenia fetida	Increased GST and CAT activities; inhibition of survival and growth	(Shi et al., 2016)
Emamectin	Earthworms	Low risk	(EFSA, 2012)
Imidacloprid	Earthworms/Eisenia fetida	Inhibition of reproduction	(Wang et al., 2019)
	Earthworms/Eisenia fetida	Avoidance behavior as well as changes in Hsp70 levels	(Dittbrenner et al., 2012)
	Earthworms/Eisenia fetida	Damage of DNA	(Wang et al., 2016)
	Springtails/Folsomia candida	Changes in GST activity, GST mRNA as well as glutathione (GSH) level	(Sillapawattana and Schaffer, 2017)
	Springtails/Folsomia candida	Lethal toxicity (effect on survival and reproduction)	(van Gestel et al., 2017)
	Springtails/Folsomia candida	Reduction in reproduction	(Alves et al., 2014)
	Earthworms/Eisenia fetida	Disruption of a balance between antioxidant enzymatic activities and reactive oxygen species (ROS) contents	(Zhang et al., 2014)
	Earthworms/Eisenia fetida	Effect on reproduction, growth, AChE and DNA	(Wang et al., 2015b)
	Earthworms/Eisenia fetida	Inhibited cellulase activities; damaged cells of epidermal layer and midgut	(Wang et al., 2015a)
Metalaxyl	Earthworms/Eisenia fetida	Accumulation of the pesticide in earthworms (enantioselective)	(Xu et al., 2011)
Profenofos	Earthworms/Eisenia fetida	Neurotoxicity as well as changes in morphology and histology	(Chakra Reddy and Venkateswara Rao, 2008)
	Earthworms/F. candida	Effect on survival, reproduction and hsp70 gene expression	(Liu et al., 2012)
Triazophos	Earthworms/Eisenia fetida	Inhibition of growth and reproduction	(Khandelwal, 2017)
	Earthworms	Oxidative stress, cell damage and tissue injury	(Yang et al., 2019)

Table 55.1 Overview of studies on the pesticides effect in soil invertebrates. The order of pesticides has been sorted alphabetically.

Note.

CbE = Carboxylesterase; AChE = Acetylcholinesterase; SOD = Superoxide dismutase; CAT = Catalase; GST = Glutathione -Stransferase; LPO = Levels of lipid peroxidation; and reduced glutathione (RG).

No information related to effects was available in the literature for pesticides and their degradation products such as quinalphos, N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine, DDE and DDD, henceforth did not appear in the table.

concer	ntration was based on	our	previc	ous stu	idy (Bha	ındari et	al., 20.	20). LO	D (H	g kg ⁻¹)	= Lim	it of d€	stection									
		Soil	l depth	ıs (cm)																		
Group	Pesticides (n=15)·1.00	0-5							15-2(0						35-4	0					
42010		No	%	Min	Max	Median	Mean	SD	8	%	Min	Max	Median	Mean	SD	No No	1 %	Min	Max	Median	Mean	SD
AD	Chlorantraniliprole; 1	16	33%	1.09	14.20	2.86	3.23	3.30	10	20%	1.08	4.22	1.80	2.06	1.21	6	18%	1.08	6.52	1.57	2.02	2.03
BD	Carbendazim; 1	11	22%	1.03	6.45	1.42	1.62	1.56	4	8%	1.10	5.55	1.67	2.01	2.07	e		1.42	3.41	2.30	2.23	1.00
MOD	Emamectin; 1	1	2%	3.30	3.30	3.30	3.30	NA	ΝA	AN	NA	NA	NA	NA	ΝA	ΝA	NA	٨A	NA	NA	NA	NA
DND	Imidacloprid; 1	15	31%	1.02	13.80	2.67	3.94	4.47	9	12%	1.17	7.57	1.71	2.06	2.45	7	14%	1.16	31.60	2.47	3.35	11.04
oc	o, p'-DDT; 1	1	2%	1.60	1.60	1.60	1.60	NA	Ч	2%	4.28	4.28	4.28	4.28	ΝA	1	2%	2.66	2.66	2.66	2.66	NA
	p, p'-DDD; 1	1	2%	1.95	1.95	1.95	1.95	NA	1	2%	8.28	8.28	8.28	8.28	NA	1	2%	11.10	11.10	11.10	11.10	NA
	p, p'-DDE; 1	10	20%	1.00	6.42	1.94	2.07	1.76	4	8%	1.09	13.90	3.90	3.85	5.67	4	8%	1.00	5.55	2.67	2.23	2.31
	p, p'-DDT; 1	4	8%	1.05	3.81	2.13	1.93	1.41	4	8%	1.13	25.50	2.65	3.70	11.72	2	4%	1.50	78.40	39.95	10.84	54.38
OP	Chlorpyrifos; 10	S	10%	13.10	177.00	25.90	32.45	69.49	4	8%	10.50	68.40	15.25	20.18	27.48	2	4%	11.50	60.60	36.05	26.40	34.72
	Profenofos; 1	ŝ	89	1.10	3.37	1.42	1.74	1.23	ΝA	AN	NA	NA	NA	NA	ΝA	1	2%	1.09	1.09	1.09	1.09	NA
	Quinalphos; 1	2	4%	1.23	2.47	1.85	1.74	0.88	Ļ	2%	1.06	1.06	1.06	1.06	ΝA	ΝA	NA	٨A	NA	NA	NA	NA
	Triazophos; 1	ŝ	%9	1.05	8.12	1.43	2.30	3.98	5	4%	1.86	2.35	2.11	2.09	0.35	1	2% 4	4.84	4.84	4.84	4.84	NA
ΡA	Metalaxyl; 1	∞	16%	1.12	6.49	2.78	2.44	1.87	2	4%	1.69	4.23	2.96	2.67	1.80	ഹ	10%	1.20	8.97	1.94	2.41	3.24
UNC	3,5,6-TCP; 2.5	16	33%	2.63	57.40	8.89	8.69	13.06	6	18%	3.03	15.90	5.94	7.26	4.75	11	22%	3.02	31.10	7.13	7.26	8.04
	N-alanine; 1	Ч	2%	2.49	2.49	2.49	2.49	AN	ΝA	AN	NA	NA	NA	ΝA	ΝA	2	4%	1.11	1.56	1.34	1.32	0.32
Note. N-alanii	ne = N-(2,6-dimethylphe	nyl)-ľ	N-(met	hoxyac	etyl)alan	ine; Pesti	cides g	roup: A) = 0	Anthra	nilic di	amide;	UNC = I	Jnclass	ified;	= HC	Organo	dsoydd	hate; l	= QNN		

Table S5.2 Descriptive statistics of pesticides and their degraded product concentrations (µg/kg) at different depths of soil. "NA" = not applicable. The

Ecological risk assessment of pesticide residues in soils from vegetable production areas: A case study in S-Nepal

Neonicotinoid; OC = Organochlorine; BD = Benzimidazole; PA = Phenylamide; and MOD = Micro-organism derived.

Table S5.3 Sum of the risk quotient (RQ) and contribution (%) of the organophosphate (mixture of 2 pesticides) at the different depths. The sum of RQ ($\sum RQ_{site}$) of a given site was computed by summing up all the RQs for each of the pesticide residues \geq LOD. The contribution (%) was not available for pesticides concentrations <LOD. The calculated values of $\sum RQ_{site}$ were categorised into 4 risk levels as: no risk ($\sum RQ_{site} < 0.01$), lower risk ($0.01 \le \sum RQ_{site} < 0.1$), moderate risk ($0.1 \le \sum RQ_{site} < 1$), and higher risk ($\sum RQ_{site} \geq 1$). Because of privacy, the location of the sites could not be shown on a map. CHL = Chlorpyrifos and PRP = Profenofos.

					Soil dept	hs (cm)						
Sites	0-5				15-20				35-40			
(n=49)	Sum of		% contrib	outed	Sum of		% contri	buted	Sum of		% contri	buted
	the RQ				the RQ				the RQ			
	based				based				based			
	on the				on the				on the			
	CA				CA				CA			
	effect	Risk	CHL	PRP	effect	Risk	CHL	PRP	effect	Risk	CHL	PRP
F4	0.00	No			0.00	No			0.00	No		
F5	0.00	No			0.00	No			0.00	No		
F7	0.00	No			0.00	No			0.00	No		
F8	0.00	No			0.00	No			0.00	No		
F19	0.00	No			0.00	No			0.00	No		
F21	0.00	No			0.00	No			0.00	No		
F22	0.00	No			0.00	No			0.00	No		
F30	0.00	No			0.00	No			0.00	No		
F32	0.00	No			0.00	No			0.00	No		
F34	0.00	No			0.00	No			0.00	No		
F35	0.00	No			2.77	Higher	100%		1.77	Higher	100%	
F36	0.00	No			0.00	No			0.00	No		
F40	0.00	No			0.00	No			0.00	No		
F52	0.00	No			0.00	No			0.00	No		
F53	1.62	Higher	100%		30.60	Higher	98%	2%	0.00	No		
F55	0.00	No			1.42	Higher		100%	0.00	No		
F56	0.00	No			0.00	No			0.00	No		
F58	0.00	No			0.00	No			0.00	No		
F59	0.00	No			0.00	No			0.00	No		
F60	0.00	No			0.00	No			0.00	No		
F61	0.00	No			0.00	No			0.00	No		
F62	0.00	No			0.00	No			0.00	No		
F76	0.00	No			0.00	No			0.00	No		
F77	0.00	No			0.00	No			0.00	No		
F83	0.00	No			0.00	No			0.00	No		
F102	0.00	No			0.00	No			0.00	No		
F103	0.00	No			0.00	No			0.00	No		
F109	0.00	No			0.00	No			0.00	No		
F129	0.00	No			0.00	No			0.00	No		
F133	0.00	No			0.00	No			0.00	No		
F136	0.00	No			0.00	No			0.00	No		
F140	0.00	No			0.00	No			0.00	No		
F143	0.00	No			0.00	No			0.00	No		
F144	0.00	No			0.00	No			0.00	No		
F146	0.00	No			0.00	No			0.00	No		
F148	0.00	No			0.00	No			0.00	No		
F149	0.00	No			2.02	Higher	100%		0.00	No		
F155	2.55	Higher	100%		3.98	Higher	100%		0.00	No		
F156	0.00	No			0.00	No			0.00	No		
F157	0.00	No			0.00	No			0.00	No		
F158	0.00	No			0.00	No			0.00	No		
F163	0.00	No			0.00	No			0.00	No		

-	0-5				15-20				35-40			
			% contril	outed	% contributed					% contributed		
	Sum of				Sum of				Sum of			
Sites	the RQ				the RQ				the RQ			
	based				based				based			
	on the				on the				on the			
	CA	Risk			CA	Risk			CA	Risk		
	effect		CHL	PRP	effect		CHL	PRP	effect		CHL	PRP
F167	0.00	No			1.10	Higher		100%	1.09	Higher		100%
F169	0.00	No			0.00	No			0.00	No		
F170	0.00	No			0.00	No			0.00	No		
F173	0.00	No			0.00	No			0.00	No		
F179	0.00	No			0.00	No			0.00	No		
F180	2.14	Higher	100%		0.00	No			0.00	No		
F183	10.52	Higher	100%		5.12	Higher	100%		9.32	Higher	100%	
Legend		No risk			Higher ri	sk						

Table S5.3 Sum of the risk quotient (RQ) and contribution (%) of the organophosphate (mixture of 2 pesticides) at the different depths......(continued).

6. Synthesis

This thesis mainly focussed on understanding (i) pesticide use and the related safety behaviour among farmers and retailers, (ii) dietary and (iii) non-dietary risks due to pesticide residues in vegetables and soil, respectively, and (iv) the ecological risks posed by pesticides. Here, I present the thesis findings according to the key objectives stated in the general introduction section and schematically shown in Figure 6.1.



Figure 6.1 The outline of this thesis with the learning objectives as noted in the rectangular boxes. Based on the application of pesticides (Chapter 2), the eggplant, chilli and tomato samples were analysed and dietary risk assessment was carried out (Chapter 3). The cultivated soils were analysed for non-dietary health risks (Chapter 4). The ecological risk of pesticides in soil was examined (Chapter 5). Data between parenthesis are italicized, representing sampling units. IPM = integrated pest management; KAB = knowledge and behaviour.

6.1 General discussion

6.1.1 Pesticide use on vegetable crops and safety behaviour of farmers and pesticide retailers

Farmers used mainly insecticides and fungicides on vegetable crops. On average, 3 active ingredients were used per vegetable. Farmers mostly applied pyrethroids, organophosphates and carbamates on different vegetable crops such as eggplant, chilli,

tomato, bottle gourd, bean, sponge gourd, broccoli, cauliflower, okra, cucumber, cabbage, and bitter gourd. The most frequently used insecticides were cypermethrin (76% of the farmers) and profenofos (>55% of the farmers). Similarly, 54% of the farmers applied the fungicide, mancozeb. However, insecticides such as carbofuran and cartap hydrochloride from the groups such as carbamate and thiocarbamate, respectively were applied the least often. Our results showed that farmers' application of pesticides on eggplants, tomatoes and chillies was much higher than their recommended doses (Chapter 2). Our results also observed that banned insecticides such as endosulfan and phorate were also applied. The use of banned pesticides in agriculture was also reported in Ethiopia (Haylamicheal and Dalvie, 2009) and India (Devi et al., 2011; Joseph et al., 2020). The average pesticide use for vegetables was 2.9 kg a.i./ha, however it depended on the types of vegetables and pesticides (Chapter 2). Organophosphate was excessively used for vegetables (for example, dichlorvos for eggplants was 23.12 kg a.i./ha). Farmers used several fungicides in vegetable farming, however the application of mancozeb was found to be the highest (7.78 kg a.i./ha). Our results showed higher rates of pesticide use in vegetables compared to previous studies in other regions (Chhetri et al., 2014; Jha and Regmi, 2009).

Pesticides can be very useful when used wisely. Integrated farming based on IPM methods uses pesticides judiciously. In our study, very few farmers adopted such methods, for example, use of ash, pheromone trap, compost manure, animal dung and urine for vegetable farming. Farmers and pesticide retailers often lack sufficient knowledge on the safe use of pesticides and eco-friendly methods for pests control (Akter et al., 2018; Atreya, 2007; Gesesew et al., 2016; Mohanty et al., 2013; Oesterlund et al., 2014; Rijal et al., 2018; Vaidya et al., 2017). Our results showed that most farmers overused pesticides in vegetable and practiced conventional farming. They regularly sprayed chemical pesticides even when there were no sign of plant diseases. Farmers mixed more than one pesticide and sprayed cocktails, which is more harmful than spraying a single pesticide (PAN, 2020). Farmers were uneducated and untrained and did not follow the minimum safety precautions while working, thus exposing themselves to pesticides. The EU regulation 2016/425 states the significance of personal protective equipment (PPE) to protect from various forms of exposure (EU, 2016). Safety behaviours such as wearing proper PPE can reduce pesticide exposure (Yarpuz-Bozdogan, 2018) and associated risks (Houbraken et al., 2016). The use of safety measures helped to avoid about 44% of pesticide-related illness in Vietnam (Dasgupta et al., 2007). The level of education and knowledge of farmers and retailers on pesticide use determined their safety behaviour (Ali et al., 2020; Sapbamrer and Thammachai, 2020). Furthermore, climate (Watson et al., 2019) and economic conditions (Damalas et al., 2006; Garrigou et al., 2020) influenced the safety behaviour of farmers. Our study also found that farmers suffered from headaches and irritation, two of the most frequently reported short-term illnesses due to pesticide handling. Previous studies also

reported chronic health problems, particularly related to reproduction, hormone disruption and mutation (Gupta, 2004; Kim et al., 2017; Sabarwal et al., 2018).

Age, education, individual beliefs, and cues to action determined the safety behaviour of farmers (Chapter 2) (Figure 6.2). Farmers perceived higher barriers for the safety measures thus used hats, long-sleeved shirts, long pants, gloves, and masks less frequently during pesticide handling. The perceived barriers mentioned above refer to the suggested clothing as being uncomfortable or unavailable, or farmers were simply unaware of the benefits of wearing such clothing during pesticide application. For example, wearing long pants and long-sleeved shirts while spraying pesticides caused farmers discomfort due to overheating. The perceived barriers were also found to inhibit the safety behaviour of farmers in Iran during pesticide handling (Sharifzadeh et al., 2019).

In our study, almost all farmers perceived the benefits of washing their hands after spraying, however a few of them did not take baths (15%) or wash their clothes (24%) after spraying. Furthermore, there were farmers who preferred to drink, smoke and eat during pesticide application which increases the risk of exposure. Educated farmers perceived higher pesticide threats and were aware of the long-term effects of pesticides on humans, animals, birds, fishes, and honeybees. Self-efficacy and cues to action had positive effects on farmers' safety behaviour. Educated and trained farmers in China (Fan et al., 2015) and Bangladesh (Akter et al., 2018) had good protective behaviours. Most farmers in our study were neither educated nor trained, thus 90% of them did not know the colour codes used for pesticides. Of all the factors, perceived barrier was the most reliable predictor of a farmer's safety behaviour (Sharifzadeh et al., 2019). Government authorities should concentrate efforts on minimizing the barriers to safety behaviour for farmers.

Perceived threats and cues to action also determined the safety behaviour of pesticide retailers (Chapter 2). Although 100% of the retailers perceived higher threats to their health and environmental entities from pesticides, some retailers (33%) did not believe the long-term detrimental health effects of pesticide exposure. A few farmers even stated that pesticides were "medicine" instead of "poison". Furthermore, some retailers (16%) did not think that they were exposed to the pesticides while handling them. Retailers who registered their businesses and renewed their licences regularly had good safety practices while handling pesticides, a finding consistent with a past study conducted among Tanzanian retailers (Lekei et al., 2014b). Retailer participation in the meetings, seminars, and workshops on pesticide use triggered their safety behaviour, however only 47% of them were given any opportunity to follow any of these events. Any training related to the safe handling of pesticides motivated retailers to improve their safety actions.



Figure 6.2 Results based on the Health Belief Model for understanding the safety behaviour of farmers and retailers. The factors that affected the pesticide safety behaviour of farmers and retailers positively (+ve) are in bold.

6.1.2 Dietary risk for humans

The survey (Chapter 2) found overuse of pesticides on eggplants, chillies and tomatoes. Pesticide residues can be taken up by plants or stay on the soil surface after application. Dietary consumption of these residues may contribute to human exposure. We tested the vegetable samples for 23 pesticides (along with their degradation products). Of the pesticides, only 14 were detected in the vegetables, and we examined dietary risks of these pesticide residues in adults and adolescents (Chapter 3). In our study, 100% of the tomato and chilli and 93% of the eggplant samples contained pesticide residues. About 7% of the eggplant samples were free from pesticide residues. Carbendazim (fungicide) in tomato was significantly higher than chilli and eggplant. Chlorpyrifos (insecticide) residues exceeded the EU maximum residue limits (MRLs) in 11% of the eggplant and 19% of the chilli and tomato samples. Although 170 active ingredients are registered and used in Nepal, the government only sets pesticide limits on 36 active ingredients in foodstuff. Of the pesticide residues detected in our study (Chapter 3), Nepalese MRLs for food grains, legumes and pulses were available for only 3 pesticides, namely dichlorvos (1000 μ g/kg), chlorpyrifos (50 μ g/kg) and carbendazim (500 μ g/kg) (Lama, 2008).

In comparison to eggplant and chilli samples, a higher number of residues were present in tomato samples. Up to 7 residues of different pesticides were present in 6% of the tomato

samples from conventional farming. Furthermore, multiple pesticide residues were detected in eggplants (56% of the samples), chillies (96% of the samples), and tomatoes (100% of the samples). Studies done elsewhere also observed contaminated fruits and vegetables with multiple pesticide residues (Jallow et al., 2017b; Liu et al., 2016a; Liu et al., 2016b).

Carbendazim (fungicide) and chlorpyrifos (insecticide) were detected most frequently in vegetables, a finding consistent with a past study (Quijano et al., 2016). A recent study conducted in Saudi Arabia contained mainly insecticides and fungicides in 100% of the chilli, 96% of the tomato and 85% of the eggplant samples studied (Ramadan et al., 2020). Both, fungicides and insecticides were major groups detected in Turkey (Golge et al., 2018). Of all foods tested by Koirala et al. (2007), a higher number of vegetable samples were contaminated with higher concentrations of organochlorines (i.e. BHC and DDT) and organophosphates (i.e. malathion and methyl-parathion). Although Nepal has banned endosulfan and methyl-parathion insecticides for their use in agriculture, Dhakal (2016) detected residues of those insecticides in fruits, indicating current application of banned pesticides in the country. Our study also detected pesticides that are not approved for use in the EU: dichlorvos, profenofos, quinalphos, triazophos and carbendazim.

Integrated Pest Management (IPM) motivates farmers to decrease the use of chemical pesticides (Gautam et al., 2017). Therefore, the concentration of insecticides and their degradation products such as imidacloprid, chlorpyrifos and 3,5,6-trichloro-2-pyridinol (TCP) in vegetables from IPM fields were significantly lower (p < 0.05) than the insecticides and their degradation products found on the vegetables from conventional fields. Furthermore, the total average concentration of the detected pesticides on vegetable samples from IPM fields (39.5µg/kg) was significantly lower (p < 0.05) than the pesticide concentrations found on the vegetables from conventional fields. Furthermore, the total average concentration of the detected pesticides on vegetable samples from IPM fields (39.5µg/kg) was significantly lower (p < 0.05) than the pesticide concentrations found on the vegetables from conventional fields (196 µg/kg). Foodstuffs from IPM fields were either free from pesticides (Polat and Tiryaki, 2019) or had lower concentrations of insecticides and fungicides (i.e. below MRL) than non-IPM fields (Baker et al., 2002; Mladenova and Shtereva, 2009; Singh et al., 2009).

Triazophos exceeded the European Union Maximum Residue Limit (EU MRLs) in 4% of eggplant and 6% of tomato samples. Likewise, chlorpyrifos exceeded the EU MRLs in eggplants (11% of the samples), chillies (18.5% of the samples) and tomatoes (19% of the samples). Both omethoate and carbendazim in 3% of the tomato samples exceeded the EU MRLs; all indicating poor agricultural practices. Overall, 44% of the tomato samples exceeded the EU MRLs. All the vegetable samples exceeding the EU MRLs were from conventional farming. A detailed list of pesticide concentration ranges in vegetables with their EU MRLs can be found in Chapter 2. Of all detected pesticides, imidacloprid

significantly exceeded its MRLs in fruits and vegetables in a study performed in Kuwait (Jallow et al., 2017b). The MRLs were set for just single pesticides at a time, ignoring the "cocktail effect". In our study, vegetables showed a group diversity of pesticide combinations, which makes it challenging to protect humans the risks of pesticide exposure. Farmers used pesticides in much higher rates than recommended on vegetables such as tomatoes, chillies and eggplants (Chapter 2), therefore we estimated dietary risk of pesticides due to the consumption of these vegetables. The dietary risk assessment of the pesticides indicated that consumption of tomatoes could cause potential health risks due to mainly organophosphates such as chlorpyrifos and triazophos. The risk might be due to the higher persistency and application rates of these compounds. Chlorpyrifos used on fruits and vegetables caused health risks elsewhere (Kariathi et al., 2016; Lehmann et al., 2017; Mojsak et al., 2018; Reiler et al., 2015). Organophosphate residues contributed health risks due to the consumption of tomatoes in Bolivia (Reiler et al., 2015). Its dietary risk for humans was demonstrated elsewhere (Abdelbagi et al., 2020; Kariathi et al., 2016; Lehmann et al., 2017; Mojsak et al., 2018; Salamzadeh et al., 2018). However, the risk was of less concern in the EU regions (Jensen et al., 2015; Quijano et al., 2016). In the studies, pesticide concentrations in foodstuffs didn't pose health risks for the Danish and Spanish consumers, indicating their lower estimated short-term intake (ESTI) and estimated daily intake (EDI) than the acute reference dose (ARfD) and the acceptable daily intake (ADI), respectively. In our study, adults were at higher risks from dietary exposures than adolescents which might be due to consumption of more food by adults than adolescents. The exposure to a pesticide is dependent on its dose in food (ICAR, 2019), thus the more you eat, the greater your potential risk. Generally, the MRL, ADI and ARfD values are only assigned for single active ingredients, thus only applicable when estimating a potential risk associated with a single residue of a pesticide. However, when estimating the risk posed by multiple pesticide residues, there is still doubt whether the risk estimation methods currently used, which were based on single MRL, ADI and ARfD values, can be considered accurate. Multiple pesticide residues in foods could lead to synergistic or antagonistic effects once they are absorbed in the human body (Prutner et al., 2013).

6.1.3 Non-dietary human health risk

As mentioned in Chapter 1, pesticides that are persistent in soil remain at different depths and concentrations due to their physical and chemical properties, as well as environmental characteristics (WHO, 2008). Overall, 15 pesticide residues, including their degradation products, were detected in three different soil depths (cm): 0-5, 15-20 and 35-40 (Chapter 4). The residues were more frequently observed in upper layers of soil. From the 147 soil samples, about 60% contained pesticide residues with a mean of 16 µg kg⁻¹, ranging from

1.0 µg/kg to 251 µg/kg. Overall, 35% of the soil samples had multiple pesticide residues, with 2% of the upper layer soils from conventional farming containing up to 7 residues per sample. Among organophosphates (OPs), chlorpyrifos was the most frequently occurring insecticide followed by triazophos, profenofos and guinalphos with their concentrations of (µg kg 1) 40.8, 3.28, 1.75 and 1.59, respectively. Soil samples contained organochlorine (OC) residues mainly o, p'-DDT (2% of the samples); p, p'-DDE (12% of the samples); p, p'-DDD (2% of the samples) and p, p'-DDT (7% of the samples). Of all the DDTs, residues of p, p'-DDE were most frequently found in soils, while p, p'-DDT residues had higher concentrations with an average of 12.1 µg kg⁻¹. The isomeric ratio of DDTs suggested ongoing use of DDT, a finding consistent with a past study (Yadav et al., 2016). DDT has been banned in Nepal since 2001, but due to the persistent nature of it, the residues are still present in soils. From the different chemical pesticide groups, OC pesticides are very persistent, toxic and have the tendency to bioaccumulate and biomagnify (Briz et al., 2011; Contreras López, 2003; Gao et al., 2013). The other pesticides and their degradation products such as chlorantraniliprole, imidacloprid carbendazim, metalaxyl, emamectin, N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine and TCP had their average concentration (µg kg⁻¹) in soil as 3.17, 5.21, 2.12, 3.04, 3.30, 1.72 and 10.3, respectively. Furthermore, higher precipitation can wash out soil pesticide residues and ultimately contaminate other sources (Pokhrel et al., 2018). For example, in our study, a few of the soil samples from IPM fields had pesticide residues.

Human health risk assessment consists of acute and chronic risk due to organophosphate (OP) and organochlorine (OC) residues in soil, respectively. The soil residues may reach humans via different exposure pathways such as inhalation, ingestion and dermal contact. Our findings suggest negligible non-cancer risk of OP residues in soils to humans through ingestion and dermal exposure (HQ and HI <1) which was consistent with a study in China (Pan et al., 2018). Toxicological reference values for profenofos and degradation products such 3,5,6-trichloro-2-pyridinol [TCP] N-(2,6-dimethylphenyl)-Nas and (methoxyacetyl)alanine [N-alanine] were not available in the databases we referred to, thus their risk assessment was not possible. However, concentrations of profenofos including pesticides were compared with their globally other detected established guidance/threshold values in soils (Chapter 4). Similarly, our findings suggested negligible cancer risk (considering all the aforementioned pathways), particularly due to OC residues as the estimated value of total incremental lifetime cancer risk was $<1\times10^{-6}$. Chapter 4 includes all the details of the USEPA procedures and parameters used in the health risk assessment. DDT and its degradation products were extensively detected in Chinese soils (Gong et al., 2004; Hu et al., 2014; Ma et al., 2016), thus a potential cancer risk was observed (Qu et al., 2015). Previous studies have established relationships between pesticides and their human health consequences (Ennaceur et al., 2008; Shinomiya and Shinomiya, 2003),
including cancer (Ennour-Idrissi et al., 2019; Louis, 2019) and non-cancer risks such as hormone disruptions and reproductive problems (Frye et al., 2012; Younglai et al., 2004). Although health risk due to pesticides in soil was of less concern, adolescents were exposed relatively more than adults which might be due to adolescent body weights being relatively less than that of adults. Of all the exposure pathways, potential exposure to OP was mainly through ingestion.

For estimating non-dietary health risk due to pesticide mixtures, we considered the additive effect for the pesticides with a common mode of action. However, residual interactions in a mixture may have been the result of either antagonistic or synergistic effects rather than the additive effects (Alexander et al., 2008), which means that the resulting risks due to such effects are unknown.

6.1.4 Ecological risk of pesticide residues detected in soil

Earthworms play an important role in soil fertility maintenance by influencing organic content, soil structure and microscopic organisms (Fragoso et al., 1997; Hole et al., 2005). However, chemical pesticides induce negative effects on ecosystems and ecosystem services (Carriquiriborde et al., 2014; Utsumi et al., 2011). In Chapter 5, we assessed ecological risks of 9 pesticides from different groups, including their degradation products such as 1 organochlorine insecticide (DDT), 2 organophosphate insecticides (profenofos and chlorpyrifos), 1 phenylamide fungicide (metalaxyl), 1 neonicotinoid insecticide (imidacloprid), 1 anthranilic diamide insecticide (chlorantraniliprole), 1 benzimidazole fungicide (carbendazim), and 2 unclassified group metabolites (TCP and N-alanine) at 3 layers (cm) of soil: 0-5, 15-20 and 35-40. The other pesticides such as DDD, DDE, quinalphos and triazophos were detected in soils but excluded from the risk assessment as their ecotoxicity data was not available in literature. The risk assessment was based on the EFSA standard procedures and test organisms such as the earthworm (Eisenia fetida), the enchytraeid (Enchytraeus crypticus), the springtail (Folsomia candida), the mite (Hypoaspis aculifer) and N as well as C mineralization microorganisms. Therefore, the toxicity exposure ratio (TER) for the aforementioned pesticides was estimated for assessing ecotoxicity at 3 depths of soil, while the risk quotient (RQ) was estimated for assessing individual and mixture toxicity of pesticides. The ecological risk of pesticides was assessed by comparing their TER with trigger values and international standards that have been adopted elsewhere (Devi et al., 2015; Vasickova et al., 2019; Wee and Aris, 2017). Furthermore, we compared the measured concentrations of pesticides with their corresponding pesticide soil regulatory guidance values.

TER values of chlorpyrifos, an organophosphate insecticide, notably contributed chronic toxicity to F. candida for all 3 depths (cm) (0-5, 15-20 and 35-40). However, the TER values of other pesticides for the depths indicated no significant risk of the pesticides for E. fetida, E. crypticus, and H. aculifer. Moreover, insecticides such as profenofos and chlorpyrifos contributed risk for 16% of the sites. Organophosphate and neonicotinoid insecticides contributed risk to soil organisms elsewhere (Chagnon et al., 2015; Giesy et al., 2014). Based on RQ, our study showed a potential ecological risk for the soil organisms due to 3 insecticides, profenotos at depths (cm): 0-5 and 35-40; imidacloprid at a depth of 35-40; and chlorpyrifos at all the depths. Additionally, compared with the Ukrainian standards for profenofos in 3% of the soil samples, the ecological risk associated with the pesticide cannot be ignored. High ecological risk due to insecticides was observed for the river ecosystems in Malaysia (Wee and Aris, 2017). However, fungicides and herbicides in the Czech Republic and China, respectively posed ecological risks (Li et al., 2018; Vasickova et al., 2019). It was found that ecotoxicity, particularly the TER associated with DDT and metalaxyl, was negligible. The findings converge with the results based on their RQ, where the pesticides did not show a significant risk. A few compounds such as p,p'-DDD, p,p'-DDE, quinalphos and triazophos were discarded from the risk assessment due to the unavailability of their ecotoxicity data, hence neither their RQs nor TERs were known. Pesticide cocktails in soil resulting from diverse groups can affect soil organisms. Our study was primarily based on the additive effects of pesticides, particularly organophosphate insecticides, therefore, ecological risk of pesticides due to other effects such as synergistic and antagonistic is still unknown. Several pesticide packets and containers were observed at different sites during the survey, indicating unsafe behaviour of farmers. Such behaviour was common among Ethiopian farmers, with the discarded containers polluting soil and water due to the residual chemicals left in them (Meguanint et al., 2019).

The summative index of farmers' knowledge and behaviour was correlated with the mixture toxicity of organophosphates, RQ (Σ RQ_{mix}), indicating that the ecological risks of the pesticides at sites is negatively associated with the pesticide knowledge and safety behaviours of farmers. Moreover, the risk can be reduced by increasing farmers' knowledge of pesticide effects on health and the environment. Enhancing awareness and providing pesticide use and safety training among farmers can minimize the environmental risks of pesticides.

6.2 General conclusions and major findings

This thesis focussed on the factors affecting safety behaviour of farmers and retailers, dietary and non-dietary risk of pesticide exposure and assessment of ecological risk of

pesticide use. The EFSA risk assessment procedures were applied because of the lack of such procedures for Nepal. The use of chemical pesticides is increasing in Nepal and unfortunately, farmers have been overusing pesticides, especially in vegetable farming, ultimately leading to health and environmental degradation. This study focussed on risk assessment of a pesticide (HQ or RQ) as well as pesticide mixtures resulting from a combined exposure (HI) based on information on individual pesticides with a similar mode of action-additive effects. As there is a lack of sufficient information on synergism and antagonism of complex mixtures of pesticides in vegetables and soils, our study could not examine the combined effects further and thus the effects are still unknown. However, this study provides a baseline risk for the current use of pesticides in Nepal. The knowledge gained from this research is crucial for enhancing the safety behaviour of farmers and retailers and is also important for minimising dietary and non-dietary exposure to pesticides from vegetables and soil, respectively.

The major findings are:

- 1. Very few farmers in our study practiced IPM techniques. Farmers practicing conventional farming used almost all groups of pesticides in vegetable cultivation. Of the pesticides used, 78% were insecticides, 16% were fungicides and 6% were herbicides. We estimated higher application rates for insecticides and fungicides than the recommended doses. We found over 30 active ingredients being used in vegetable farming. However, farmers were unaware of the negative effects of pesticides and pesticide toxicity labels, thus they often used banned pesticides. The average application of fungicides such as mancozeb and insecticides such as dichlorvos were the highest. The highest application rates were observed in eggplant, chilli, and tomato. Interestingly, retailers had a good knowledge of pesticides and their health and environmental consequences, however, they lacked sufficient safety measures at their shops. Therefore, both stakeholders have a higher risk of pesticide exposure. Education and training support for the farmers and retailers, and implementation of legal framework, are important for the safety of human health and the environment.
- 2. Vegetables from IPM fields contained notably lower concentrations of pesticides than from conventional fields, indicating that the IPM foods were comparatively safer. Up to 7 different pesticide residues were detected in individual sample from conventional farming forming pesticide cocktails. The percentage of vegetable samples containing pesticide cocktails in conventional and IPM fields was 88 and 47, respectively. In eggplants, tomatoes and chillies from the conventional fields, the concentration of insecticides (i.e. triazophos, omethoate and chlorpyrifos) and a fungicide (i.e. carbendazim) exceeded the EU MRLs, indicating poor agricultural

practices. The estimated dietary intake of pesticide residues (ESTI, mg kg⁻¹ body weight day⁻¹) through consumption of vegetables, particularly tomatoes from conventional fields, was higher than their human health based toxicological reference values (ARfD, mg kg⁻¹ body weight day⁻¹), posing acute health risk for consumers. The dietary risk was mainly due to organophosphate insecticides such as chlorpyrifos and triazophos.

- 3. Pesticide residues were more frequently detected in the upper soil layer. Ingestion was the main pathway of pesticide exposure. However, non-dietary consumption of their residues did not pose a significant human health risk, including cancer and non-cancer. The pesticides in soils from non-IPM fields had notably higher concentrations of residues than IPM fields. The predicted concentration (PEC) of most pesticides in soil based on the EFSA procedures were much higher than their measured soil concentrations, thus their predicted values do not coincide with real field measurements.
- 4. Insecticides, mainly profenofos, imidacloprid and chlorpyrifos, posed ecological risks for soil organisms particularly in conventional farms/sites. *F. candida* was the most sensitive to pesticides. There was also an association between the increase in the ecological risk due to the inadequate knowledge of farmers and their lack of safety measures with regards to pesticide use. The cocktail effect of pesticides was unknown.

6.3 Implications and recommendations

6.3.1 Implications for management of pesticide use and safety practices

This study identified factors affecting pesticide use safety behaviour of farmers and retailers. Government agencies should provide education on and awareness of good agricultural practices, especially the safe use of chemical pesticides for farmers, and offer training for the retailers on pesticide alternatives. The training packages would enhance retailers' knowledge and capacity and encourage them to follow precautionary measures while handling pesticides. This information will help policymakers to make effective pesticide policies on pesticide use and safety behaviour in Nepal. Illegal import of banned pesticides through open Nepal-India boarders warrants severe punishment. When pesticide application doses exceeded their recommended rates, harmful effects to humans and the environment are inevitable (Meena et al., 2020). Good agricultural practices such as proper pesticide application, including safe pre-harvest intervals, and following recommended

doses and application frequencies could increase agricultural production without hampering human health and the environment.

6.3.2 Implications for developing food safety and promoting integrated farming

To meet an increasing demand for food, famers have been using many chemical pesticides to protect their crops from pests and plant diseases. Globally, pesticide application has minimized crop loss and increased agricultural production. However, pesticide persistence and farmers' poor agricultural practices have raised several issues, including residues of pesticides in foods. While food safety has been considered a serious issue in developed nations, developing countries such as Nepal are still using toxic and persistent chemical pesticides on vegetable crops that are banned in the EU. Toxicological and risk assessment studies have been conducted worldwide and authorities have established pesticide MRLs in food commodities to ensure food safety. However, the government of Nepal has not yet established pesticide MRL legal framework for the most common foodstuffs, including vegetables, that would prevent consumer exposure to pesticides. Pesticide toxicity and risk assessment should also be conducted to achieve agricultural sustainability and food safety (Damalas and Koutroubas, 2016). However, risk assessment tools as well as comprehensive safety levels of pesticides in Nepalese foods have not yet developed.

The current use of OP insecticides, especially chlorpyrifos and triazophos, has a negative effect on human health via consumption of vegetables, particularly tomatoes. The vegetables from non-IPM fields contained higher concentrations and numbers of pesticides than IPM fields, therefore, this study recommends extensive IPM programs focussing especially on rural vegetable growers. Farmers should also be encouraged to adopt good agricultural practices to ensure food safety. Furthermore, regular monitoring of pesticide residues in different commodities may help to reduce dietary exposures in humans. This thesis (Chapter 3) examined the risk of insecticides due to tomato consumption. The current use of chlorpyrifos and triazophos in vegetables should be managed by enforcing strict regulations and an effective control mechanism.

6.3.3 Implications for occupational safety, including non-dietary exposure to pesticides

While working in fields treated with pesticides, farmers can be exposed to the residues present in soil. This also holds true for the people coming into contact with pesticides at

their workplace. Due to pesticides drift, populations in the immediate areas surrounding agricultural lands are also exposed via different exposure pathways such as inhalation, ingestion and dermal contact. As mentioned in Chapter 1, in developing countries including Nepal, farmers do not use proper safety measures while handling pesticides and can be exposed even when mixing and applying pesticides. Most Nepalese farmers ignored sanitation behaviours such as washing hands, wearing clothes and taking baths after spraying and they did not follow safe pesticide use practices (Chapter 2). Despite the poor safety practices of farmers during spraying and working in the fields, exposure to pesticides via the non-dietary pathways was negligible. Of all pathways, ingestion contributed higher risks, although the risks were not serious. As soils from conventional fields had higher concentrations of residues than IPM fields, current exposure due to ingestion of contaminated soil particles may further increase health risks in the future. Therefore, it is important to regularly monitor the pesticide residues in soil, especially soils from the conventional fields, to gauge the potential non-dietary exposure to pesticides.

6.3.4 Implications for the protection of soil communities in Nepal

The aim of sustainable agricultural development is to feed increasing populations without hampering health and the environment. Indiscriminate use of pesticides in intensive agriculture is a major threat to non-target species, including soil organisms (Mojsak et al., 2018; Schulz, 2004). Pesticides adversely affect soil health, including microbes and their diversity, composition and biological processes (Meena et al., 2020). In Chapter 5, we assessed the ecological risk of pesticides detected in soils. Of all the pesticides studied, organophosphates such as profenofos and chlorpyrifos as well as neonicotinoids such as imidacloprid showed higher ecological risks. Therefore, pesticides belonging to these groups should be accounted for in future ecological risk assessments. Pesticide risk at sites increased with a decrease in farmers' knowledge of the effects of pesticides and their personal safety behaviour. Farmer awareness of the ecological effects of pesticides and the effective safety management of their packets and containers may help to reduce the current ecological risk. Furthermore, higher tiered future risk assessment studies for different ecosystems, focusing especially on organophosphates and neonicotinoids, are needed to safeguard biota and maintain ecosystem services.

6.4 Research challenges and future research directions

We conducted surveys with farmers and retailers focused on their knowledge, attitude and behaviour. This thesis adds value to the understanding of the safety behaviour and pesticide

use of farmers and retailers. This was an attempt to identify factors affecting pesticide safety behaviour, assessing human health and environmental risk. We encountered a few challenges during the research period. Data on Maximum Residue Limits (MRLs), Acute Reference Dose (ARfD), Acceptable Daily Intake (ADI), and No Observed Effect Concentration (NOEC) for some pesticides detected in this study, including their degradation products, were not available. Furthermore, ADI and ARfD were based on No-Observed-Adverse-Effect level (NOAEL) that involved a default uncertainty factor (Alexander et al., 2008).

Future studies could be directed towards:

- 1. Upscaling a survey by including pesticide importers, formulators, and professional sprayers.
- 2. For a comprehensive dietary risk analysis, the number of samples and active ingredients should be increased, including imported foodstuffs. A 24-hour dietary recall method can be adopted during a survey for a more precise intake and risk estimation. A gender- based dietary risk study could be another option.
- 3. Continuous monitoring of pesticide residues in topsoil and including more pesticides that are used in vegetable farming during different seasons. Researchers should prioritize assessing the health and environmental risks posed by persistent and toxic OP insecticides that are currently applied.
- 4. Higher tiered risk assessment studies focusing on ecological risk to organisms in different ecosystems should conducted. Acute risk based on LD (LC)₅₀ might be conducted. The cocktail effect of pesticides should be assessed deploying a holistic risk assessment approach.

Literature cited

- Aamir, M., Khan, S., Li, G., 2018. Dietary exposure to HCH and DDT congeners and their associated cancer risk based on Pakistani food consumption. Environmental Science and Pollution Research 25, 8465-8474.
- Abdelbagi, A.O., Ismail, R.E.A., Ahmed Ishag, A.E.S., Ali Hammad, A.M., 2020. Pesticide Residues in Eggplant Fruit from Khartoum State, Sudan. Journal of Health & Pollution 10, 1-11.
- Abdollahzadeh, G., Sharifzadeh, M.S., Damalas, C.A., 2015. Perceptions of the beneficial and harmful effects of pesticides among Iranian rice farmers influence the adoption of biological control. Crop Protection 75, 124-131.
- Abrol, D.P., Shankar, U., 2014. Pesticides, Food Safety and Integrated Pest Management, Integrated Pest Management. Springer, NewYork, pp. 167-199.
- Adhikari, P.R., 2017. An overview of pesticide management in Nepal. The Journal of Agriculture and Environment 18, 95-105.
- Ahuja, D.B., Ahuja, U.R., Singh, S.K., Singh, N., 2015. Comparison of Integrated Pest Management approaches and conventional (non-IPM) practices in late-winter-season cauliflower in Northern India. Crop Protection 78, 232-238.
- AICC, 2018. Agriculture Diary (in Nepali). Agriculture Information and Communication Center. Ministry of Agricultural Development, Hariharbhawan, Lalitpur.
- Aislabie, J.M., Richards, N.K., Boul, H.L., 2010. Microbial degradation of DDT and its residues—A review. New Zealand Journal of Agricultural Research 40, 269-282.
- Akomea-Frempong, S., Ofosu, I.W., Owusu-Ansah, E.J., Darko, G., 2017. Health risks due to consumption of pesticides in ready-to-eat vegetables (salads) in Kumasi, Ghana. International Journal of Food Contamination 4, 11.
- Akoto, O., Andoh, H., Darko, G., Eshun, K., Osei-Fosu, P., 2013. Health risk assessment of pesticides residue in maize and cowpea from Ejura, Ghana. Chemosphere 92, 67-73.
- Akoto, O., Gavor, S., Appah, M.K., Apau, J., 2015. Estimation of human health risk associated with the consumption of pesticide-contaminated vegetables from Kumasi, Ghana. Environmental Monitoring and Assessment 187, 244.
- Aktar, M.W., Sengupta, D., Chowdhury, A., 2009. Impact of pesticides use in agriculture: their benefits and hazards. Interdisciplinary Toxicology 2, 1-12.
- Akter, M., Fan, L., Rahman, M.M., Geissen, V., Ritsema, C.J., 2018. Vegetable farmers' behaviour and knowledge related to pesticide use and related health problems: A case study from Bangladesh. Journal of Cleaner Production 200, 122-133.
- Al Zadjali, S., Morse, S., Chenoweth, J., Deadman, M., 2014. Factors determining pesticide use practices by farmers in the Sultanate of Oman. Science of the Total Environment 476/477, 505-512.
- Alamdar, A., Syed, J.H., Malik, R.N., Katsoyiannis, A., Liu, J., Li, J., Zhang, G., Jones, K.C., 2014. Organochlorine pesticides in surface soils from obsolete pesticide dumping ground in Hyderabad City, Pakistan: contamination levels and their potential for air-soil exchange. Science of the Total Environment 470/471, 733-741.
- Alexander, J., Hetland, R.B., Vikse, R., Dybing, E., Eriksen, G.S., Farstad, W., Jenssen, B.M.,

Paulsen, J.E., Skåre, J.U., Steffensen, I., Øvrebø, S., 2008. Combined toxic effects of multiple chemical exposures. Norwegian Scientific Committee for Food Safety, Oslo, p. 105.

- Ali, M.P., Kabir, M.M.M., Haque, S.S., Qin, X., Nasrin, S., Landis, D., Holmquist, B., Ahmed, N., 2020. Farmer's behavior in pesticide use: Insights study from smallholder and intensive agricultural farms in Bangladesh. Science of the Total Environment 747, 141160.
- Alves, P.R., Cardoso, E.J., Martines, A.M., Sousa, J.P., Pasini, A., 2014. Seed dressing pesticides on springtails in two ecotoxicological laboratory tests. Ecotoxicology and Environmental Safety 105, 65-71.
- Amirahmadi, M., Shoeibi, S., Abdollahi, M., Rastegar, H., Khosrokhavar, R., Hamedani, M.P., 2013. Monitoring of some pesticides residue in consumed tea in Tehran market. Journal of Environmental Health Sciences & Engineering 10, 1-6.
- Anastassiades, M., Lehotay, S.J., 2003. Fast and Easy Multiresidue Method Employing Acetonitrile Extraction/Partitioning and "Dispersive Solid-Phase Extraction" for the Determination of Pesticide Residues in Produce. Journal of AOAC International 86, 412-431.
- Archer, D.W., Franco, J.G., Halvorson, J.J., Pokharel, K.P., 2019. Integrated Farming Systems. Earth Systems and Environmental Sciences 4, 508-514.
- Arias, L.A., Bojacá, C.R., Ahumada, D.A., Schrevens, E., 2014. Monitoring of pesticide residues in tomato marketed in Bogota, Colombia. Food Control 35, 213-217.
- Arora, S., 2009. Analysis of insecticides in okra and brinjal from IPM and non-IPM fields. Environmental Monitoring and Assessment 151, 311-315.
- Aryal, K.K., Neupane, S., Lohani, G.R., Jors, E., Neupane, D., 2014. Health Effects of Pesticide among Vegetable Farmers and the Adaptation Level of Integrated Pest Management Program in Nepal p. 58.
- Atreya, K., 2007. Pesticide use knowledge and practices A gender differences in Nepal. Environmental Research 104, 305-311.
- Atreya, K., 2008a. Health costs from short-term exposure to pesticides in Nepal. Social Science and Medicine 67, 511-519.
- Atreya, K., 2008b. Probabilistic assessment of acute health symptoms related to pesticide use under intensified Nepalese agriculture. International Journal of Environmental Health Research 18, 187-208.
- Atreya, K., Johnsen, F.H., Sitaula, B.K., 2011. Health and environmental costs of pesticide use in vegetable farming in Nepal. Environment, Development and Sustainability 14, 477-493.
- Atreya, K., Sitaula, B.K., 2011. Mancozeb: growing risk for agricultural communities? Himalayan Journal of Sciences 6, 9-10.
- Atreya, K., Sitaula, B.K., Overgaard, H., Bajracharya, R.M., Sharma, S., 2012. Knowledge, attitude and practices of pesticide use and acetylcholinesterase depression among farm workers in Nepal. International Journal of Environmental Health Research 22, 401-415.
- Awasthi, K.R., Adefemi, K., Awasthi, M.S., Chalise, B., 2017. Public Health Interventions for Control of Malaria in the Population Living in the Terai Region of Nepal. Journal of Nepal Health Research Council 15, 202-207.
- Baker, B.P., Benbrook, C.M., Groth, E., 3rd, Lutz Benbrook, K., 2002. Pesticide residues in

conventional, integrated pest management (IPM)-grown and organic foods: insights from three US data sets. Food Additives and Contaminants 19, 427-446.

- Bakirci, G.T., Yaman Acay, D.B., Bakirci, F., Otles, S., 2014. Pesticide residues in fruits and vegetables from the Aegean region, Turkey. Food Chemistry 160, 379-392.
- Band, P.R., Abanto, Z., Bert, J., Lang, B., Fang, R., Gallagher, R.P., Le, N.D., 2011. Prostate cancer risk and exposure to pesticides in British Columbia farmers. Prostate 71, 168-183.
- Baskaran, S., Kookana, R.S., Naidu, R., 2003. Contrasting behaviour of chlorpyrifos and its primary metabolite, TCP (3,5,6-trichloro-2-pyridinol), with depth in soil profiles. Australian Journal of Soil Research 41, 749-760.
- Bay, A., Heshmati, H., 2016. Factors Associated with Pesticide Use Behaviors among Farmworkers Based On Health Belief Model. Iranian Journal of Public Health 42, 276-277.
- Bempah, C.K., Agyekum, A.A., Akuamoa, F., Frimpong, S., Buah-Kwofie, A., 2016. Dietary exposure to chlorinated pesticide residues in fruits and vegetables from Ghanaian markets. Journal of Food Composition and Analysis 46, 103-113.
- Bento, C.P.M., Yang, X., Gort, G., Xue, S., van Dam, R., Zomer, P., Mol, H.G.J., Ritsema, C.J., Geissen, V., 2016. Persistence of glyphosate and aminomethylphosphonic acid in loess soil under different combinations of temperature, soil moisture and light/darkness. Science of the Total Environment 572, 301-311.
- Bhandari, G., 2014. An Overview of Agrochemicals and Their Effects on Environment in Nepal. Applied Ecology and Environmental Sciences 2, 66-73.
- Bhandari, G., Atreya, K., Scheepers, P.T.J., Geissen, V., 2020. Concentration and distribution of pesticide residues in soil: Non-dietary human health risk assessment. Chemosphere 253, 126594.
- Bhandari, G., Atreya, K., Yang, X., Fan, L., Geissen, V., 2018. Factors affecting pesticide safety behaviour: The perceptions of Nepalese farmers and retailers. Science of the Total Environment 631/632, 1560-1571.
- Bhandari, G., Zomer, P., Atreya, K., Mol, H.G.J., Yang, X., Geissen, V., 2019. Pesticide residues in Nepalese vegetables and potential health risks. Environmental Research 172, 511-521.
- Bhandari, P.S., 2012. Farmers Perception about gains from Integrated Pest Management Farmer Field School. International Journal of Agricultural Science, Research and Technology 2, 137-142.
- Bhanti, M., Taneja, A., 2007. Contamination of vegetables of different seasons with organophosphorous pesticides and related health risk assessment in northern India. Chemosphere 69, 63-68.
- Bidleman, T.F., Leone, A.D., 2004. Soil–air exchange of organochlorine pesticides in the Southern United States. Environmental Pollution 128, 49-57.
- Biswas, B., Qi, F., Biswas, J., Wijayawardena, A., Khan, M., Naidu, R., 2018. The Fate of Chemical Pollutants with Soil Properties and Processes in the Climate Change Paradigm—A Review. Soil Systems 2, 1-20.
- Boul, H.L., 2010. DDT residues in the environment—A review with a New Zealand perspective. New Zealand Journal of Agricultural Research 38, 257-277.
- Brain, R.A., Anderson, J.C., 2019. The agro-enabled urban revolution, pesticides, politics, and

popular culture: a case study of land use, birds, and insecticides in the USA. Environmental Science and Pollution Research 26, 21717-21735.

- Briz, V., Molina-Molina, J.M., Sanchez-Redondo, S., Fernandez, M.F., Grimalt, J.O., Olea, N., Rodriguez-Farre, E., Sunol, C., 2011. Differential estrogenic effects of the persistent organochlorine pesticides dieldrin, endosulfan, and lindane in primary neuronal cultures. Toxicological Sciences 120, 413-427.
- Brown, K.A., Willis, P.G., Prussia, G.E., 2000. Predicting safe employee behavior in the steel industry: Development and test of a sociotechnical model. Journal of Operations Management 18, 445-465.
- Buglar, M.E., White, K.M., Robinson, N.G., 2010. The role of self-efficacy in dental patients' brushing and flossing: testing an extended Health Belief Model. Patient Education and Counseling 78, 269-272.
- Bundschuh, M., Goedkoop, W., Kreuger, J., 2014. Evaluation of pesticide monitoring strategies in agricultural streams based on the toxic-unit concept--experiences from long-term measurements. Science of the Total Environment 484, 84-91.
- Bürger, J., de Mol, F., Gerowitt, B., 2008. The "necessary extent" of pesticide use—Thoughts about a key term in German pesticide policy. Crop Protection 27, 343-351.
- Burrows, L.A., Edwards, C.A., 2002. The use of integrated soil microcosms to predict effects of pesticides on soil ecosystems. European Journal of Soil Biology 38, 245–249.
- Cabrera, N.L., Leckie, J., 2009. Pesticide Risk Communication, Risk Perception, and Self-Protective Behaviors Among Farmworkers in California's Salinas Valley. Hispanic Journal of Behavioral Sciences 3, 258-272.
- Cai, D.W., 2008. Understand the role of chemical pesticides and prevent misuses of pesticides. Bulletin of Agricultural Science and Technology, pp. 36-38.
- Cang, T., Dai, D., Yang, G., Yu, Y., Lv, L., Cai, L., Wang, Q., Wang, Y., 2017. Combined toxicity of imidacloprid and three insecticides to the earthworm, Eisenia fetida (Annelida, Oligochaeta). Environmental Science and Pollution Research 24, 8722-8730.
- Carniel, L.S.C., 2019. Novel approaches and future directions for pesticide ecological risk assessment to in-soil fauna, Department of Soils and Natural Resources. Santa Catarina State University, Brazil, p. 197.
- Carriquiriborde, P., Mirabella, P., Waichman, A., Solomon, K., Van den Brink, P.J., Maund, S., 2014. Aquatic risk assessment of pesticides in Latin America. Integrated Environmental Assessment and Management 10, 539-542.
- Carvalho, F.P., 2017. Pesticides, environment, and food safety. Food and Energy Security 6, 48-60.
- Casabé, N., Piola, L., Fuchs, J., Oneto, M.L., Pamparato, L., Basack, S., Giménez, R., Massaro, R., Papa, J.C., Kesten, E., 2007. Ecotoxicological assessment of the effects of glyphosate and chlorpyrifos in an Argentine soya field. Journal of Soils and Sediments 7, 232-239.
- Castaneda, A.R., Bhuiyan, S.I., 1996. Groundwater contamination by ricefield pesticides and some influencing factors. Journal of Environmental Science and Health. Part A: Environmental Science and Engineering and Toxicology 31, 83-99.
- CBS, 2012. Rupandehi Population Census 2011. Central Bureau of Statistics, Thapathali, Kathmandu, pp. 1-38.
- CBS, 2013. Environment Statistics of Nepal 2013. Central Bureau of Statistics, Ramshah Path, Thapathali, Kathmandu, p. 205.

- CBS, 2015. Compendium of Environment Statistics. Government of Nepal, National Planning Commission Secretariat, Central Bureau of Statistics, pp. 47-68.
- CBS, 2019. Environment Statistics of Nepal Central Bureau of Statistics, Kathmandu, Nepal, p. 239.
- Chabukdhara, M., Nema, A.K., 2013. Heavy metals assessment in urban soil around industrial clusters in Ghaziabad, India: probabilistic health risk approach. Ecotoxicology and Environmental Safety 87, 57-64.
- Chagnon, M., Kreutzweiser, D., Mitchell, E.A., Morrissey, C.A., Noome, D.A., Van der Sluijs, J.P., 2015. Risks of large-scale use of systemic insecticides to ecosystem functioning and services. Environmental Science and Pollution Research 22, 119-134.
- Chakra Reddy, N., Venkateswara Rao, J., 2008. Biological response of earthworm, Eisenia foetida (Savigny) to an organophosphorous pesticide, profenofos. Ecotoxicology and Environmental Safety 71, 574-582.
- Champion, V.L., Skinner, C.S., 2008. The Health Belief Model. Health behavior and health education: theory, research and practice, 45-62.
- Chattopadhyay, S., Chattopadhyay, D., 2015. Remediation of DDT and Its Metabolites in Contaminated Sediment. Current Pollution Reports 1, 248-264.
- Chen, C., Qian, Y., Chen, Q., Tao, C., Li, C., Li, Y., 2011. Evaluation of pesticide residues in fruits and vegetables from Xiamen, China. Food Control 22, 1114-1120.
- Chen, C., Qian, Y., Liu, X., Tao, C., Liang, Y., Li, Y., 2012. Risk assessment of chlorpyrifos on rice and cabbage in China. Regulatory Toxicology and Pharmacology 62, 125-130.
- Chen, C., Wang, Y., Zhao, X., Wang, Q., Qian, Y., 2014. Comparative and combined acute toxicity of butachlor, imidacloprid and chlorpyrifos on earthworm, Eisenia fetida. Chemosphere 100, 111-115.
- Chen, C., Zou, W., Chen, S., Zhang, K., Ma, L., 2020a. Ecological and health risk assessment of organochlorine pesticides in an urbanized river network of Shanghai, China. Environmental Sciences Europe 32.
- Chen, C., Zou, W., Cui, G., Tian, J., Wang, Y., Ma, L., 2020b. Ecological risk assessment of current-use pesticides in an aquatic system of Shanghai, China. Chemosphere, 127222.
- Chen, C.S., 2005. Ecological risk assessment for aquatic species exposed to contaminants in Keelung River, Taiwan. Chemosphere 61, 1142-1158.
- Chen, H., Hao, Z., Wang, Q., Jiang, Y., Pan, R., Wang, C., Liu, X., Lu, C., 2015. Occurrence and risk assessment of organophosphorus pesticide residues in Chinese tea. Human and Ecological Risk Assessment: An International Journal 22, 28-38.
- Chhetri, B., Chhetri, G.K., Palikhe, B.R., Shrestha, K.P., 2014. Study on National Pesticide Consumption Statistics in Nepal, pp. 1-89.
- Clever, J., 2017. China's National Food Safety Standard for Maximum Residue Limits for Pesticides in Foods, p. 20.
- Contreras López, M.C., 2003. Determination of potentially bioaccumulating complex mixtures of organochlorine compounds in wastewater: a review. Environment International 28, 751-759.
- Coppens, M., 2016. Understanding limited glove use among pesticide applicators: a qualitative study on Java island, Indonesia. Wageningen University, p. 72.
- Coronado, G.D., Holte, S., Vigoren, E., Griffith, W.C., Barr, D.B., Faustman, E., Thompson, B., 2011. Organophosphate pesticide exposure and residential proximity to nearby fields:

evidence for the drift pathway. Journal of Occupational and Environmental Medicine 53, 884-891.

- Cristache, S.-E., Vuţă, M., Marin, E., Cioacă, S.-I., Vuţă, M., 2018. Organic versus Conventional Farming—A Paradigm for the Sustainable Development of the European Countries. Sustainability 10, 4279.
- DADO, 2015. Annual Agriculture Development Programs and Statistical Report. District Agricultural Development Office, Rupandehi, Nepal, pp. 1-177.
- Dahal, B.M., Sitaula, B.K., Bajracharya, R.M., 2008. Sustainable Agricultural Intensification for Livelihood and Food Security in Nepal. Asian Journal of Water, Environment and Pollution 5, 1-12.
- Damalas, C., Koutroubas, S., Abdollahzadeh, G., 2019. Drivers of Personal Safety in Agriculture: A Case Study with Pesticide Operators. Agriculture 9, 34.
- Damalas, C.A., Abdollahzadeh, G., 2016. Farmers' use of personal protective equipment during handling of plant protection products: Determinants of implementation. Science of the Total Environment 571, 730-736.
- Damalas, C.A., Georgiou, E.B., Theodorou, M.G., 2006. Pesticide use and safety practices among Greek tobacco farmers: a survey. International Journal of Environmental Health Research 16, 339-348.
- Damalas, C.A., Hashemi, M.S., 2010. Pesticide risk perception and use of personal protective equipment among young and old cotton growers in Northern Greece. Agrociencia 44, 363-371.
- Damalas, C.A., Khan, M., 2017. Pesticide use in vegetable crops in Pakistan: Insights through an ordered probit model. Crop Protection 99, 59-64.
- Damalas, C.A., Koutroubas, S.D., 2016. Farmers' Exposure to Pesticides: Toxicity Types and Ways of Prevention. Toxics 4.
- Damalas, C.A., Koutroubas, S.D., 2017. Farmers' Training on Pesticide Use Is Associated with Elevated Safety Behavior. Toxics 5, 1-10.
- Damodaran, T.V., 2019. Peripheral Nervous System Toxicity Biomarkers, Biomarkers in Toxicology. Academic Press, USA, pp. 187-207.
- Darko, G., Akoto, O., 2008. Dietary intake of organophosphorus pesticide residues through vegetables from Kumasi, Ghana. Food and Chemical Toxicology 46, 3703-3706.
- Dasgupta, S., Meisner, C., Wheeler, D., Xuyen, K., Thi Lam, N., 2007. Pesticide poisoning of farm workers-implications of blood test results from Vietnam. International Journal of Hygiene and Environmental Health 210, 121-132.
- de Lima, E.S.C., Brennan, N., Brouwer, J.M., Commandeur, D., Verweij, R.A., van Gestel, C.A.M., 2017. Comparative toxicity of imidacloprid and thiacloprid to different species of soil invertebrates. Ecotoxicology 26, 555-564.
- De Silva, P.M.C.S., Pathiratne, A., van Gestel, C.A.M., 2010. Toxicity of chlorpyrifos, carbofuran, mancozeb and their formulations to the tropical earthworm Perionyx excavatus. Applied Soil Ecology 44, 56-60.
- Delcour, I., Spanoghe, P., Uyttendaele, M., 2015. Literature review: Impact of climate change on pesticide use. Food Research International 68, 7-15.
- Devi, N.L., Qi, S., Chakraborty, P., Zhang, G., Yadav, I.C., 2011. Passive air sampling of organochlorine pesticides in a northeastern state of India, Manipur. Journal of Environmental Sciences 23, 808-815.

- Devi, N.L., Yadav, I.C., Raha, P., Shihua, Q., Dan, Y., 2015. Spatial distribution, source apportionment and ecological risk assessment of residual organochlorine pesticides (OCPs) in the Himalayas. Environmental Science and Pollution Research 22, 20154-20166.
- DFTQC, 2012. Food composition table for Nepal. Department of Food Technology and Quality Control, Ministry of Agriculture Development, Nepal Government, Kathmandu, Nepal
- Dhakal, D., 2016. Study of Pesticide Residues Status in Strawberries of Nuwakot District, Nepal. Intellectual Integrity & Research Journal 1, 128-141.
- Dhimal, M., Ahrens, B., Kuch, U., 2014. Malaria control in Nepal 1963-2012: challenges on the path towards elimination. Malaria Journal 13, 1-14.
- Dinh Van, K., Janssens, L., Debecker, S., Stoks, R., 2014. Warming increases chlorpyrifos effects on predator but not anti-predator behaviours. Aquatic Toxicology 152, 215-221.
- Diop, A., Diop, Y.M., Thiare, D.D., Cazier, F., Sarr, S.O., Kasprowiak, A., Landy, D., Delattre, F., 2016. Monitoring survey of the use patterns and pesticide residues on vegetables in the Niayes zone, Senegal. Chemosphere 144, 1715-1721.
- Dittbrenner, N., Capowiez, Y., Köhler, H.-R., Triebskorn, R., 2012. Stress protein response (Hsp70) and avoidance behaviour in Eisenia fetida, Aporrectodea caliginosa and Lumbricus terrestris when exposed to imidacloprid. Journal of Soils and Sediments 12, 198-206.
- Diwakar, J., Prasai, T., Pant, S.R., Jayana, B.L., 2008. Study on major pesticides and fertilizers used in Nepal. Scientific World 6, 76-80.
- EC, 2002. Guidance Document on Terrestrial Ecotoxicology Under Council Directive 91/414/EEC. SANCO/10329/2002 rev 2 final, p. 39.
- EC, 2003. Technical Guidance Document on Risk Assessment. Institute for Health and Consumer Protection, European Commission, Luxembourg, p. 337.
- EC, 2004. Draft Assessment Report (DAR) for Dimethomorph, Rapporteur Member State (Germany) as in Article 8(2) of Council Directive 91/414/EEC.
- EC, 2009. Regulation (EC) No 1107/2009 of the European Parliament and of the Council of 21 October 2009 Concerning the Placing of Plant Protection Products on the Market and Repealing Council Directives 79/117/EEC and 91/414/EEC.
- EC, 2017. Guidance document on analytical quality control and method validation procedures for pesticide residues and analysis in food and feed p. 42.
- Ecobichon, D.J., 2001. Pesticide use in developing countries. Toxicology 160, 27-33.
- ECOFRAM, 1999. Ecological Committee on FIFRA Risk Assessment Methods (ECOFRAM), https://www.epa.gov/sites/production/files/2015-08/documents/terrreport.pdf, USA, p. 677.
- EFSA, 2005. European Food Safety Authority Report on Conclusion on the peer review of triclopyr, pp. 1-103.
- EFSA, 2010. Management of left-censored data in dietary exposure assessment of chemical substances. EFSA Journal 8, 96.
- EFSA, 2012. Conclusion on the peer review of the pesticide risk assessment of the active substance [emamectin]. EFSA Journal 10 (11): 2955, p. 89.
- EFSA, 2013a. Conclusion on the peer review of the pesticide risk assessment of the active substance chlorantraniliprole. EFSA Journal 11, 107.

- EFSA, 2013b. Scientific Opinion on the identification of pesticides to be included in cumulative assessment groups on the basis of their toxicological profile. EFSA Journal 11, 3293.
- EFSA, Brancato, A., Brocca, D., Ferreira, L., Greco, L., Jarrah, S., Leuschner, R., Medina, P., Miron, I., Nougadere, A., Pedersen, R., Reich, H., Santos, M., Stanek, A., Tarazona, J., Theobald, A., Villamar-Bouza, L., 2018. Use of EFSA Pesticide Residue Intake Model (EFSA PRIMo revision 3). EFSA Journal 16, 43.
- EFSA, Ockleford, C., Adriaanse, P., Berny, P., Brock, T., Duquesne, S., Grilli, S., Hernandez-Jerez, A.F., Bennekou, S.H., Klein, M., Kuhl, T., Laskowski, R., Machera, K., Pelkonen, O., Pieper, S., Stemmer, M., Sundh, I., Teodorovic, I., Tiktak, A., Topping, C.J., Wolterink, G., Craig, P., de Jong, F., Manachini, B., Sousa, P., Swarowsky, K., Auteri, D., Arena, M., Rob, S., 2017. Scientific Opinion addressing the state of the science on risk assessment of plant protection products for in-soil organisms. EFSA Journal 15, 225.
- Ellis, S.R., Hodson, M.E., Wege, P., 2007. The influence of different artificial soil types on the acute toxicity of carbendazim to the earthworm Eisenia fetida in laboratory toxicity tests. European Journal of Soil Biology 43, S239-S245.
- Elmore, R.C., Arcury, T.A., 2001. Pesticide Exposure Beliefs Among Latino Farmworkers in North Carolina's Christmas Tree Industry. American Journal of Industrial Medicine 40, 153-160.
- Ennaceur, S., Ridha, D., Marcos, R., 2008. Genotoxicity of the organochlorine pesticides 1,1dichloro-2,2- bis(p-chlorophenyl)ethylene (DDE) and hexachlorobenzene (HCB) in cultured human lymphocytes. Chemosphere 71, 1335-1339.
- Ennour-Idrissi, K., Ayotte, P., Diorio, C., 2019. Persistent Organic Pollutants and Breast Cancer: A Systematic Review and Critical Appraisal of the Literature. Cancers 11.
- Enserink, M., Hines, P.J., Vignieri, S.N., Wigginton, N.S., Yeston, J.S., 2013. The Pesticide Paradox. Science 341, 728-729.
- Ernst, F., Alonso, B., Colazzo, M., Pareja, L., Cesio, V., Pereira, A., Marquez, A., Errico, E., Segura, A.M., Heinzen, H., Perez-Parada, A., 2018. Occurrence of pesticide residues in fish from south American rainfed agroecosystems. Science of the Total Environment 631/632, 169-179.
- Esechie, J.O., Ibitayo, O.O., 2011. Pesticide use and related health problems among greenhouse workers in Batinah Coastal Region of Oman. Journal of Forensic and Legal Medicine 18, 198-203.
- Essumang, D.K., Dodoo, D.K., Adokoh, C.K., Fumador, E.A., 2008. Analysis of Some Pesticide Residues in Tomatoes in Ghana. Human and Ecological Risk Assessment: An International Journal 14, 796-806.
- EU, 2009. Pesticides: framework for Community action to achieve a sustainable use of pesticides, p. 16.
- EU, 2010. Commission regulation (EU) No 600/2010. Official Journal of the European Union 174, 18-39.
- EU, 2016. Regulation (EU) 2016/425 of the European Parliament and of the council of 9 March 2016 on personal protective equipment and repealing Council Directive 89/686/EEC. Official Journal of the European Union, 48.
- EU, 2017. Guidance document on analytical quality control and method validation procedures for pesticide residues and analysis in food and feed, pp. 1-46.

EU, 2018. European Union pesticides database 2018 http://ec.europa.eu/food/plant/pesticides/eu-pesticides-

```
database/public/?event=activesubstance.selection&language=EN.
```

- Evangelou, E., Ntritsos, G., Chondrogiorgi, M., Kavvoura, F.K., Hernandez, A.F., Ntzani, E.E., Tzoulaki, I., 2016. Exposure to pesticides and diabetes: A systematic review and metaanalysis. Environment International 91, 60-68.
- Fan, L., Liu, G., Wang, F., Geissen, V., Ritsema, C.J., 2013. Factors affecting domestic water consumption in rural households upon access to improved water supply: insights from the Wei River Basin, China. PloS One 8, 1-9.
- Fan, L., Niu, H., Yang, X., Qin, W., Bento, C.P., Ritsema, C.J., Geissen, V., 2015. Factors affecting farmers' behaviour in pesticide use: Insights from a field study in northern China. Science of the Total Environment 537, 360-368.
- Fang, L., Zhang, S., Chen, Z., Du, H., Zhu, Q., Dong, Z., Li, H., 2015. Risk assessment of pesticide residues in dietary intake of celery in China. Regulatory Toxicology and Pharmacology 73, 578-586.
- Fantke, P., Wieland, P., Wannaz, C., Friedrich, R., Jolliet, O., 2013. Dynamics of pesticide uptake into plants: From system functioning to parsimonious modeling. Environmental Modelling & Software 40, 316-324.
- FAO, 1995. Guidelines on Good Labelling Practice for Pesticides. Food and Agriculture Organization of the United Nations.
- FAO, 1999. Recommended methods of sampling for the determination of pesticide residues for compliance with MRLs, pp. 1-18.
- FAO, 2013. Food Security and Nutrition in the Southern and Eastern Rim of the Mediterranean Basin. Food and Agriculture Organization of the United Nations, Cairo.
- FAO, 2016. Submission and evaluation of pesticide residues data for the estimation of maximum residue levels in food and feed, 3 ed. Food and Agriculture Organization of the United Nations, Rome, p. 298.
- FAO, 2017a. Food and Agriculture Organization of the United Nations: FAOSTAT Inputs/Pesticides Use. http://www.fao.org/faostat/en/#data/RP.
- FAO, 2017b. Global assessment of the impact of plant protection products on soil functions and soil ecosystems. Food and Agriculture Organization of the United Nations, Rome, p. 40.
- FAO, 2020. Agricultural Pollution: Pesticides. http://documents.worldbank.org/curated/en/689281521218090562/pdf/124345-BRI-p153343-PUBLIC-march-22-9-pm-WB-Knowledge-Pesticides.pdf, p. 6.
- FAO/WHO, 2018. Codex Alimentarius 2018 http://www.fao.org/fao-whocodexalimentarius/codex-texts/dbs/pestres/pesticides/en/.
- FAOSTAT, 2019. The Pesticides Use database. http://www.fao.org/faostat/en/#data/RP.
- FICCI, 2015. Study on sub standard spurious counterfeit pesticides in India. Federation of Indian Chambers of Commerce and Industry India, pp. 1-94.
- Florence, C., Philippe, L., Magalie, L.J., 2015. Organochlorine (chlordecone) uptake by root vegetables. Chemosphere 118, 96-102.
- Fosu-Mensah, B.Y., Okoffo, E.D., Darko, G., Gordon, C., 2016. Organophosphorus pesticide residues in soils and drinking water sources from cocoa producing areas in Ghana. Environmental Systems Research 5, 1-12.

- Fountain, M., Hopkin, S., 2005. Folsomia Candida (Collembola): A "Standard" Soil Arthropod. Annual Review of Entomology, 201.
- Fournier, M.L., Echeverria-Saenz, S., Mena, F., Arias-Andres, M., de la Cruz, E., Ruepert, C., 2018. Risk assessment of agriculture impact on the Frio River watershed and Cano Negro Ramsar wetland, Costa Rica. Environmental Science and Pollution Research 25, 13347-13359.
- Fragoso, C., Brown, G.G., Patron, J.C., Blanchart, E., Lavelle, P., Pashanasi, B., Senapati, B., Kumar, T., 1997. Agricultural intensification, soil biodiversity and agroecosystem function in the tropics: the role of earthworms. Applied Soil Ecology 6, 17-35.
- Frampton, G.K., 2000. Recovery responses of soil surface Collembola after spatial and temporal changes in long-term regimes of pesticide use. Pedobiologia 44, 489-501.
- Frye, C.A., Bo, E., Calamandrei, G., Calza, L., Dessi-Fulgheri, F., Fernandez, M., Fusani, L., Kah, O., Kajta, M., Le Page, Y., Patisaul, H.B., Venerosi, A., Wojtowicz, A.K., Panzica, G.C., 2012. Endocrine disrupters: a review of some sources, effects, and mechanisms of actions on behaviour and neuroendocrine systems. Journal of Neuroendocrinology 24, 144-159.
- Furlong, M., Tanner, C.M., Goldman, S.M., Bhudhikanok, G.S., Blair, A., Chade, A., Comyns, K., Hoppin, J.A., Kasten, M., Korell, M., Langston, J.W., Marras, C., Meng, C., Richards, M., Ross, G.W., Umbach, D.M., Sandler, D.P., Kamel, F., 2015. Protective glove use and hygiene habits modify the associations of specific pesticides with Parkinson's disease. Environment International 75, 144-150.
- Gaber, S., Abdel-Latif, S.H., 2012. Effect of education and health locus of control on safe use of pesticides: a cross sectional random study. Journal of Occupational Medicine and Toxicology 7, 1-7.
- Gad Alla, S.A., Loutfy, N.M., Shendy, A.H., Ahmed, M.T., 2015. Hazard index, a tool for a long term risk assessment of pesticide residues in some commodities, a pilot study. Regulatory Toxicology and Pharmacology 73, 985-991.
- Gao, J., Zhou, H., Pan, G., Wang, J., Chen, B., 2013. Factors influencing the persistence of organochlorine pesticides in surface soil from the region around the Hongze Lake, China. Science of the Total Environment 443, 7-13.
- Gao, X., Li, J., Wang, X., Zhou, J., Fan, B., Li, W., Liu, Z., 2019. Exposure and ecological risk of phthalate esters in the Taihu Lake basin, China. Ecotoxicology and Environmental Safety 171, 564-570.
- Garrigou, A., Laurent, C., Berthet, A., Colosio, C., Jas, N., Daubas-Letourneux, V., Jackson Filho, J.M., Jouzel, J.N., Samuel, O., Baldi, I., Lebailly, P., Galey, L., Goutille, F., Judon, N., 2020. Critical review of the role of PPE in the prevention of risks related to agricultural pesticide use. Safety Science 123, 104527.
- Gautam, S., Schreinemachers, P., Uddin, M.N., Srinivasan, R., 2017. Impact of training vegetable farmers in Bangladesh in integrated pest management (IPM). Crop Protection 102, 161-169.
- GC, S., Neupane, J., 2019. Pesticides Use in Nepal and its Effects on Human Health A Review. Acta Scientific Agriculture 3, 114-117.
- Gesesew, H.A., Woldemichael, K., Massa, D., Mwanri, L., 2016. Farmers Knowledge, Attitudes, Practices and Health Problems Associated with Pesticide Use in Rural Irrigation Villages, Southwest Ethiopia. PloS One 11, e0162527.

- Ghanbari, F., Moattar, F., Monavari, S.M., Arjmandi, R., 2017. Human health risk assessment of organophosphorus pesticide in rice crop from selected districts of Anzali International Wetland basin, Iran. Human and Experimental Toxicology 36, 438-444.
- Giesy, J.P., Solomon, K.R., Cutler, G.C., Giddings, J.M., Mackay, D., Moore, D.R., Purdy, J., Williams, W.M., 2014. Ecological risk assessment of the uses of the organophosphorus insecticide chlorpyrifos, in the United States. Reviews of Environmental Contamination and Toxicology 231, 1-11.
- Giri, Y.P., Aryal, S., Paneru, R.B., 2012. Review and future strategies on the pesticide residue works of agricultural commodities in Nepal, Workshop on Review and Strategy Development of Entomological Research Works in Nepal. Nepal Agricultural Research Council, Entomology Division, Khumaltar, Lalitpur, Nepal, pp. 64-73.
- Goldman, L., Eskenazi, B., Bradman, A., Jewell, N.P., 2004. Risk behaviors for pesticide exposure among pregnant women living in farmworker households in Salinas, California. American Journal of Industrial Medicine 45, 491-499.
- Golge, O., Hepsag, F., Kabak, B., 2018. Health risk assessment of selected pesticide residues in green pepper and cucumber. Food and Chemical Toxicology 121, 51-64.
- Gong, Z.M., Tao, S., Xu, F.L., Dawson, R., Liu, W.X., Cui, Y.H., Cao, J., Wang, X.J., Shen, W.R., Zhang, W.J., Qing, B.P., Sun, R., 2004. Level and distribution of DDT in surface soils from Tianjin, China. Chemosphere 54, 1247-1253.
- GRM, 2018. Gaidahawa Rural Municipality (GRM) Village Profile 2018, Gaidahawa Rural Municipality p. 68.
- Guler, G.O., Cakmak, Y.S., Dagli, Z., Aktumsek, A., Ozparlak, H., 2010. Organochlorine pesticide residues in wheat from Konya region, Turkey. Food and Chemical Toxicology 48, 1218-1221.
- Gupta, P.K., 2004. Pesticide exposure—Indian scene. Toxicology 198, 83-90.
- Gupta, S., Gupta, K., 2020. Bioaccumulation of Pesticides and Its Impact on Biological Systems, Pesticides in Crop Production: Physiological and Biochemical Action. John Wiley & Sons Ltd., pp. 55-67.
- Gyenwali, D., Vaidya, A., Tiwari, S., Khatiwada, P., Lamsal, D.R., Giri, S., 2017. Pesticide poisoning in Chitwan, Nepal: a descriptive epidemiological study. BMC Public Health 17, 619.
- Haj-Younes, J., Huici, O., Jors, E., Moral, M., 2015. Sale, storage and use of legal, illegal and obsolete pesticides in Bolivia. Cogent Food & Agriculture 1, 1-12.
- Han, Y., Mo, R., Yuan, X., Zhong, D., Tang, F., Ye, C., Liu, Y., 2017. Pesticide residues in nutplanted soils of China and their relationship between nut/soil. Chemosphere 180, 42-47.
- Hanson, J.A., Benedict, J.A., 2002. Use of the Health Belief Model to Examine Older Adults' Food-Handling Behaviors. Journal of Nutrition Education and Behavior 34, S25-S30.
- Hardy, T., Bopp, S., Egsmose, M., Fontier, H., Mohimont, L., Steinkellner, H., Streissl, F., 2012. Risk assessment of plant protection products. EFSA Journal 10, 10.
- Haylamicheal, I.D., Dalvie, M.A., 2009. Disposal of obsolete pesticides, the case of Ethiopia. Environment International 35, 667-673.
- Herbert, I.N., Svendsen, C., Hankard, P.K., Spurgeon, D.J., 2004. Comparison of instantaneous rate of population increase and critical-effect estimates in Folsomia candida exposed to four toxicants. Ecotoxicology and Environmental Safety 57, 175-

183.

- Hernandez-Valero, M.A., Bondy, M.L., Spitz, M.R., Zahm, S.H., 2001. Evaluation of Mexican American Migrant Farmworker Work Practices and Organochlorine Pesticide Metabolites. American Journal of Industrial Medicine 40, 554-560.
- Hicks, B., 2019. Agricultural Pesticides and Human Health, Department of Earth Sciences, Montana State University, USA. https://serc.carleton.edu/NAGTWorkshops/health/case studies/pesticides.html.
- Hole, D.G., Perkins, A.J., Wilson, J.D., Alexander, I.H., Grice, P.V., Evans, A.D., 2005. Does organic farming benefit biodiversity? Biological Conservation 122, 113-130.
- Hossain, L., Rahman, R., Khan, M.S., 2017. Alternatives of Pesticides, in: Khan, M.S., Rahman, M.S. (Eds.), Pesticide Residue in Foods. Springer Cham, Switzerland pp. 147-165.
- Hou, X., Han, M., Dai, X., Yang, X., Yi, S., 2013. A multi-residue method for the determination of 124 pesticides in rice by modified QuEChERS extraction and gas chromatography-tandem mass spectrometry. Food Chemistry 138, 1198-1205.
- Houbraken, M., Bauweraerts, I., Fevery, D., Van Labeke, M.C., Spanoghe, P., 2016. Pesticide knowledge and practice among horticultural workers in the Lam Dong region, Vietnam: A case study of chrysanthemum and strawberries. Science of the Total Environment 550, 1001-1009.
- Houbraken, M., Habimana, V., Senaeve, D., Lopez-Davila, E., Spanoghe, P., 2017. Multiresidue determination and ecological risk assessment of pesticides in the lakes of Rwanda. Science of the Total Environment 576, 888-894.
- Hu, W., Huang, B., Zhao, Y., Sun, W., Gu, Z., 2014. Distribution, sources and potential risk of HCH and DDT in soils from a typical alluvial plain of the Yangtze River Delta region, China. Environmental Geochemistry and Health 36, 345-358.
- Huan, Z., Luo, J., Xu, Z., Xie, D., 2016. Acute Toxicity and Genotoxicity of Carbendazim, Main Impurities and Metabolite to Earthworms (Eisenia foetida). Bulletin of Environmental Contamination and Toxicology 96, 62-69.
- Huguier, P., Manier, N., Owojori, O.J., Bauda, P., Pandard, P., Rombke, J., 2015. The use of soil mites in ecotoxicology: a review. Ecotoxicology 24, 1-18.
- IARC, 2015. DDT, Lindane, and 2,4-D. IARC Monographs on the Evaluation of Carcinogenic Risks to Humans, Lyon, France, p. 513.
- Ibitayo, O.O., 2006. Egyptian farmers' attitudes and behaviors regarding agricultural pesticides: implications for pesticide risk communication. Risk Analysis 26, 989-995.
- ICAR, 2019. Pesticide risk assessment. National Research Centre for Integrated Pest Management, Pusa Campus, New Delhi, India.
- Iturburu, F.G., Calderon, G., Ame, M.V., Menone, M.L., 2019. Ecological Risk Assessment (ERA) of pesticides from freshwater ecosystems in the Pampas region of Argentina: Legacy and current use chemicals contribution. Science of the Total Environment 691, 476-482.
- Iwaniuk, A.N., Koperski, D.T., Cheng, K.M., Elliott, J.E., Smith, L.K., Wilson, L.K., Wylie, D.R., 2006. The effects of environmental exposure to DDT on the brain of a songbird: changes in structures associated with mating and song. Behavioural Brain Research 173, 1-10.
- Jaabiri Kamoun, I., Jegede, O.O., Owojori, O.J., Bouzid, J., Gargouri, R., Rombke, J., 2017. Effects of deltamethrin, dimethoate, and chlorpyrifos on survival and reproduction of

the collembolan Folsomia candida and the predatory mite Hypoaspis aculeifer in two African and two European soils. Integrated Environmental Assessment and Management 14, 92-104.

- Jager, T., Crommentuijn, T., van Gestel, C.A., Kooijman, S.A., 2007. Chronic exposure to chlorpyrifos reveals two modes of action in the springtail Folsomia candida. Environmental Pollution 145, 452-458.
- Jallow, M.F., Awadh, D.G., Albaho, M.S., Devi, V.Y., Thomas, B.M., 2017a. Pesticide risk behaviors and factors influencing pesticide use among farmers in Kuwait. Science of the Total Environment 574, 490-498.
- Jallow, M.F.A., Awadh, D.G., Albaho, M.S., Devi, V.Y., Ahmad, N., 2017b. Monitoring of Pesticide Residues in Commonly Used Fruits and Vegetables in Kuwait. International Journal of Environmental Research and Public Health 14, 1-12.
- Jänsch, S., Frampton, G.F., Römbke, J., Van den Brink, P.J., Scott-Fordsmand, J.J., 2006. Effects of pesticides on soil invertebrates in model ecosystem and field studies: a review and comparison with laboratory toxicity data. Environmental Toxicology and Chemistry 25, 2490–2501.
- Javed, M., Usmani, N., 2016. Accumulation of heavy metals and human health risk assessment via the consumption of freshwater fish Mastacembelus armatus inhabiting, thermal power plant effluent loaded canal. SpringerPlus 5, 1-8.
- Jegede, O.O., Owojori, O.J., Rombke, J., 2017. Temperature influences the toxicity of deltamethrin, chlorpyrifos and dimethoate to the predatory mite Hypoaspis aculeifer (Acari) and the springtail Folsomia candida (Collembola). Ecotoxicology and Environmental Safety 140, 214-221.
- Jensen, B.H., Petersen, A., Nielsen, E., Christensen, T., Poulsen, M.E., Andersen, J.H., 2015. Cumulative dietary exposure of the population of Denmark to pesticides. Food and Chemical Toxicology 83, 300-307.
- Jeong, H.R., Lim, S.J., Cho, J.Y., 2012. Monitoring and risk assessment of pesticides in fresh omija (Schizandra chinensis Baillon) fruit and juice. Food and Chemical Toxicology 50, 385-389.
- Jha, R.K., Regmi, A.P., 2009. Productivity of Pesticides in Vegetable Farming in Nepal, South Asian Network for Development and Environmental Economics (SANDEE) pp. 1-28.
- Jiang, Y.F., Wang, X.T., Jia, Y., Wang, F., Wu, M.H., Sheng, G.Y., Fu, J.M., 2009. Occurrence, distribution and possible sources of organochlorine pesticides in agricultural soil of Shanghai, China. Journal of Hazardous Materials 170, 989-997.
- Jin, J., Wang, W., He, R., Gong, H., 2017. Pesticide Use and Risk Perceptions among Small-Scale Farmers in Angiu County, China. International Journal of Environmental Research and Public Health 14, 1-10.
- Jongbloed, R.H., Traas, T.P., Luttik, R., 1996. A Probabilistic Model for Deriving Soil Quality Criteria Based on Secondary Poisoning of Top Predators. Ecotoxicology and Environmental Safety 34, 279-306.
- Jørs, E., Lander, F., Huici, O., Morant, R.C., Gulis, G., Konradsen, F., 2014. Do Bolivian small holder farmers improve and retain knowledge to reduce occupational pesticide poisonings after training on Integrated Pest Management? Environmental Health, 13:75.
- Joseph, L., Paulose, S.V., Cyril, N., Santhosh, S.K., Varghese, A., Nelson, A.B., Kunjankutty, S.V.,

Kasu, S., 2020. Organochlorine pesticides in the soils of Cardamom Hill Reserve (CHR), Kerala, India: Geo spatial distribution, ecological and human health risk assessment. Environmental Chemistry and Ecotoxicology 2, 1-11.

- Joshi, D., Rajbhandari, B.P., Bhattarai, B.P., Sah, L., 2017. Management practices adopted by commercial tomato growers against Tuta absoluta. Nepalese Journal of Agricultural Sciences 15, 93-97.
- Kafilzadeh, F., 2015. Assessment of Organochlorine Pesticide Residues in Water, Sediments and Fish from Lake Tashk, Iran. Achievements in the Life Sciences 9, 107-111.
- Kafle, L., GC, Y.D., Yang, J.T., Bhattarai, S., Tiwari, S., Katuwal, M., 2014. Integrated Pest Management in Nepal, The Fifth International Conference of Clinical Plant Science, pp. 113-124.
- Kapsi, M., Tsoutsi, C., Paschalidou, A., Albanis, T., 2019. Environmental monitoring and risk assessment of pesticide residues in surface waters of the Louros River (N.W. Greece). Science of the Total Environment 650, 2188-2198.
- Kariathi, V., Kassim, N., Kimanya, M., Yildiz, F., 2016. Pesticide exposure from fresh tomatoes and its relationship with pesticide application practices in Meru district. Cogent Food & Agriculture 2, 1-12.
- Karunamoorthi, K., Mohammed, M., Wassie, F., 2012. Knowledge and practices of farmers with reference to pesticide management: implications on human health. Archives of Environmental & Occupational Health 67, 109-116.
- Keesstra, S.D., Bouma, J., Wallinga, J., Tittonell, P., Smith, P., Cerdà, A., Montanarella, L., Quinton, J.N., Pachepsky, Y., van der Putten, W.H., Bardgett, R.D., Moolenaar, S., Mol, G., Jansen, B., Fresco, L.O., 2016. The significance of soils and soil science towards realization of the United Nations Sustainable Development Goals. Soil 2, 111-128.
- Keikotlhaile, B.M., Spanoghe, P., Steurbaut, W., 2010. Effects of food processing on pesticide residues in fruits and vegetables: a meta-analysis approach. Food and Chemical Toxicology 48, 1-6.
- Khan, M., 2010. Using the Health Belief Model to Understand Pesticide Use Decisions. The Pakistan Development Review, pp. 941-956.
- Khan, M., Damalas, C.A., 2015a. Factors preventing the adoption of alternatives to chemical pest control among Pakistani cotton farmers. International Journal of Pest Management 61, 9-16.
- Khan, M., Damalas, C.A., 2015b. Farmers' knowledge about common pests and pesticide safety in conventional cotton production in Pakistan. Crop Protection 77, 45-51.
- Khan, M., Husnain, M.I.U., Mahmood, H.Z., Akram, W., 2013. Understanding Pesticide Use Safety Decisions: Application of Health Behavior Theory. American-Eurasian Journal of Agricultural & Environmental Sciences 13, 440-448.
- Khanal, G., Singh, A., 2016. Patterns of Pesticide Use and Associated Factors Among the Commercial Farmers of Chitwan, Nepal. Environmental Health Insights 10, 1-7.
- Khandelwal, A., 2017. A study on the impact of pesticides and herbicides on the growth and reproduction of earthworm in Kota region, Department of Zoology. University of Kota, Rajasthan, India, p. 175.
- Kien, H.T., 2015. Factors influencing safety pesticide use behavior among farmers in Thai Nguyen province, Vietnam, Faculty of Nursing. Burapha University, Thailand p. 140.
- Kim, K.H., Kabir, E., Jahan, S.A., 2017. Exposure to pesticides and the associated human

health effects. Science of the Total Environment 575, 525-535.

- Klemick, H., Lichtenberg, E., 2008. Pesticide Use and Fish Harvests in Vietnamese Rice Agroecosystems. American Journal of Agricultural Economics 90, 1-14.
- Knežević, Z., Serdar, M., Ahel, M., 2012. Risk assessment of the intake of pesticides in Croatian diet. Food Control 23, 59-65.
- Koirala, P., Khadka, D.B., Mishra, A., 2007. Pesticide Residues as Environmetal Contaminants in Foods in Nepal. The Journal of Agriculture and Environment 8, 96-100.
- Kumari, B., Kumar, R., Madan, V.K., Singh, R., Singh, J., Kathpal, T.S., 2003. Magnitude of pesticidal contamination in winter vegetables from Hisar, Haryana. Environmental Monitoring and Assessment, 311-318.
- Kumari, P.L., Reddy, K.G., 2013. Knowledge and Practices of safety use of Pesticides among Farm workers. Journal of Agriculture and Veterinary Science 6, 1-8.
- Kurai, T., 2017. Japan's Maximum Residue Levels Ministry of Health, Labor and Welfare p. 34.
- Lama, J.P., 2008. Standard setting on pesticide residues to ensure food safety. The Journal of Agriculture and Environment 9, 46-53.
- Lamichhane, R., Lama, N., Subedi, S., Singh, S.B., Sah, R.B., Yadav, B.K., 2019. Use of Pesticides and Health Risk among Farmers in Sunsari District, Nepal. Journal of the Nepal Health Research Council 17, 66-70.
- Landrigan, P.J., Goldman, L.R., 2011. Children's vulnerability to toxic chemicals: a challenge and opportunity to strengthen health and environmental policy. Health Aff (Millwood) 30, 842-850.
- Latif, Y., Sherazi, S.T., Bhanger, M.I., 2011. Assessment of pesticide residues in commonly used vegetables in Hyderabad, Pakistan. Ecotoxicology and Environmental Safety 74, 2299-2303.
- Lavelle, P., Decaëns, T., Aubert, M., Barot, S., Blouin, M., Bureau, F., Margerie, P., Mora, P., Rossi, J.P., 2006. Soil invertebrates and ecosystem services. European Journal of Soil Biology 42, S3-S15.
- Lavtizar, V., Berggren, K., Trebse, P., Kraak, M.H.S., Verweij, R.A., van Gestel, C.A.M., 2016. Comparative ecotoxicity of chlorantraniliprole to non-target soil invertebrates. Chemosphere 159, 473-479.
- Lehmann, E., Turrero, N., Kolia, M., Konate, Y., de Alencastro, L.F., 2017. Dietary risk assessment of pesticides from vegetables and drinking water in gardening areas in Burkina Faso. Science of the Total Environment 601/602, 1208-1216.
- Lekei, E.E., Ngowi, A.V., London, L., 2014a. Farmers' knowledge, practices and injuries associated with pesticide exposure in rural farming villages in Tanzania. BMC Public Health 14, 1-13.
- Lekei, E.E., Ngowi, A.V., London, L., 2014b. Pesticide retailers' knowledge and handling practices in selected towns of Tanzania. Environmental Health 13, 1-9.
- Levesque, D.L., Arif, A.A., Shen, J., 2012. Effectiveness of pesticide safety training and knowledge about pesticide exposure among Hispanic farmworkers. Journal of Occupational and Environmental Medicine 54, 1550-1556.
- Lewis, K., Tzilivakis, J., Green, A., Warner, D., 2006. Pesticide Properties DataBase (PPDB). University of Hertfordshire. https://sitem.herts.ac.uk/aeru/ppdb/en/atoz.htm, UK.
- Lewis, K.A., Tzilivakis, J., Warner, D.J., Green, A., 2016a. An international database for

pesticide risk assessments and management. Human and Ecological Risk Assessment: An International Journal 22, 1050-1064.

- Lewis, S.E., Silburn, D.M., Kookana, R.S., Shaw, M., 2016b. Pesticide Behavior, Fate, and Effects in the Tropics: An Overview of the Current State of Knowledge. Journal of Agricultural and Food Chemistry 64, 3917-3924.
- Li, L., Zhang, Y., Zheng, L., Lu, S., Yan, Z., Ling, J., 2018. Occurrence, distribution and ecological risk assessment of the herbicide simazine: A case study. Chemosphere 204, 442-449.
- Li, Z., 2018. The use of a disability-adjusted life-year (DALY) metric to measure human health damage resulting from pesticide maximum legal exposures. Science of the Total Environment 639, 438-456.
- Li, Z., Jennings, A., 2017. Worldwide Regulations of Standard Values of Pesticides for Human Health Risk Control: A Review. International Journal of Environmental Research and Public Health 14, 1-41.
- Lichtenberg, E., Zimmerman, R., 1999. Adverse Health Experiences, Environmental Attitudes, and Pesticide Usage Behavior of Farm Operators. Risk Analysis 19, 283-294.
- Liu, K., Wang, S., Luo, K., Liu, X., Yu, Y., 2013a. Amelioration of acidic soil increases the toxicity of the weak base carbendazim to the earthworm Eisenia fetida. Environmental Toxicology and Chemistry 32, 2870-2873.
- Liu, T., Wang, X., Chen, D., Li, Y., Wang, F., 2018. Growth, reproduction and biochemical toxicity of chlorantraniliprole in soil on earthworms (Eisenia fetida). Ecotoxicology and Environmental Safety 150, 18-25.
- Liu, X., Song, Q., Tang, Y., Li, W., Xu, J., Wu, J., Wang, F., Brookes, P.C., 2013b. Human health risk assessment of heavy metals in soil-vegetable system: A multi-medium analysis. 463/464, 530-540.
- Liu, Y., Li, S., Ni, Z., Qu, M., Zhong, D., Ye, C., Tang, F., 2016a. Pesticides in persimmons, jujubes and soil from China: Residue levels, risk assessment and relationship between fruits and soils. Science of the Total Environment 542, 620-628.
- Liu, Y., Shen, D., Li, S., Ni, Z., Ding, M., Ye, C., Tang, F., 2016b. Residue levels and risk assessment of pesticides in nuts of China. Chemosphere 144, 645-651.
- Liu, Y.R., Zheng, Y.M., He, J.Z., 2012. Toxicity of profenofos to the springtail, Folsomia candida, and ammonia-oxidizers in two agricultural soils. Ecotoxicology 21, 1126-1134.
- Lockie, S., Lyons, K., Lawrence, G., Grice, J., 2004. Choosing organics: a path analysis of factors underlying the selection of organic food among Australian consumers. Appetite 43, 135-146.
- Louis, L.M., 2019. An examination of organochlorine insecticide exposures and associated cancer risks among the Agricultural Health Study (AHS) farm spouses, School of Graduate Studies. Rutgers, The State University of New Jersey, USA, p. 146.
- Lozowicka, B., 2015. Health risk for children and adults consuming apples with pesticide residue. Science of the Total Environment 502, 184-198.
- Lu, M.X., Jiang, W.W., Wang, J.L., Jian, Q., Shen, Y., Liu, X.J., Yu, X.Y., 2014. Persistence and dissipation of chlorpyrifos in Brassica chinensis, lettuce, celery, asparagus lettuce, eggplant, and pepper in a greenhouse. PloS One 9, 1-8.
- Ma, J., Pan, L.B., Yang, X.Y., Liu, X.L., Tao, S.Y., Zhao, L., Qin, X.P., Sun, Z.J., Hou, H., Zhou, Y.Z., 2016. DDT, DDD, and DDE in soil of Xiangfen County, China: Residues, sources, spatial distribution, and health risks. Chemosphere 163, 578-583.

Mansour, S.A., 2004. Pesticide exposure--Egyptian scene. Toxicology 198, 91-115.

- Mariyono, J., 2008. Direct and indirect impacts of integrated pest management on pesticide use: a case of rice agriculture in Java, Indonesia. Pest Management Science 64, 1069-1073.
- Maumbe, B.M., Swinton, S.M., 2003. Hidden health costs of pesticide use in Zimbabwe's smallholder cotton growers. Social Science and Medicine 57, 1559-1571.
- Mebdoua, S., Lazali, M., Ounane, S.M., Tellah, S., Nabi, F., Ounane, G., 2017. Evaluation of pesticide residues in fruits and vegetables from Algeria. Food Additives & Contaminants: Part B 10, 91-98.
- Meena, R., Kumar, S., Datta, R., Lal, R., Vijayakumar, V., Brtnicky, M., Sharma, M., Yadav, G., Jhariya, M., Jangir, C., Pathan, S., Dokulilova, T., Pecina, V., Marfo, T., 2020. Impact of Agrochemicals on Soil Microbiota and Management: A Review. Land 9, 34.
- Meimaridou, A., Haasnoot, W., Shelver, W.L., Franek, M., Nielen, M.W., 2013. Multiplex immunoassay for persistent organic pollutants in tilapia: comparison of imaging- and flow cytometry-based platforms using spectrally encoded paramagnetic microspheres. Food Additives & Contaminants: Part A 30, 843-852.
- Mekonen, S., Argaw, R., Simanesew, A., Houbraken, M., Senaeve, D., Ambelu, A., Spanoghe,
 P., 2016. Pesticide residues in drinking water and associated risk to consumers in Ethiopia. Chemosphere 162, 252-260.
- Mengistie, B.T., Mol, A.P.J., Oosterveer, P., 2015. Pesticide use practices among smallholder vegetable farmers in Ethiopian Central Rift Valley. Environment, Development and Sustainability 19, 301-324.
- Mengistie, B.T., Mol, A.P.J., Oosterveer, P., 2016. Private Environmental Governance in the Ethiopian Pesticide Supply Chain: Importation, Distribution and Use. NJAS -Wageningen Journal of Life Sciences 76, 65-73.
- Mequanint, C., Getachew, B., Mindaye, Y., Amare, D.E., Guadu, T., Dagne, H., 2019. Practice towards pesticide handling, storage and its associated factors among farmers working in irrigations in Gondar town, Ethiopia, 2019. BMC Research Notes 12, 709.
- Mishra, K., Sharma, R.C., Kumar, S., 2012. Contamination levels and spatial distribution of organochlorine pesticides in soils from India. Ecotoxicology and Environmental Safety 76, 215-225.
- Mladenova, R., Shtereva, D., 2009. Pesticide residues in apples grown under a conventional and integrated pest management system. Food Additives and Contaminants 26, 854-858.
- MoAD, 2016. Strengthening pesticides management in agriculture to reduce risks to health and environment. TCP/NEP/3502, Kathmandu, p. 2.
- MoAD, 2017. Statistical information on Nepalese agriculture 2015/16. Agri Statistics Section, Ministry of Agricultural Development, Government of Nepal Singha Durbar, Kathmandu, p. 219.
- MoEP, 2018. Environmental Quality Standard for Soils (GB 15618-2018). Ministry of Environmental Protection, China.
- MoF, 2017. Economic Survey 2017/18, Singh Durbar, Kathmandu, p. 286.
- MoFE, 2018. Nepal's Sixth National Report to the Convention on Biological Diversity. Ministry of Forests and Environment, Kathmandu, p. 137.
- MoH, ERA, N., ICF, 2017. Demographic and Health Survey 2016. Ministry of Health,

Kathmandu, p. 636.

- Mohanty, M.K., Behera, B.K., Jena, S.K., Srikanth, S., Mogane, C., Samal, S., Behera, A.A., 2013. Knowledge attitude and practice of pesticide use among agricultural workers in Puducherry, South India. Journal of Forensic and Legal Medicine 20, 1028-1031.
- Moher, D., Liberati, A., Tetzlaff, J., Altman, D.G., 2009. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Medicine 6, e1000097.
- Mojsak, P., Lozowicka, B., Kaczynski, P., 2018. Estimating acute and chronic exposure of children and adults to chlorpyrifos in fruit and vegetables based on the new, lower toxicology data. Ecotoxicology and Environmental Safety 159, 182-189.
- Montuori, P., Aurino, S., Garzonio, F., Sarnacchiaro, P., Polichetti, S., Nardone, A., Triassi, M., 2016. Estimates of Tiber River organophosphate pesticide loads to the Tyrrhenian Sea and ecological risk. Science of the Total Environment 559, 218-231.
- MoPE, 2000. Implementation of UN Convention to combat desertification, National Report, Kathmandu, pp. 1-50.
- MPI, 2017. Pesticides in Fresh and Frozen Produce. Systems Audit, Assurance and Monitoring, Directorate of the Ministry for Primary Industries, New Zealand pp. 1-26.
- Mtashobya, L.A., 2017. Assessment of pesticide residues in vegetables from the Western Usambara and Uruguru Mountains in Tanzania. Environmental Monitoring and Assessment 189, 1-14.
- Muangphra, P., Tharapoom, K., Euawong, N., Namchote, S., Gooneratne, R., 2016. Chronic toxicity of commercial chlorpyrifos to earthworm Pheretima peguana. Environmental Toxicology 31, 1450-1459.
- Naidoo, S., London, L., Rother, H.A., Burdorf, A., Naidoo, R.N., Kromhout, H., 2010. Pesticide safety training and practices in women working in small-scale agriculture in South Africa. Occupational and Environmental Medicine 67, 823-828.
- Neupane, D., Jors, E., Brandt, L., 2014. Pesticide use, erythrocyte acetylcholinesterase level and self-reported acute intoxication symptoms among vegetable farmers in Nepal: a cross-sectional study. Environmental Health 13, 1-7.
- Ngowi, A.V., Mbise, T.J., Ijani, A.S., London, L., Ajayi, O.C., 2007. Smallholder vegetable farmers in Northern Tanzania: Pesticides use practices, perceptions, cost and health effects. Crop Protection 26, 1617-1624.
- Nicolopoulou-Stamati, P., Maipas, S., Kotampasi, C., Stamatis, P., Hens, L., 2016. Chemical Pesticides and Human Health: The Urgent Need for a New Concept in Agriculture. Frontiers in Public Health 4, 148.
- Nougadere, A., Sirot, V., Cravedi, J.P., Vasseur, P., Feidt, C., Fussell, R.J., Hu, R., Leblanc, J.C., Jean, J., Riviere, G., Sarda, X., Merlo, M., Hulin, M., 2020. Dietary exposure to pesticide residues and associated health risks in infants and young children Results of the French infant total diet study. Environment International 137, 105529.
- Nougadere, A., Sirot, V., Kadar, A., Fastier, A., Truchot, E., Vergnet, C., Hommet, F., Bayle, J., Gros, P., Leblanc, J.C., 2012. Total diet study on pesticide residues in France: levels in food as consumed and chronic dietary risk to consumers. Environment International 45, 135-150.
- Novais, S.C., Gomes, N.C., Soares, A.M., Amorim, M.J., 2014. Antioxidant and neurotoxicity markers in the model organism Enchytraeus albidus (Oligochaeta): mechanisms of

response to atrazine, dimethoate and carbendazim. Ecotoxicology 23, 1220-1233.

- Novais, S.C., Soares, A.M., Amorim, M.J., 2010. Can avoidance in Enchytraeus albidus be used as a screening parameter for pesticides testing? Chemosphere 79, 233-237.
- Oesterlund, A.H., Thomsen, J.F., Sekimpi, D.K., Maziina, J., Racheal, A., Jors, E., 2014. Pesticide knowledge, practice and attitude and how it affects the health of small-scale farmers in Uganda: a cross-sectional study. African Health Sciences 14, 420-433.
- Okoffo, E.D., Mensah, M., Fosu-Mensah, B.Y., 2016. Pesticides exposure and the use of personal protective equipment by cocoa farmers in Ghana. Environmental Systems Research 5, 1-15.
- Palis, F.G., Flor, R.J., Warburton, H., Hossain, M., 2006. Our farmers at risk: behaviour and belief system in pesticide safety. Journal of Public Health 28, 43-48.
- Palma, G., Sanchez, A., Olave, Y., Encina, F., Palma, R., Barra, R., 2004. Pesticide levels in surface waters in an agricultural-forestry basin in Southern Chile. Chemosphere 57, 763-770.
- Palma, P., Kock-Schulmeyer, M., Alvarenga, P., Ledo, L., Barbosa, I.R., Lopez de Alda, M., Barcelo, D., 2014. Risk assessment of pesticides detected in surface water of the Alqueva reservoir (Guadiana basin, southern of Portugal). Science of the Total Environment 488/489, 208-219.
- PAN, 2020. The Cocktail Effect https://www.pan-uk.org/the-cocktail-effect/. Pesticide Action Network UK, UK.
- Pan, L., Sun, J., Li, Z., Zhan, Y., Xu, S., Zhu, L., 2018. Organophosphate pesticide in agricultural soils from the Yangtze River Delta of China: concentration, distribution, and risk assessment. Environmental Science and Pollution Research 25, 4-11.
- Pant, K.P., Acharya, B.N., Sedai, D., Mishra, R.N., Chand, N., Kharel, G.P., 2014. Biosecurity Status of Food and Agriculture in Nepal. Food and Agriculture Organization of the United Nations Kathmandu, Nepal p. 124.
- Panuwet, P., Siriwong, W., Prapamontol, T., Ryan, P.B., Fiedler, N., Robson, M.G., Barr, D.B., 2012. Agricultural Pesticide Management in Thailand: Situation and Population Health Risk. Environmental Science & Policy 17, 72-81.
- Pettinger, C.B., 2000. Improving Occupational Safety & Health Interventions A Comparison of Safety Self-Efficacy & Safety Stages of Change, Department of Psychology. Virginia Polytechnic Institute & State University, pp. 1-168.
- Phung, D.T., Connell, D., Miller, G., Rutherford, S., Chu, C., 2012. Pesticide regulations and farm worker safety: the need to improve pesticide regulations in Vietnam. Bulletin of the World Health Organization 90, 468-473.
- Pimentel, D., 2009. Pesticides and Pest Control. In: Peshin R., Dhawan A.K. (eds) Integrated Pest Management: Innovation-Development Process. Springer, Dordrecht., pp. 83-87.
- Pivato, A., Lavagnolo, M.C., Manachini, B., Vanin, S., Raga, R., Beggio, G., 2017. Ecological risk assessment of agricultural soils for the definition of soil screening values: A comparison between substance-based and matrix-based approaches. Heliyon 3, e00284.
- Pokhrel, B., Gong, P., Wang, X., Chen, M., Wang, C., Gao, S., 2018. Distribution, sources, and air-soil exchange of OCPs, PCBs and PAHs in urban soils of Nepal. Chemosphere 200, 532-541.
- Polat, B., Tiryaki, O., 2019. Determination of some pesticide residues in conventional-grown

and IPM-grown tomato by using QuEChERS method. Journal of Environmental Science and Health, Part B 54, 112-117.

- Posthuma, L., Brown, C.D., de Zwart, D., Diamond, J., Dyer, S.D., Holmes, C.M., Marshall, S., Burton, G.A., Jr., 2018. Prospective mixture risk assessment and management prioritizations for river catchments with diverse land uses. Environmental Toxicology and Chemistry 37, 715-728.
- Poudel, K.K., Huang, Z., Neupane, P.R., Steel, R., Poudel, J.K., 2017. Hospital-Based Cancer Incidence in Nepal from 2010 to 2013. Nepal Journal of Epidemiology 7, 659-665.
- PPD, 2013. Annual Progress Report, Plant Protection Directorate. Ministry of Agriculture, Hariharbhawan, Lalitpur.
- PPD, 2014. Annual Progress Report. Plant Protection Directorate, Hariharbhawan, Lalitpur, Nepal.
- PPD, 2015. Annual Progress Report, Plant Protection Directorate. Ministry of Agriculture, GoN, Hariharbhawan, Lalitpur.
- PPDB, 2019. Pesticide Properties database, Agriculture & Environment Research Unit (AERU), University of Hertfordshire, UK. https://sitem.herts.ac.uk/aeru/ppdb/en/atoz.htm.
- PQPMC, 2019. Registered Pesticides List and Pesticide Consumption Statistics. Plant Quarantine and Pesticides Management Centre, Lalitpur, Nepal p. 167.
- Pretty, J., Bharucha, Z.P., 2015. Integrated Pest Management for Sustainable Intensification of Agriculture in Asia and Africa. Insects 6, 152-182.
- Prisco, G.D., Cavaliere, V., Annoscia, D., Varricchio, P., 2013. Neonicotinoid clothianidin adversely affects insect immunity and promotes replication of a viral pathogen in honey bees. Proceedings of the National Academy of Sciences of the United States of America 110, 18466-18471.
- PRMD, 2013. Pesticide Use Pattern Manual (in Nepali: Panjikrit Bisadiharuko Prayog Bidhi Pustika). Pesticide Registration and Management Division, Plant Protection Directorate, Department of Agriculture, Ministry of Agricultural Development, Government of Nepal pp. 1-75.
- PRMD, 2015. Pesticide Statistic Book. Pesticide Registration and Management Division, Plant Protection Directorate, Department of Agriculture, Ministry of Agricultural Development, Government of Nepal pp. 1-50.
- PRMD, 2016. Pesticide Use Pattern Manaul. Pesticide Registration and Management Division, Plant Protection Directorate, Department of Agriculture, Ministry of Agricultural Development, Government of Nepal pp. 1-60.
- Prutner, W., Nicken, P., Haunhorst, E., Hamscher, G., Steinberg, P., 2013. Effects of single pesticides and binary pesticide mixtures on estrone production in H295R cells. Archives of Toxicology 87, 2201-2214.
- Qiu, X.H., Zhu, T., Li, J., Pan, H.S., Li, Q.L., Miao, G.F., Gong, J.C., 2004. Organochlorine Pesticides in the Air around the Taihu Lake, China. Environmental Science & Technology 38, 1368-1374.
- Qiu, X.H., Zhu, T., Yao, B., Hu, J., Hu, S., 2005. Contribution of Dicofol to the Current DDT Pollution in China. Environmental Science & Technology 39, 4385-4390.
- Qu, C., Qi, S., Yang, D., Huang, H., Zhang, J., Chen, W., Yohannes, H.K., Sandy, E.H., Yang, J.,
 Xing, X., 2015. Risk assessment and influence factors of organochlorine pesticides (OCPs) in agricultural soils of the hill region: A case study from Ningde, southeast China.

Journal of Geochemical Exploration 149, 43-51.

- Quijano, L., Yusa, V., Font, G., Pardo, O., 2016. Chronic cumulative risk assessment of the exposure to organophosphorus, carbamate and pyrethroid and pyrethrin pesticides through fruit and vegetables consumption in the region of Valencia (Spain). Food and Chemical Toxicology 89, 39-46.
- Rafique, N., Tariq, S.R., Ahmed, D., 2016. Monitoring and distribution patterns of pesticide residues in soil from cotton/wheat fields of Pakistan. Environmental Monitoring and Assessment 188: 695, 1-12.
- Raheem, S., Al-Dossary, M., AL-Saad, H., 2017. Determination of Carbendazim Fungicide and Oxymatrine Insecticide Residues in the Soils of Four Agriculture Stations in Basrah Governorate by HPLC. Journal of Biology, Agriculture and Healthcare 7, 10-16.
- Raksanam, B., Suklim, N., Songthap, A., 2014a. Model development to reduce pesticide risk behaviors among rubber farmers in Khogyang Community, Trang, Thailand. European Journal of Research on Education 2, 101-108.
- Raksanam, B., Taneepanichskul, S., Robson, M.G., Siriwong, W., 2014b. Health risk behaviors associated with agrochemical exposure among rice farmers in a rural community, Thailand: a community-based ethnography. Asia-Pacific Journal of Public Health 26, 588-595.
- Raksanam, B., Taneepanichskul, S., Siriwong, W., Robson, M.G., 2012. Factors Associated with Pesticide Risk Behaviors among Rice Farmers in Rural Community, Thailand. Journal of Environment and Earth Science 2, 32-39.
- Ramadan, M.F.A., Abdel-Hamid, M.M.A., Altorgoman, M.M.F., AlGaramah, H.A., Alawi, M.A., Shati, A.A., Shweeta, H.A., Awwad, N.S., 2020. Evaluation of Pesticide Residues in Vegetables from the Asir Region, Saudi Arabia. Molecules 25.
- Ranga Rao, G.V., Sahrawat, K.L., Srinivasa Rao, C., Das, B., Reddy, K.K., Bharath, B.S., Rao, V.R., Murthy, K.V.S., Wani, S.P., 2009. Insecticide Residues in Vegetable crops Grown in Kothapalli Watershed, Andhra Pradesh, India : A case Study. Indian Journal of Dryland Agricultural Research and Development 24, 21-27.
- Rasmussen, P.E., Subramanian, K.S., Jessiman, B.J., 2001. A multi-element profile of housedust in relation to exterior dust and soils in the city of Ottawa, Canada. Science of the Total Environment 267, 125-140.
- Raut, N., Sitaula, B.K., Bajracharya, R.M., 2011. A Discourse on Agricultural Intensification in the Mid-Hills of Nepal. Asian Journal of Agriculture and Development 8, 79-90.
- Rawal, D.S., Rana, N., Shrestha, S., Sijapati, J., 2012. Assessment of Pesticide Residues in Some Vegetables Grown in Kavrepalanchok and Bhaktapur Districts. Nepal Journal of Science and Technology 13, 45-50.
- Reffstrup, T.K., Larsen, J.C., Meyer, O., 2010. Risk assessment of mixtures of pesticides. Current approaches and future strategies. Regulatory Toxicology and Pharmacology 56, 174-192.
- Reiler, E., Jors, E., Baelum, J., Huici, O., Alvarez Caero, M.M., Cedergreen, N., 2015. The influence of tomato processing on residues of organochlorine and organophosphate insecticides and their associated dietary risk. Science of the Total Environment 527/528, 262-269.
- Renaud, M., Akeju, T., Natal-da-Luz, T., Leston, S., Rosa, J., Ramos, F., Sousa, J.P., Azevedo-Pereira, H., 2018. Effects of the neonicotinoids acetamiprid and thiacloprid in their

commercial formulations on soil fauna. Chemosphere 194, 85-93.

- Rezaei, R., Damalas, C.A., Abdollahzadeh, G., 2018. Understanding farmers' safety behaviour towards pesticide exposure and other occupational risks: The case of Zanjan, Iran. Science of the Total Environment 616/617, 1190-1198.
- Rico, A., Sabater, C., Castillo, M.A., 2016. Lethal and sub-lethal effects of five pesticides used in rice farming on the earthworm Eisenia fetida. Ecotoxicology and Environmental Safety 127, 222-229.
- Rijal, J., Regmi, R., Ghimire, R., Puri, K., Gyawaly, S., Poudel, S., 2018. Farmers' Knowledge on Pesticide Safety and Pest Management Practices: A Case Study of Vegetable Growers in Chitwan, Nepal. Agriculture 8, 1-11.
- Ritz, B., Yu, F., 2000. Parkinson's disease mortality and pesticide exposure in California 1984– 1994. International Journal of Epidemiology 29, 323–329.
- RIVM, 2015. Evaluation of ecological risk limits for DDT and drins in soil. National Institute for Public Health and the Environment, The Netherlands p. 96.
- Roberts, J.R., Reigart, J.R., 2013. Recognition and Management of Pesticide Poisonings 6th edition. Environmental Protection Agency, US p. 277.
- Rockström, J., Williams, J., Daily, G., Noble, A., Matthews, N., Gordon, L., Wetterstrand, H., DeClerck, F., Shah, M., Steduto, P., de Fraiture, C., Hatibu, N., Unver, O., Bird, J., Sibanda, L., Smith, J., 2016. Sustainable intensification of agriculture for human prosperity and global sustainability. A Journal of the Human Environment 46, 4-17.
- Rodríguez-Eugenio, N., McLaughlin, M., Pennock, D., 2018. Soil pollution: A hidden reality. Food and Agriculture Organization of the United Nations, Rome, p. 142.
- Romeh, A.A., 2018. Integrated Pest Management for Sustainable Agriculture, Sustainability of Agricultural Environment in Egypt: Part II. Springer, Cham, Switzerland, pp. 215-234.
- Rose, D.C., Sutherland, W.J., Barnes, A.P., Borthwick, F., Ffoulkes, C., Hall, C., Moorby, J.M., Nicholas-Davies, P., Twining, S., Dicks, L.V., 2019. Integrated farm management for sustainable agriculture: Lessons for knowledge exchange and policy. Land Use Policy 81, 834-842.
- Rutgers, M., Tuinstra, J., Spijker, J., Mesman, M., Wintersen, A., Posthuma, L., 2008. Ecological risks of soil contamination in the second step of the remediation criterion. RIVM, Bilthoven, The Netherlands p. 77.
- Sabarwal, A., Kumar, K., Singh, R.P., 2018. Hazardous effects of chemical pesticides on human health-Cancer and other associated disorders. Environmental Toxicology and Pharmacology 63, 103-114.
- Salamzadeh, J., Shakoori, A., Moradi, V., 2018. Occurrence of multiclass pesticide residues in tomato samples collected from different markets of Iran. Journal of Environmental Health Science and Engineering 16, 55-63.
- Salvatore, A.L., Bradman, A., Castorina, R., Camacho, J., Lopez, J., Barr, D.B., Snyder, J., Jewell, N.P., Eskenazi, B., 2008. Occupational behaviors and farmworkers' pesticide exposure: findings from a study in Monterey County, California. American Journal of Industrial Medicine 51, 782-794.
- Samsel, A., Seneff, S., 2013. Glyphosate's Suppression of Cytochrome P450 Enzymes and Amino Acid Biosynthesis by the Gut Microbiome: Pathways to Modern Diseases. Entropy 15, 1416-1463.
- Sánchez-Bayo, F., Baskaran, S., Robert Kennedy, I., 2002. Ecological relative risk (EcoRR):

another approach for risk assessment of pesticides in agriculture. Agriculture, Ecosystems & Environment 91, 37-57.

- Sánchez-González, S., Pose-Juan, E., Herrero-Hernández, E., Álvarez-Martín, A., Sánchez-Martín, M.J., Rodríguez-Cruz, S., 2013. Pesticide residues in groundwaters and soils of agricultural areas in the Águeda River Basin from Spain and Portugal. International Journal of Environmental Analytical Chemistry 93, 1585-1601.
- Sanchez-Hernandez, J.C., Narvaez, C., Sabat, P., Martinez Mocillo, S., 2014. Integrated biomarker analysis of chlorpyrifos metabolism and toxicity in the earthworm Aporrectodea caliginosa. Science of the Total Environment 490, 445-455.
- Sanchez-Hernandez, J.C., Rios, J.M., Attademo, A.M., 2018. Response of digestive enzymes and esterases of ecotoxicological concern in earthworms exposed to chlorpyrifostreated soils. Ecotoxicology 27, 890-899.
- Sankararamakrishnan, N., Kumar Sharma, A., Sanghi, R., 2005. Organochlorine and organophosphorous pesticide residues in ground water and surface waters of Kanpur, Uttar Pradesh, India. Environment International 31, 113-120.
- Sapbamrer, R., Thammachai, A., 2020. Factors affecting use of personal protective equipment and pesticide safety practices: A systematic review. Environmental Research 185, 109444.
- Schneider Medeiros, M., S, P.R., M, P.S., Schumacher-Schuh, A.F., Mello Rieder, C.R., 2020. Occupational pesticide exposure and the risk of death in patients with Parkinson's disease: an observational study in southern Brazil. Environmental Health 19, 1-8.
- Schreinemachers, D., 2003. Birth Malformations and Other Adverse Perinatal Outcomes in Four U.S. Wheat-Producing States. Environmental Health Perspectives 111, 1259-1264.
- Schreinemachers, P., Chen, H.P., Nguyen, T.T.L., Buntong, B., Bouapao, L., Gautam, S., Le, N.T., Pinn, T., Vilaysone, P., Srinivasan, R., 2017. Too much to handle? Pesticide dependence of smallholder vegetable farmers in Southeast Asia. Science of the Total Environment 593-594, 470-477.
- Schreinemachers, P., Grovermann, C., Praneetvatakul, S., Heng, P., Nguyen, T.T.L., Buntong,
 B., Le, N.T., Pinn, T., 2020. How much is too much? Quantifying pesticide overuse in vegetable production in Southeast Asia. Journal of Cleaner Production 244, 118738.
- Schreinemachers, P., Schad, I., Tipraqsa, P., Williams, P.M., Neef, A., Riwthong, S., Sangchan, W., Grovermann, C., 2012. Can public GAP standards reduce agricultural pesticide use? The case of fruit and vegetable farming in northern Thailand. Agriculture and Human Values 29, 519-529.
- Schulz, R., 2004. Field Studies on Exposure, Effects, and Risk Mitigation of Aquatic Nonpoint-Source Insecticide Pollution: A Review. Journal of Environmental Quality 33, 419-448.
- Seenivasan, S., Muraleedharan, N., 2011. Survey on the pesticide residues in tea in south India. Environmental Monitoring and Assessment 176, 365-371.
- Seo, Y.H., Cho, T.H., Hong, C.K., Kim, M.S., Cho, S.J., Park, W.H., Hwang, I.S., Kim, M.S., 2013. Monitoring and risk assessment of pesticide residues in commercially dried vegetables. Preventive Nutrition and Food Science 18, 145-149.
- Shabeer, T.P.A., Banerjee, K., Jadhav, M., Girame, R., Utture, S., Hingmire, S., Oulkar, D., 2015. Residue dissipation and processing factor for dimethomorph, famoxadone and cymoxanil during raisin preparation. Food Chemistry 170, 180-185.
- Shah, Y., Katuwal, A., Pun, R., Pant, K., Sherchand, S.P., Pandey, K., Joshi, D.D., Pandey, B.D.,

2012. Dengue in Western Terai Region of Nepal. Journal of Nepal Health Research Council 10, 152-155.

- Shammi, M., Sultana, A., Hasan, N., Mostafizur Rahman, M., Saiful Islam, M., Bodrud-Doza, M., Khabir Uddin, M., 2020. Pesticide exposures towards health and environmental hazard in Bangladesh: A case study on farmers' perception. Journal of the Saudi Society of Agricultural Sciences 19, 161-173.
- Sharifzadeh, M.S., Abdollahzadeh, G., Damalas, C.A., Rezaei, R., Ahmadyousefi, M., 2019. Determinants of pesticide safety behavior among Iranian rice farmers. Science of the Total Environment 651, 2953-2960.
- Sharifzadeh, M.S., Damalas, C.A., Abdollahzadeh, G., 2017. Perceived usefulness of personal protective equipment in pesticide use predicts farmers' willingness to use it. Science of the Total Environment 609, 517-523.
- Sharma, A., Kumar, V., Shahzad, B., Tanveer, M., Sidhu, G.P.S., Handa, N., Kohli, S.K., Yadav, P., Bali, A.S., Parihar, R.D., Dar, O.I., Singh, K., Jasrotia, S., Bakshi, P., Ramakrishnan, M., Kumar, S., Bhardwaj, R., Thukral, A.K., 2019. Worldwide pesticide usage and its impacts on ecosystem. SN Applied Sciences 1, 16.
- Sharma, D.R., 2015. Use of pesticides and its residue on vegetable crops in Nepal. The Journal of Agriculture and Environment 16, 33-42.
- Sharma, D.R., Thapa, R.B., Manandhar, H.K., Shrestha, S.M., Pradhan, S.B., 2012. Use of pesticides in Nepal and impacts on human health and environment. The Journal of Agriculture and Environment 13, 67-74.
- Sharma, R., Peshin, R., 2016. Impact of integrated pest management of vegetables on pesticide use in subtropical Jammu, India. Crop Protection 84, 105-112.
- Shelton, J.F., Geraghty, E.M., Tancredi, D.J., Delwiche, L.D., Schmidt, R.J., Ritz, B., Hansen, R.L., Hertz-Picciotto, I., 2014. Neurodevelopmental disorders and prenatal residential proximity to agricultural pesticides: the CHARGE study. Environmental Health Perspectives 122, 1103-1109.
- Shi-ping, Z., Chang-qun, D., Hui, F., Yu-hui, C., Xue-hua, W., Ze-fen, Y., 2007. Toxicity assessment for chlorpyrifos-contaminated soil with three different earthworm test methods. Journal of Environmental Sciences 19, 854–858.
- Shi, Y., Zhang, Q., Huang, D., Zheng, X., Shi, Y., 2016. Survival, growth, detoxifying and antioxidative responses of earthworms (Eisenia fetida) exposed to soils with industrial DDT contamination. Pesticide Biochemistry and Physiology 128, 22-29.
- Shinomiya, N., Shinomiya, M., 2003. Dichlorodiphenyltrichloroethane suppresses neurite outgrowth and induces apoptosis in PC12 pheochromocytoma cells. Toxicology Letters 137, 175-183.
- Sieke, C., Michalski, B., Kuhl, T., 2018. Probabilistic dietary risk assessment of pesticide residues in foods for the German population based on food monitoring data from 2009 to 2014. Journal of Exposure Science and Environmental Epidemiology 28, 46-54.
- Sijtsma, K., 2009. On the Use, the Misuse, and the Very Limited Usefulness of Cronbach's Alpha. Psychometrika 74, 107-120.
- Sillapawattana, P., Schaffer, A., 2017. Effects of imidacloprid on detoxifying enzyme glutathione S-transferase on Folsomia candida (Collembola). Environmental Science and Pollution Research 24, 11111-11119.
- Silva, V., Mol, H.G.J., Zomer, P., Tienstra, M., Ritsema, C.J., Geissen, V., 2019. Pesticide

residues in European agricultural soils - A hidden reality unfolded. Science of the Total Environment 653, 1532-1545.

- Silva, V., Montanarella, L., Jones, A., Fernandez-Ugalde, O., Mol, H.G.J., Ritsema, C.J., Geissen, V., 2018. Distribution of glyphosate and aminomethylphosphonic acid (AMPA) in agricultural topsoils of the European Union. Science of the Total Environment 621, 1352-1359.
- Singh, S.B., Mukherjee, I., Maisnam, J., Kumar, P., Gopal, M., Kulshrestha, G., 2009. Determination of pesticide residues in integrated pest management and nonintegrated pest management samples of apple (Malus pumila Mill.). Journal of Agricultural and Food Chemistry 57, 11277-11283.
- Sinha, S.N., Rao, M.V.V., Vasudev, K., 2012. Distribution of pesticides in different commonly used vegetables from Hyderabad, India. Food Research International 45, 161-169.
- Snipes, S.A., Thompson, B., Connor, K.O., Shell-Duncan, B., 2009. "Pesticides Protect the Fruit, but Not the People": Using Community-Based Ethnography to Understand Farmworker Pesticide-Exposure Risks. American Journal of Public Health 99, S616-S618.
- Sosan, M.B., Akingbohungbe, A.E., Ojo, I.A., Durosinmi, M.A., 2008. Insecticide residues in the blood serum and domestic water source of cacao farmers in Southwestern Nigeria. Chemosphere 72, 781-784.
- Stadlinger, N., Mmochi, A.J., Dobo, S., Gyllbäck, E., Kumblad, L., 2011. Pesticide use among smallholder rice farmers in Tanzania. Environment, Development and Sustainability 13, 641-656.
- Stanley, J., Preetha, G., 2016. Pesticide Toxicity to Microorganisms: Exposure, Toxicity and Risk Assessment Methodologies. 351-410.
- Strong, L.L., Thompson, B., Koepsell, T.D., Meischke, H., 2008. Factors associated with pesticide safety practices in farmworkers. American Journal of Industrial Medicine 51, 69-81.
- Sun, H., Qi, Y., Zhang, D., Li, Q.X., Wang, J., 2016. Concentrations, distribution, sources and risk assessment of organohalogenated contaminants in soils from Kenya, Eastern Africa. Environmental Pollution 209, 177-185.
- Sun, Z., Chen, J., 2018. Risk Assessment of Potentially Toxic Elements (PTEs) Pollution at a Rural Industrial Wasteland in an Abandoned Metallurgy Factory in North China. International Journal of Environmental Research and Public Health 15, 1-16.
- Swarnam, T.P., Velmurugan, A., 2013. Pesticide residues in vegetable samples from the Andaman Islands, India. Environmental Monitoring and Assessment 185, 6119-6127.
- Sweetman, A.J., Valle, M.D., Prevedouros, K., Jones, K.C., 2005. The role of soil organic carbon in the global cycling of persistent organic pollutants (POPs): interpreting and modelling field data. Chemosphere 60, 959-972.
- Syed, J.H., Alamdar, A., Mohammad, A., Ahad, K., Shabir, Z., Ahmed, H., Ali, S.M., Sani, S.G., Bokhari, H., Gallagher, K.D., Ahmad, I., Eqani, S.A., 2014. Pesticide residues in fruits and vegetables from Pakistan: a review of the occurrence and associated human health risks. Environmental Science and Pollution Research 21, 13367-13393.
- Tadevosyan, A., Tadevosyan, N., Kelly, K., Gibbs, S.G., Rautiainen, R.H., 2013. Pesticide use practices in rural Armenia. Journal of Agromedicine 18, 326-333.
- Tan, H., Li, Q., Zhang, H., Wu, C., Zhao, S., Deng, X., Li, Y., 2020. Pesticide residues in agricultural topsoil from the Hainan tropical riverside basin: Determination,

distribution, and relationships with planting patterns and surface water. Science of the Total Environment 722, 137856.

- Tarcau, D., Cucu-Man, S., Boruvkova, J., Klanova, J., Covaci, A., 2013. Organochlorine pesticides in soil, moss and tree-bark from North-Eastern Romania. Science of the Total Environment 456-457, 317-324.
- Tariq, M.I., Afzal, S., Hussain, I., Sultana, N., 2007. Pesticides exposure in Pakistan: a review. Environment International 33, 1107-1122.
- Thapa, C.B., 2017. Survey of Integrated Pest Management (IPM) Practice in Vegetable Crops of Rupandehi District, Western Nepal. International Journal of Applied Sciences and Biotechnology 5, 237-242.
- Theocharopoulos, S.P., Wagner, G., Sprengart, J., Mohr, M.E., Desaules, A., Muntau, H., Christou, M., Quevauviller, P., 2001. European soil sampling guidelines for soil pollution studies. Science of the Total Environment 264, 51-62.
- Thode, H.C., 2002. Testing for Normality. CRC press, US, p. 368.
- Thomatou, A.-A., Zacharias, I., Hela, D., Konstantinou, I., 2013. Determination and risk assessment of pesticide residues in lake Amvrakia (W. Greece) after agricultural land use changes in the lake's drainage basin. International Journal of Environmental Analytical Chemistry 93, 780-799.
- Tiwari, R.K., Singh, S., Pandey, R.S., 2019. Assessment of acute toxicity and biochemical responses to chlorpyrifos, cypermethrin and their combination exposed earthworm, Eudrilus eugeniae. Toxicology Reports 6, 288-297.
- Toan, P.V., Sebesvari, Z., Blasing, M., Rosendahl, I., Renaud, F.G., 2013. Pesticide management and their residues in sediments and surface and drinking water in the Mekong Delta, Vietnam. Science of the Total Environment 452-453, 28-39.
- Tong, Y., Niu, H., Fan, L., 2016. Willingness of Farmers to Transform Vacant Rural Residential Land into Cultivated Land in a Major Grain-Producing Area of Central China. Sustainability 8, 1-15.
- Tsaboula, A., Papadakis, E.N., Vryzas, Z., Kotopoulou, A., Kintzikoglou, K., Papadopoulou-Mourkidou, E., 2016. Environmental and human risk hierarchy of pesticides: A prioritization method, based on monitoring, hazard assessment and environmental fate. Environment International 91, 78-93.
- Turgut, C., Ornek, H., Cutright, T.J., 2011. Determination of pesticide residues in Turkey's table grapes: the effect of integrated pest management, organic farming, and conventional farming. Environmental Monitoring and Assessment 173, 315-323.
- UN, 2015. Transforming our world: the 2030 Agenda for Sustainable Development United Nations, New York, United States.
- UN, 2019. World Population Prospects. United Nations Department of Economic and Social Affair, New York.
- USEPA, 1989. Risk Assessment Guidance for Superfund Volume I, Human Health Evaluation Manual (Part A), Interim Final. U.S. Environmental Protection Agency Washington, D.C., Office of Emergency and Remedial Response, p. 291.
- USEPA, 2001. Risk Assessment Guidance for Superfund. Office of Emergency and Remedial Response, U.S. Environmental Protection Agency, Washington, DC, p. 385.
- USEPA, 2002. Supplemental Guidance for Developing Soil Screening Levels for Superfund Sites. U.S. Environmental Protection Agency, Washington, DC, p. 187.

- USEPA, 2004. An examination of EPA risk assessment principles and practices. U.S. Environmental Protection Agency, Washington, DC, p. 193.
- USEPA, 2009. Chlorpyrifos Summary Document Registration Review. United States Environmental Protection Agency, Washington, DC.
- USEPA, 2018. Regional Screening Level (RSL) Summary Table https://epaprgs.ornl.gov/chemicals/download/master_sl_table_run_JUN2011.pdf.
- USEPA, 2019. Regional Screening Table User's Guide, Mid-Atlantic Risk Assessment https://www.epa.gov/risk/regional-screening-levels-rsls-generic-tables.
- Utsumi, T., Miyamoto, M., Katagi, T., 2011. Ecotoxicological Risk Assessment of Pesticides in Terrestrial Ecosystems. Sumitomo Chemical Co., Ltd., Japan p. 19.
- Uwizeyimana, H., Wang, M., Chen, W., Khan, K., 2017. The eco-toxic effects of pesticide and heavy metal mixtures towards earthworms in soil. Environmental Toxicology and Pharmacology 55, 20-29.
- Vaidya, A., Gyenwali, D., Tiwari, S., Pande, B.R., Jørs, E., 2017. Changes in Perceptions and Practices of Farmers and Pesticide Retailers on Safer Pesticide Use and Alternatives: Impacts of a Community Intervention in Chitwan, Nepal. Environmental Health Insights 11, 1-12.
- Valcke, M., Bourgault, M.H., Rochette, L., Normandin, L., Samuel, O., Belleville, D., Blanchet, C., Phaneuf, D., 2017. Human health risk assessment on the consumption of fruits and vegetables containing residual pesticides: A cancer and non-cancer risk/benefit perspective. Environment International 108, 63-74.
- van Gestel, C.A.M., de Lima, E.S.C., Lam, T., Koekkoek, J.C., Lamoree, M.H., Verweij, R.A., 2017. Multigeneration toxicity of imidacloprid and thiacloprid to Folsomia candida. Ecotoxicology 26, 320-328.
- Vasickova, J., Hvezdova, M., Kosubova, P., Hofman, J., 2019. Ecological risk assessment of pesticide residues in arable soils of the Czech Republic. Chemosphere 216, 479-487.
- Venkateswara Rao, J., Surya Pavan, Y., Madhavendra, S.S., 2003. Toxic effects of chlorpyrifos on morphology and acetylcholinesterase activity in the earthworm, Eisenia foetida. Ecotoxicology and Environmental Safety 54, 296-301.
- Vig, K., Singh, D.K., Sharma, P.K., 2006. Endosulfan and quinalphos residues and toxicity to soil microarthropods after repeated applications in a field investigation. Journal of Environmental Science and Health, Part B 41, 681-692.
- Vryzas, Z., Alexoudis, C., Vassiliou, G., Galanis, K., Papadopoulou-Mourkidou, E., 2011. Determination and aquatic risk assessment of pesticide residues in riparian drainage canals in northeastern Greece. Ecotoxicology and Environmental Safety 74, 174-181.
- Wagner, H., Kim, A.J., Gordon, L., 2013. Relationship between Personal Protective Equipment, Self-Efficacy, and Job Satisfaction of Women in the Building Trades. Journal of Construction Engineering and Management 139, 1-7.
- Walton, A.L., LePrevost, C.E., Linnan, L., Sanchez-Birkhead, A., Mooney, K., 2017. Benefits, Facilitators, Barriers, and Strategies to Improve Pesticide Protective Behaviors: Insights from Farmworkers in North Carolina Tobacco Fields. International Journal of Environmental Research and Public Health 14, 1-13.
- Wang, B., Yu, G., Huang, J., Yu, Y., Hu, H., Wang, L., 2009. Tiered aquatic ecological risk assessment of organochlorine pesticides and their mixture in Jiangsu reach of Huaihe River, China. Environmental Monitoring and Assessment 157, 29-42.

- Wang, H., Cheng, Z., Yuan, H., Zhu, N., Lou, Z., Otieno, P., 2020. Occurrence of banned and commonly used pesticide residues in concentrated leachate: Implications for ecological risk assessment. Science of the Total Environment 710, 136287.
- Wang, J., Tao, J., Yang, C., Chu, M., Lam, H., 2017a. A general framework incorporating knowledge, risk perception and practices to eliminate pesticide residues in food: A Structural Equation Modelling analysis based on survey data of 986 Chinese farmers. Food Control 80, 143-150.
- Wang, J., Wang, J., Wang, G., Zhu, L., Wang, J., 2016. DNA damage and oxidative stress induced by imidacloprid exposure in the earthworm Eisenia fetida. Chemosphere 144, 510-517.
- Wang, J.H., Zhu, L.S., Liu, W., Wang, J., Xie, H., 2012. Biochemical responses of earthworm (Eisenia foetida) to the pesticides chlorpyrifos and fenvalerate. Toxicology Mechanisms and Methods 22, 236-241.
- Wang, K., Pang, S., Mu, X., Qi, S., Li, D., Cui, F., Wang, C., 2015a. Biological response of earthworm, Eisenia fetida, to five neonicotinoid insecticides. Chemosphere 132, 120-126.
- Wang, K., Qi, S., Mu, X., Chai, T., Yang, Y., Wang, D., Li, D., Che, W., Wang, C., 2015b. Evaluation of the Toxicity, AChE Activity and DNA Damage Caused by Imidacloprid on Earthworms, Eisenia fetida. Bulletin of Environmental Contamination and Toxicology 95, 475-480.
- Wang, W., Jin, J., He, R., Gong, H., 2017b. Gender differences in pesticide use knowledge, risk awareness and practices in Chinese farmers. Science of the Total Environment 590-591, 22-28.
- Wang, X., Wang, D., Qin, X., Xu, X., 2008. Residues of organochlorine pesticides in surface soils from college school yards in Beijing, China. Journal of Environmental Sciences 20, 1090-1096.
- Wang, X., Zhu, X., Peng, Q., Wang, Y., Ge, J., Yang, G., Wang, X., Cai, L., Shen, W., 2019. Multilevel ecotoxicological effects of imidacloprid on earthworm (Eisenia fetida). Chemosphere 219, 923-932.
- Wang, Z., 2007. Regional Study on Soil Polycyclic Aromatic Hydrocarbons in Liaoning: Patterns, Sources and Cancer Risks. Dalian University of Technology, Dalian, China.
- Wardell, J.D., Read, J.P., Colder, C.R., Merrill, J.E., 2012. Positive alcohol expectancies mediate the influence of the behavioral activation system on alcohol use: a prospective path analysis. Addictive Behaviors 37, 435-443.
- Watson, C., Troynikov, O., Lingard, H., 2019. Design considerations for low-level risk personal protective clothing: a review. Industrial Health 57, 306–325.
- Watts, M., 2012. Chlorpyrifos as a possible global POP https://www.paneurope.info/old/News/PR/121009_Chlorpyrifos_as_POP_final.pdf.
- Wee, S.Y., Aris, A.Z., 2017. Ecological risk estimation of organophosphorus pesticides in riverine ecosystems. Chemosphere 188, 575-581.
- Wenzel, K.D., Manz, M., Hubert, A., Schuurmann, G., 2002. Fate of POPs (DDX, HCHs, PCBs) in upper soil layers of pine forests. Science of the Total Environment, 143-154.
- WHO, 2008. Pesticides: Children's Health and the Environment https://www.who.int/ceh/capacity/Pesticides.pdf.
- WHO, 2011. Generic risk assessment model for indoor and outdoor space spraying of
insecticides. WHO, Geneva 27, Switzerland, p. 72.

WHO, 2012. FAO/WHO Joint Meeting on Pesticide Management (JMPM) https://www.who.int/neglected_diseases/vector_ecology/pesticide-

management/Sixth_FAO_WHO_jmpm_2013.pdf?ua=1, Rome, p. 44.

- WHO,2013.ConsolidatedARVguidelines2013https://www.who.int/hiv/pub/guidelines/arv2013/intro/keyterms/en/.
- WHO, 2018. Pesticide residues in food. https://www.who.int/news-room/fact-sheets/detail/pesticide-residues-in-food.
- Wilkinson, C.F., Christoph, G.R., Julien, E., Kelley, J.M., Kronenberg, J., McCarthy, J., Reiss, R., 2000. Assessing the risks of exposures to multiple chemicals with a common mechanism of toxicity: how to cumulate? Regulatory Toxicology and Pharmacology 31, 30-43.
- Wong, H.L., Garthwaite, D.G., Ramwell, C.T., Brown, C.D., 2017. How does exposure to pesticides vary in space and time for residents living near to treated orchards? Environmental Science and Pollution Research 24, 26444-26461.
- Wong, L.I., Labrecque, M.P., Ibuki, N., Cox, M.E., Elliott, J.E., Beischlag, T.V., 2015. p,p'-
Dichlorodiphenyltrichloroethane(p,p'-DDT)andp,p'-
dichlorodiphenyldichloroethylene (p,p'-DDE) repress prostate specific antigen levels in
human prostate cancer cell lines. Chemico-Biological Interactions 230, 40-49.
- Xu, L., Granger, C., Dong, H., Mao, Y., Duan, S., Li, J., Qiang, Z., 2020. Occurrences of 29 pesticides in the Huangpu River, China: Highest ecological risk identified in Shanghai metropolitan area. Chemosphere 251, 126411.
- Xu, M., Qiu, Y., Bignert, A., Zhou, Y., Zhu, Z., Zhao, J., 2015. Organochlorines in free-range hen and duck eggs from Shanghai: occurrence and risk assessment. Environmental Science and Pollution Research 22, 1742-1749.
- Xu, P., Diao, J., Liu, D., Zhou, Z., 2011. Enantioselective bioaccumulation and toxic effects of metalaxyl in earthworm Eisenia foetida. Chemosphere 83, 1074-1079.
- Xu, R., Kuang, R., Pay, E., Dou, H., de Snoo, G.R., 2008. Factors contributing to overuse of pesticides in western China. Environmental Sciences 5, 235-249.
- Yadav, I.C., Devi, N.L., Li, J., Zhang, G., Breivik, K., 2017. Possible emissions of POPs in plain and hilly areas of Nepal: Implications for source apportionment and health risk assessment. Environmental Pollution 220, 1289-1300.
- Yadav, I.C., Devi, N.L., Li, J., Zhang, G., Shakya, P.R., 2016. Occurrence, profile and spatial distribution of organochlorines pesticides in soil of Nepal: Implication for source apportionment and health risk assessment. Science of the Total Environment 573, 1598-1606.
- Yahia, D., Elsharkawy, E.E., 2014. Multi pesticide and PCB residues in Nile tilapia and catfish in Assiut city, Egypt. Science of the Total Environment 466/467, 306-314.
- Yang, D., Qi, S.-H., Zhang, J.-Q., Tan, L.-Z., Zhang, J.-P., Zhang, Y., Xu, F., Xing, X.-L., Hu, Y., Chen,
 W., Yang, J.-H., Xu, M.-H., 2012. Residues of Organochlorine Pesticides (OCPs) in
 Agricultural Soils of Zhangzhou City, China. Pedosphere 22, 178-189.
- Yang, F.-W., Li, Y.-X., Ren, F.-Z., Wang, R., Pang, G.-F., 2019. Toxicity, residue, degradation and detection methods of the insecticide triazophos. Environmental Chemistry Letters 17, 1769-1785.
- Yang, X., Wang, F., Bento, C.P.M., Xue, S., Gai, L., van Dam, R., Mol, H., Ritsema, C.J., Geissen,

V., 2015. Short-term transport of glyphosate with erosion in Chinese loess soil-a flume experiment. Science of the Total Environment 512/513, 406-414.

- Yang, X., Wang, F., Meng, L., Zhang, W., Fan, L., Geissen, V., Ritsema, C.J., 2014. Farmer and retailer knowledge and awareness of the risks from pesticide use: a case study in the Wei River catchment, China. Science of the Total Environment 497/498, 172-179.
- Yarpuz-Bozdogan, N., 2018. The importance of personal protective equipment in pesticide applications in agriculture. Current Opinion in Environmental Science & Health 4, 1-4.
- Yasmin, S., D'Souza, D., 2007. Effect of pesticides on the reproductive output of Eisenia fetida. Bulletin of Environmental Contamination and Toxicology 79, 529-532.
- Yazdanpanah, M., Forouzani, M., Hojjati, M., 2015. Willingness of Iranian young adults to eat organic foods: Application of the Health Belief Model. Food Quality and Preference 41, 75-83.
- Younglai, E.V., Holloway, A.C., Lim, G.E., Foster, W.G., 2004. Synergistic effects between FSH and 1,1-dichloro-2,2-bis (P-chlorophenyl)ethylene (P,P'-DDE) on human granulosa cell aromatase activity. Human Reproduction 19, 1089-1093.
- Yuantari, M.G., Van Gestel, C.A., Van Straalen, N.M., Widianarko, B., Sunoko, H.R., Shobib, M.N., 2015. Knowledge, attitude, and practice of Indonesian farmers regarding the use of personal protective equipment against pesticide exposure. Environmental Monitoring and Assessment 187, 1-7.
- Zbytniewski, R., Buszewski, B., 2002. Sorption of Pesticides in Soil and Compost. Polish Journal of Environmental Studies 11, 179-184.
- Zhang, A., Luo, W., Sun, J., Xiao, H., Liu, W., 2015. Distribution and uptake pathways of organochlorine pesticides in greenhouse and conventional vegetables. Science of the Total Environment 505, 1142-1147.
- Zhang, H.B., Luo, Y.M., Zhao, Q.G., Wong, M.H., Zhang, G.L., 2006. Residues of organochlorine pesticides in Hong Kong soils. Chemosphere 63, 633-641.
- Zhang, Q., Zhang, B., Wang, C., 2014. Ecotoxicological effects on the earthworm Eisenia fetida following exposure to soil contaminated with imidacloprid. Environmental Science and Pollution Research 21, 12345-12353.
- Zhang, W., Jiang, F., Ou, J., 2011. Global pesticide consumption and pollution: with China as a focus, International Academy of Ecology and Environmental Sciences, pp. 125-144.
- Zheng, X., Liu, X., Liu, W., Jiang, G., Yang, R., 2009. Concentrations and source identification of organochlorine pesticides (OCPs) in soils from Wolong Natural Reserve. Science Bulletin 54, 743-751.
- Zhou, S., Duan, C., Michelle, W.H.G., Yang, F., Wang, X., 2011. Individual and combined toxic effects of cypermethrin and chlorpyrifos on earthworm. Journal of Environmental Sciences 23, 676-680.
- Zou, X., Xiao, X., Zhou, H., Chen, F., Zeng, J., Wang, W., Feng, G., Huang, X., 2018. Effects of soil acidification on the toxicity of organophosphorus pesticide on Eisenia fetida and its mechanism. Journal of Hazardous Materials 359, 365-372.
- Zyoud, S.H., Sawalha, A.F., Sweileh, W.M., Awang, R., Al-Khalil, S.I., Al-Jabi, S.W., Bsharat, N.M., 2010. Knowledge and practices of pesticide use among farm workers in the West Bank, Palestine: safety implications. Environmental Health and Preventive Medicine 15, 252-261.

English summary

Globally, crop loss has decreased with the increased application of chemical pesticides. Nonetheless, food safety has become an emerging challenge. Farmers prefer to use chemical pesticides because they are quick and effective at controlling plant diseases, insects and pests. Although the use of pesticides has minimized crop loss and helped maintain yields, pesticide misuse and overuse has ultimately put humans and the environment at great risk. Pesticides persist in the environment and have been shown to traverse the placenta and cause abnormalities in the fetus.

Every year, Nepal imports tons of pesticides mainly from two countries: India and China. These pesticides are mostly used for agriculture, particularly in the Terai region where vegetables are cultivated. The high use and misuse of chemical pesticides may lead to land pollution which can affect human health and ultimately cause ecosystem degradation. In Nepal, there is very little legal framework and few policies concerning pesticide use. Due to insufficient monitoring and supervision, even the policies that are in place are ineffective. For instance, Nepal has banned persistent and toxic pesticides such as phorate and endosulfan for many years, however they are still used in rural areas. Most of the rural farmers in Nepal lack education and are unaware of the toxic effects of the pesticides they use in their fields. Most pesticide retailers (agrovets) also lack sufficient training and knowledge. This makes them incapable of properly diagnosing plant diseases and recommending the appropriate chemical pesticides to farmers. Farmers and retailers are exposed to harmful chemicals when handling pesticides which puts them at risk. The fate and behaviour of pesticides differ with differences in biotic and abiotic factors such as soil, weather, climate, flora, farming practice and pesticide properties, all of which influence ecotoxicity.

Chapter 2 of this thesis aimed to examine pesticide use in a rural area of Nepal. We looked at pesticide use on vegetable crops and identified factors affecting the safety behaviour of farmers and pesticide retailers. We analysed interview data from 183 farmers and 45 retailers from the Gaidahawa Rural Municipality, Nepal. The Chapter made use of the Health Belief Model (HBM) to present the study results. Correlations between HBM constructs and the safety behaviour of farmers revealed that factors such as education, perceived threat and benefit, self-efficacy, and cues to action positively affected their pesticide use safety behaviour. Farmers perceived that pesticides posed low threats to humans and ecosystems. Due to the perceived lower threats from pesticides, farmers perceived that the use of personal protective equipment (PPE) had low benefits and high barriers such as unavailability and discomfort while using. Likewise, for retailers, perceived threat and cues

to action affected their safety behaviour. Although few retailers were aware of the effects of pesticides on their health and the environment, most retailers complained that there was not enough training on pesticide safety behaviour. While age and perceived barrier negatively affected farmers' safety behaviour, there was no such relationship seen in retailers. Farmers applied the highest amount of organophosphate insecticides and carbamate fungicides on eggplants, chillies and tomatoes. The short-term risks associated with pesticide exposure in farmers were headaches (>70%), skin irritation (>60%) and eye irritation (>30%). Most farmers perceived pesticides as a "medicine" and not a poison since pesticides cured plant diseases. This Chapter recommends that more educational activities be made available to farmers and retailers. These educational activities could include organizing pesticide-related documentaries and talk shows, broadcasting news via the radio, television, newspapers and mobile applications in order to increase the levels of knowledge and awareness among farmers and retailers. The government or scientists should provide them with training on how to safely handle pesticides and implement the best integrated pest management methods. The illegal import of chemical pesticides (including banned pesticides) in and around the border areas in Nepal is running rampant.

The study in Chapter 3 presents data from two different farming types: integrated pest management (IPM) and conventional farming. In this study, based on interview with farmers we looked at 23 pesticide residues, including their metabolites, in three vegetable crops: chillies (n=27), tomatoes (n=32) and eggplants (n=27). We used our results to assess the possible health risks from dietary exposure to pesticides. The study detected 14 different pesticide residues, including insecticides and fungicides, in the vegetables. The number of pesticides in cocktails from IPM and conventional farming systems ranged from 0-6 and 1-7, respectively. About 88% of the vegetables from conventional farming contained pesticide cocktails, while from IPM it was 47%. The study assessed adults' and adolescents' dietary risk of 6 organophosphate (OP) insecticides (dichlorvos, omethoate, dimethoate, triazophos, chlorpyrifos, profenofos), 1 neonicotinoid insecticide (imidacloprid), 1 anthranilic diamide insecticide (chlorantraniliprole), 1 benzimidazole fungicide (carbendazim), 1 phenylamide fungicide (metalaxyl), and 1 micro-organism derived insecticide (emamectin). OP insecticides such as quinalphos and other metabolites such as 3,5,6-trichloro-2-pyridinol (TCP) and N-(2,6-dimethylphenyl)-N-(methoxyacetyl)alanine lack human health-based toxicological reference values such as Acute Reference Dose (ARfD) and Acceptable Daily Intake (ADI), thus the health risks were unknown. The study adopted EFSA's risk assessment methods: hazard quotient (HQ) and hazard index (HI). ARfD and ADI information was obtained from the EFSA, JMPR and EU databases.

We detected residues of pesticides in over 90% of the sampled eggplant crops and 100% of the sampled tomato and chilli. The concentration of triazophos, omethoate, chlorpyrifos and carbendazim residues exceeded the EU MRLs (triazophos, 10 μ g/kg; omethoate, 10

 μ g/kg; chlorpyrifos, 10-100 μ g/kg; carbendazim, 100-500 μ g/kg) in some samples. One individual tomato sample from conventional farming system contained up to 7 different residues, forming pesticide cocktails. The average concentration of imidacloprid, chlorpyrifos and its metabolite TCP in the vegetables sampled from conventional farming were found to have higher concentrations (p <0.05) than the vegetables sampled from IPM farming. This shows that the adoption of IPM farming may significantly reduce pesticide residues in vegetables. Of all the vegetables and pesticides, carbendazim in tomatoes was high (p <0.05).

The dietary risk assessment showed that the consumption of tomatoes could impose an acute risk from triazophos and chlorpyrifos for adults and adolescents. This Chapter recommends strengthening legal frameworks and offering alternative programs that support food safety. It recommends to either ban or strongly restrict the use of triazophos and chlorpyrifos in tomatoes. We also suggest the adoption of IPM farming to assure the safety of vegetables meant for human consumption.

The study in Chapter 4 aimed to assess non-dietary risk due to pesticide residues in soils by examining three major exposure pathways in humans: dermal contact, direct ingestion, and inhalation. We measured the concentration and distribution of pesticides at three depths (0-5 cm, 15-20 cm and 35-40 cm) of soil samples (n=147) from 2 farming types: integrated pest management (IPM) and conventional. We predicted the concentration of pesticides and their degradation products and compared these values with their internationally adopted soil regulatory guidance values. Hazard quotient and hazard index were used to characterize non-cancer risk in adults and adolescents. We used the USEPA models to assess cancer risk based on the average daily dose of pesticide exposure via the different exposure routes. We found pesticide residues more frequently in upper layers of soil (0-5 cm). We detected up to 7 residues in an individual soil sample from conventional farming. Soils from eggplant fields had a higher number of pesticide residues (p < 0.05) than tomato fields. Likewise, soils from conventional farming had a significantly higher number of pesticide residues (p <0.05) than the soils from IPM farming. TCP and chlorantraniliprole residues were detected the most frequently. Of all tested pesticides and their degradation products (n=23), we detected 15 in soils, where chlorpyrifos and p,p'-DDT residues were found to be in the highest concentrations. Overall, the human health cancer and non-cancer risks posed by the pesticides in soils was negligible. Predicted concentrations of pesticides (PEC) were higher than their guidance values for most pesticides. This Chapter recommends promotion of IPM methods for reducing pesticide pollution in soils. We also urge future research focused on the risk that water pollution poses to humans.

Chapter 5 assesses ecological risk of pesticides (n=9) at 3 depths of soil. We performed a correlation analysis between the risk at a farmer's site and the farmer's knowledge and

184

behaviour related to pesticide effects and waste management. We adopted globally used methods for risk assessment such as toxicity exposure ratio (TER) and risk quotient (RQ). We estimated risk for EFSA's soil organisms such as earthworms (Eisenia fetida), enchytraeids (Enchytraeus crypticus), springtails (Folsomia candida), mites (Hypoaspis aculifer) and nitrogen and carbon mineralization microorganisms. We made a comparison between the measured pesticide concentrations in soils and the pesticide permissible concentration (or the soil regulatory guidance value for pesticides whose RQ and TER could not be calculated). Profenofos, imidacloprid and chlorpyrifos showed a higher risk (RQ>1) at 2 depths (0-5 cm and 35-40 cm), 1 depth (35-40 cm) and 3 depths (0-5 cm, 15-20 cm and 35-40 cm), respectively. Estimated TER for most pesticides and their metabolites did not show chronic risks of pesticide exposure for soil organisms, except for F. candida. The TER of chlorpyrifos was <5 at all depths of soil which posed a chronic risk for *F. candida*. About 16% of the sites posed ecological risks, determining chlorpyrifos to be a main soil pollutant. The risk at sites increased significantly when farmers had poor pesticide knowledge and waste management practices. Ecotoxicity data related to soil organisms was not sufficient for all the pesticides, thus the risk assessment was not possible for all of the pesticides detected in soils. For pesticides to which the risk assessment was not possible, we compared their concentration with the existing threshold or guidance values in soil. A higher tiered method of risk assessment including residue cocktails could create a better picture of pesticide risks for ecological entities that future research might focus on.

Acknowledgements

The time has come to acknowledge and express my very great appreciation to all who helped me in my PhD journey. I would like to thank various people and institutions in The Netherlands and Nepal whose help made me to complete my PhD thesis.

First of all, my deepest thanks to Chair, Soil Physics and Land Management Group (SLM)-Prof. Dr. Coen Ritsema. I cannot forget the day you wrote an email that I got awarded Netherlands Universities Foundation for International Cooperation (Nuffic) fellowship for my PhD. Thank you so much for your trust and support during my PhD period. I would like to thank the Nuffic for funding me to pursue my PhD.

I would like to express my sincere and deepest gratitude to my promotor Prof. Dr. Violette Geissen for your support during the long journey of my PhD. Prof. Violette, thank you so much for accepting me as a Sandwich PhD under your supervision. I remember the first days in Wageningen University and Research (WUR) when we met at Atlas building and you helped me to connect with the staffs and students of our research group, SLM. Since then, I never felt alone and have nice memories. Your skilful, energetic and positive attitude really amazed me. I got generous freedom to be creative that helped me to think critically and act logically. You helped me a lot throughout my PhD providing reminders, suggestions, ideas and feedbacks that made me able to publish articles. At the time when I was in Nepal, I had many skype calls and discussions with you when I struggled while writing articles. I am also thankful to you for putting me in contact with Dr. Liangxin Fan and Prof. Paul T.J. Scheepers from School of Surveying and Land Information Engineering, Henan Polytechnic University, China and Radboudumc, Radboud Institute for Health Sciences, Netherlands, respectively. In a nutshell, I enjoyed very much working with you and thank you so much for everything!

I also want to express my gratitude to my co-promotors Dr. Kishor Atreya and Dr. Xiaomei Yang, for your patient guidance, enthusiastic encouragement and useful critiques during my PhD. Dr. Kishor, thank you so much for your support. Your help during my field survey and guidance on data entry was invaluable and countless. Your dedication towards research and publication inspired me and motivated to work harder. You used to ring me at the time when internet was not possible, which helped me to keep my progress on schedule. Thank you for all your timely and constructive comments, edits, corrections, suggestions and discussions in all the published articles. Thank you for your warm welcome to your office and home for guidance and discussions. Dr. Xiaomei, your teaching and encouragement were crucial for me, I learned a lot with you since I started my PhD. You helped me in preparing my PhD proposal focusing on pesticide residues and risk assessment before submitting it to the Graduate school, WIMEK. You were always available for help when I was in need during my PhD period.

Moreover, I would like to thank Prof. Paul T.J. Scheepers, Dr. Liangxin Fan, Dr Hans Mol and Dr. Jana Vašíčková (RECETOX) for your corrections and suggestions as well as co-authorship with my publications. Your critical comments to the manuscripts helped me learn more and successfully publish articles.

I am also grateful to Paul Zomer, Rita Boerrigter-Eenling, Joost Memelink, Nicolas Beriot and Suryati Sunarto for all their help and sharing of knowledge during my lab work in Wageningen Food Safety Research, part of WUR. I learned a lot about pesticide analytics as well as GC-MS/MS and LC-MS/MS with you. Your guidance and support helped me a lot and the lab work ended well.

Moreover, I would like to thank all the colleagues who greatly helped me a lot during questionnaire survey as well as vegetables sample collection at the field. Thank you Garima Joshi, Samikshya Pandit, Sunil Ghimire, Kumar Pun Magar, Dipak Pandey and Ravi Raj Mishra for your help in the survey. Thank you Dinesh Ghimire, Prashant Chaudhary, Santosh Gupta and Ramesh Lodh for your help in the sample collection. I would like to thank all the residents of the study area who helped me by responding during the survey. Thank you for the Gaidahawa Rural Municipality officials who cooperated during the survey and provided the information of the municipality. Special thanks to Santosh Bhusal, agricultural extension officer of the municipality for your advice in selecting the villages and sampling locations during the fieldwork. Thank you Prof. Dr. Roshan M. Bajracharya, Prof. Dr. Keshab R. Adhakari, Dr. Bhaba Prasad Tripathi, Dr. Jagannath Joshi, Dr. Sanjeev Bhuchar, Samden Lama Sherpa, Sarkardeep Shrestha, Dil Bahadur Bhattarai, Bihani Thapa, Shova Shrestha for hosting Dr. Johannes Botschek and his team as well as helping me to organize a field visit for them in a course entitled "Human Induced Land Degradation". Thank you Standy Christianto for providing me the field photos. I would also like to thank Shikhar Prasad Lamichhane for your help and skills with graphics that I needed for some of my articles. I would like to thank Poshan Niraula for teaching me basics of the ArcGIS. Thank you Sunil Shrestha for your help in booking air tickets and accommodation during my travel in and out.

A special thanks goes to all staff members from SLM research group for their help, friendship and cooperation. I would like to thank Dr. Jos van Dam for all your help in refining the PhD research proposal. I had my first experience of canoeing in Netherlands and you helped me to do it. Thank you Marnella van der Tol and Rianne Maasen for all your help and support, including documentation, accommodation as well as solving technical issues when needed. I am thankful to Marnella as you received me very well in Atlas building for the first time and helped me to move my luggage to my room with your nice car. Thank you Robin Palmer and Jaclyn Bandy for your help in English revision of the articles. I want to thank Geert Belgers and Kees van der Klift who handled administrative and financial matters throughout my PhD study. I want to convey special thanks to my WUR colleagues: Shyam Shrestha, Ashok Subedi, Carlos Faundez Urbina, Ammar Ali, Vera Felix da Graca Silva, Raoul Kpegli, Karrar Mahdi, Yueling Qi, Ricardo Teixeira da Silva, Mousumi Akter, Nicolas Beriot, Murilo Vianna, Kaveh Hosseini and Godeberthe Nkeshimana for your friendship during the last 4.5 years. Thank you Dr. Isaurinda Baptista and Dr. Suman Bajracharya for inviting me to your PhD party that inspired a lot to make me confident enough since the early days of my PhD journey. Thanks to my previous Nepalese colleagues in WUR: BP Dhakal, TN Joshi, Sudip Subedi, RP Mainali, Binita Thapa, Aakash Koirala, Deny Kumar Shrestha, Kriti Shrestha, Pragya Shrestha and Dr. Arun Kumar Pratihast for your wonderful friendship. I can remember the typical Nepalese food that we had together in the past that made a homely environment for me during my stay in The Netherlands. Thanks to my corridor international friends in Bornsesteeg: Kashmira Bandal and Ade Ryane for celebrating with international foods in weekends.

I am grateful to my colleagues at the Progressive Sustainable Developers Nepal (PSD-Nepal) for arranging everything before and during my PhD period. I would like to thank Dr. Sangeeta Poudel, Dr. Peter Matthews, Dr. Banalata Sen, Dr. Prakash Bhave, Ambika Pandey, Suman Dangol, Sioux Cumming, Roshan Nepal, Pradip Shrestha, Bhanu Panthi and Jeevan Thapa for helping in moving the PSD-Nepal projects smoothly during my absence.

Last but not least, I am very grateful to my family members, including my father, mother, sisters and brothers-in-law. My loving wife Ambika Bhandari and my beautiful daughter Aaravi Bhandari deserve very special thanks for their continuous moral support, pray and blessings. Ambika, you were always there to support me and encourage me when I felt bad due to rejections of articles and frustrations. During the last year of my PhD, I could stay in Nepal with you and your endless love and support which I will never forget.

Govinda Bhandari Wageningen, The Netherlands

About the author

Govinda Bhandari (1984), son of Thaneshwar Bhandari and Kamala Devi Bhandari, was born in the village of Bastu in Gulmi district, Nepal. His father is a retired government secretary of a village in the district, while his mother is a house wife.

Govinda completed his 10+2 level from Terai, Rupandehi district. Later, he went to the capital city, Kathmandu to do his BSc in Environmental Science. After the completion of BSc from Tribhuvan University, he got an opportunity to do MSc in Ecology and Environment from Sikkim Manipal University, India. After completing his MSc, he worked as a Lecturer at Hilbert International College, Kamaladi, Kathmandu for 2 years. Later, started an Non-Government Office (NGO) named Progressive Sustainable Developers Nepal (PSD-Nepal), which aimed primarily for working in the areas of health and environment in Nepal. In the NGO, he is the President and actively engaged in research training and capacity building for Nepalese junior researchers as well as helping them with their thesis and reports. His interests grew in scientific writing and publishing. Within a short period after the NGO, he started a scientific journal, the *International Journal of Environment* (eISSN: 2091-2854), an official journal of PSD-Nepal. He is the Editor-in-Chief of the journal since 2013.

In the year 2016, he was awarded a Nuffic fellowship through a project NFP-PhD-5160957247, funded by the Dutch government to do his Sandwich PhD in the Wageningen University and Research, The Netherlands. He then started his PhD work in the Soil Physics and Land Management Group (SLM) supervised by Prof. Dr. Violette Geissen and co supervised by Dr. Xiaomei Yang and Dr. Kishor Atreya. During his PhD, he tried to explore the pesticide residues in Nepalese vegetable and soils as well as assessed their risk to human health and the environment, which is quite a new contribution for science and a baseline for pesticide perspectives in Nepal.

During his PhD research, Govinda participated in workshops, trainings and conferences organized in Nepal and The Netherlands. He organized capacity building trainings on scientific writing and publishing for Nepalese researchers. In his free time, Govinda likes playing badminton and listening Nepalese folk songs. After his PhD, Govinda is looking forward to working in projects related to ecotoxicology in Nepal or elsewhere in the globe.

Latest publications

Bhandari G, Atreya K, Yang X, Fan L, Geissen V. Factors affecting pesticide safety behaviour: The perceptions of Nepalese farmers and retailers. Science of The Total Environment 2018; 631-632: 1560-1571.

Bhandari G, Zomer P, Atreya K, Mol HGJ, Yang X, Geissen V. Pesticide residues in Nepalese vegetables and potential health risks. Environ Res 2019; 172: 511-521.

Bhandari G, Atreya K, Scheepers PTJ, Geissen V. Concentration and distribution of pesticide residues in soil: Non-dietary human health risk assessment. Chemosphere 2020; 253: 126594.

Manuscript (under review)

Ecological risk assessment of pesticide residues in soils from vegetable production areas: A case study in S-Nepal. Environmental Pollution.

Acknowledgements of financial support

This PhD research was financially supported by the Netherlands Universities Foundation for International Cooperation (Nuffic)-through the PhD grant number NFP-PhD-5160957247.



Netherlands Research School for the Socio-Economic and Natural Sciences of the Environment

DIPLOMA

for specialised PhD training

The Netherlands research school for the Socio-Economic and Natural Sciences of the Environment (SENSE) declares that

Govinda Bhandari

born on 23 February 1984 in Gulmi, Nepal

has successfully fulfilled all requirements of the educational PhD programme of SENSE.

Wageningen, 2 February 2021

Chair of the SENSE board

Prof. dr. Martin Wassen

The SENSE Director

Prof. Philipp Pattberg

The SENSE Research School has been accredited by the Royal Netherlands Academy of Arts and Sciences (KNAW)



KONINKLIJKE NEDERLANDSE AKADEMIE VAN WETENSCHAPPEN



The SENSE Research School declares that **Govinda Bhandari** has successfully fulfilled all requirements of the educational PhD programme of SENSE with a work load of 33.7 EC, including the following activities:

SENSE PhD Courses

- o Environmental research in context (2016)
- o Basic Statistics (2016)
- Research in context activity: 'Organising seminar and the Nepal excursion 'Human induced land degradation' (2017)

Other PhD and Advanced MSc Courses

- Information Literacy PhD including EndNote Introduction, Wageningen Graduate Schools (2016)
- o Reviewing a Scientific Paper, Wageningen Graduate Schools (2016)
- Training course on GIS and GPS for faculty members, researchers, MSc and PhD students, Nepal GIS Society (2017)
- o Journal Publishing Practices and Standards, INASP and Tribhuvan University (2017)
- o Peer Review course for researchers and journal editors, INASP (2019)
- Publication Ethics course for researchers and journal editors, INASP (2020)
- o Copyright and Licensing course for researchers and journal editors, INASP (2020)

Management and Didactic Skills Training

- o Guest facilitator in the AuthorAID Online Course 'Research Writing' (2016)
- Teaching in the MSc course 'Scientific writing and publishing' (2019)
- Organising a workshop on 'Scientific writing and publishing' in Kathmandu Valley at for faculty members, researchers and PhD candidates (2019)

Oral Presentations

• Land Degradation in Nepal: A Case of Pesticide Pollution. Human induced Land Degradation, 2-10 September 2017, Kathmandu, Nepal

SENSE coordinator PhD education Dr. ir. Peter Vermeulen